ALBUQUERQUE OPERATIONS MANUAL

SECTIONS 16, 17



JACOBS ENGINEERING GROUP INC. ENGINEERS . CONSTRUCTORS

9102080296 910205 PDR WASTE WM-43 PDR



JE JACOBS ENGINEERING GROUP INC. SECTION 10.2.1

PAGE 2 OF 18

3.10 Sample Bottles: Use pre-cleaned, oven dried containers as per EPA. A source of bottles is: I-CHEM 23787 Eichler St., Bldg. F Hayward, CA 94545 (415) 782-3905

- 3.11 Plastic Sneet(s) and Aluminum Foil
- 3.12 Parafilm or its equivalent
- 3.13 Rope
- 3.14 Pen, pencil, magic marker, paper, and masking tape
- 3.15 Sampling Data Forms: Water Quality Sampling Record. Chain-of-Custody Forms, etc.

4.0 CHEMICALS AND REAGENTS

NOTE: Whenever required by the procedure, JEG purchases and uses prepared, calibrated, and certified solutions from chemical supply firms. The following are listed in the event solutions are not readily available or are used whenever procedure parameters do not require certified solutions. In the event that solutions require preparation, they will be prepared at the laboratory before going into the field.

- 4.1 Hydrochloric Acid: HCl, 30-38%, ~12 M, (~12 N). Reagent Grade. Purchased from chemical supply firms. Handle bottle with rubber gloves and use a hood.
- 4.2 No N HCl: Add 250 mL of concentrated Hydrochloric Acid to a 500 mL volumetric flask containing ~200 mL of distilled water, and mix solution well. Cool solution in an ice water path. Add distilled water to 500 mL mark, stopper flask, and mix solution well. Store solution in a clean polyetnylene bottle labelled with "~6 N HCl" "250 mL HC1/250 mL distilled water", "Corrosive Acid", date, and preparer's initials. NOTE: Always add acid to water. Snake 5 N HCl bottle to mix solution prior to use.
- 4.3 ~2 N HCl: Add 167 mL of concentrated Hydrochloric Acid to a 1000 mL volumetric flask containing 750 mL of distilled water and mix solution. Add distilled water to 1000 mL mark, stopper flask, and mix solution well. Store solution in a clean polyethylene bottle labelled with "2 N HC1", "167 mL HC1/1000 mL distilled water", "Corrosive Acid", date, and preparer's initials. Shake solution prior to use.
- 4.4 Nitric Acid: HNO3, ~70%, ~16 M (~16 N). Reagent Grade. Purchased from chemical supply firms. Handle bottle with rubber gloves and use a hood.

PAGE 3 OF 1.8

- 4.5 ~8 N HNO3: Add 250 mL of concentrated Nitric Acid to a 500 mL volumetric flask containing 200 mL of distilled water and mix solution well. Cool solution in an ice water bath. Add distilled water to 500 mL mark, stopper flask, and mix solution well. Store solution in a clean polyethylene bottle labelled with "~8 N HNO3", "250 mL HN03/250 mL distilled water", "Corrosive Acid", date, and preparer's initials. NOTE: Always add acid to water. Shake 8 N HNO3 bottle and contents to mix solution prior to use.
- 4.6 Sulfuric Acid: H2SO4, ~98%, ~18 M (~30 N). Reagent Grade. Purchased from chemical supply firms. Handle pottle with rubber gloves and use a hood.
- 4.7 ~9 M H2SO4: While swirling water, slowly add 250 mL of concentrated H2S04 to a 500 mL volumetric flask containing 200 mL of distilled water, and mix solution. Cool solution in an ice water bath. Add distilled water to the 500 mL mark, stopper flask, and mix solution well. Store solution in a clean polyethylene bottle labelled with "~9 M H2SO4 (~18 N)", "250 mL H2SO4/250 mL distilled water", "Corrosive Acid", date, and preparer's initials. NOTE: Always add this acid to water. Use caution when handling H2SO4: this acid is extremely corrosive and will sputter when adding to water. Keep a bottle of ~1 N NaOH to neutralize any spilled H2SO4 and a bottle of distilled water nearby.
- 4.8 Sodium Hydroxide: NaOH. Reagent Grade. In pellet form. Purchased from chemical supply firms.
- 4.9 wid N NaOH: Dissolve ?2 grams of Sodium Hydroxide pellets in a 100 mL volumetric flask containing ~60 mL of distilled water. Swirl solution and cool in an ice bath. If necessary, add 20 mL of distilled water to dissolve remainder of NaOH pellets. Cool solution if necessary. Add distilled water to 100 mL mark of flask, stopper flasks, and mix solution well. Store solution in a polyethylene bottle labeled with "~18 N NaOH", "72 g NaOH/100 mL distilled water", date, and preparer's initials.
- 4.10 ~1 N NaOH: Dissolve 4.0 grams of Sodium Hydroxide in a 100 mL volumetric flask containing ~75 mL of distilled water and mix to dissolve NaOH pellets. Cool solution in an ice bath. Add distilled water to 100 mL mark of the flask, stopper flask, and mix solution well. Store solution in a polyethylene bottle lapeled with "01 N NaOH", "4.0 g NaOH/100 mL distilled water", date, and preparer's initials. NOTE: Prepare fresh solution every 30 days.
- 4.11 Zinc Acetate: Zn(C2H3O2)2, also Zn(Ac)2. Reagent Grade. Purchased from chemical supply firms.
- 4.12 v220 g Zn(Ac)₂/L: Dissolve 22 grams of Zinc Acetate in a 100 mL volumetric flask containing v75 mL of distilled water and mix solution to dissolve the salt. If necessary, add v15 mL of distilled water to dissolve remaining salt, and mix solution. Add distilled water to the 100 mL mark, stopper flask, and mix solution

PAGE 4 OF 18

well. Store solution in a polyethylene bottle labeled with "-220 a Zn(Ac)2/L", "22 g Zn(Ac)2/100 mL distilled water", date, and preparer's initials. NOTE: Prepare fresh solution every 30 days and discard old solution.

4.13 Hexane. Reagent Grade. Purchased from chemical supply firms.

4.14 pH paper (1-10): Hydrion or its equivalent.

5.0 PROCEDURE: PREPARATION FOR THE COLLECTION OF WATER SAMPLES

Two (2) weeks prior to water sampling, the following steps must be initiated and completed by appropriate members of the TAC and its subcontractors.

- 5.1 The Site Hydrogeologist (SH) will determine the site locations to be sampled, number of samples per location, and analytes to be determined in each sample by the analytical laboratory.
 - 5.1.1 Determine for which constituents each sample is to be analyzed. Prepare a work order plan and give to water BOA TR. Attachment 16.2.1.3 is a plank work order plan.
 - 5.1.2 The SH shall inform the JEG water Samplers of any changes in site locations, samples to be collected, sample volume, nuclides/analytes, or other information which will enhance water quality determinations.
 - 5.1.3 Group the samples into lots. A lot will consist of: 9 field samples, 4 replicates of one of the field samples, and one known or control sample. As far as it is practicable, lots should be composed of chemically similar samples. At least one out of two lots will contain splits. The TDS content of the known solution should be similar to that of the lot with which it is associated. One blank sample will be prepared for each sit .. Notify the Water Sampling Manager of the samples in _ memo (see Attachment 10.2.1.2 as an example) including any special instructions and/or maps.
- 5.2 The SH or Technical Representative (Tech Rep) shall prepare a work order plan and a requisition, with information from 5.1 above, for review and approval by the Task Manager of Hydrological Services.
 - 5.2.1 Upon approval and signature of the Task Manager, the Tech Rep shall deliver the plan and requisition to the designated Contract Representative for the Basic Ordering Agreement (BOA) Tech Rep.
 - 5.2.2 Prepare UNC work order plan with cover letter to UNC (see Attachment 16, 2.1.4). Determine the desired TDS for each known solution from eitner the site hydrologist or previous analyzation rounds.

PAGE 5 OF 18

- 5.3 The BOA Tech Rep shall inform the analytical laboratory of the:
 - 5.3.1 Planned collection of surface and/or groundwater samples.
 - 5.3.2 Approximate number of surface and/or groundwater samples to be collected and snipped to the laporatory.
 - 5.3.3 Analytes to be determined in each water sample, and
 - 5.3.4 Approximate date samples to be delivered to the laboratory.
- 5.4 The JEG Water Sampling Manager shall perform the following duties:
 - 5.4.1 Determine the water sampling schedule. Advise the Water BOA TR of any changes.
 - 5.4.2 Prevare maps for the water samplers and the Health & Safety Group showing all sample locations. Inform the samplers in writing of the type of samples to be collected at each location. Make sure samplers have all required keys.
 - 5.4.3 Coordinate with UNC the snipping date of the known or control samples to the laboratory.
 - 5.4.4 Provide the water samplers with all equipment and supplies required for the site visit.
- 5.5 The JEG Water Samplers shall prepare for the collection of water samples by performing the following duties.
 - 5.5.1 Inventory the equipment to make sure that required equipment is present. See Attachment 16.2.1.5.
 - 5.5.2 Review data from previous sample collections. For example. obtain previous temperature, conductivity, and pH measurements for use as a field reference when sampling and testing water samples.
 - 5.5.3 Make sure that there is a sufficient quantity of sample bottles in the proper sizes, filters (if required), and pre-addressed lancis.
 - 5.5.4 Make sure that there is a sufficient volume of reagents and/or solutions for preservation, field measurements, or field analysis(es).

6.0 PROCEDURE: COLLECTION OF SURFACE WATER SAMPLES

Surface water samples will be collected by the "grab" method (i.e., random immersion of a clean sample container into a body of surface water

PAGE 5 OF 17

- 4.23 Data Collection Sheets: Water Quality Sampling Record Forms, Chain-of-Custody Forms, and Sample Collection Procedures.
- 5.0 PROCEDURE: Field Measurements of a Collected Water Sample
 - 5.1 Whenever practical, final field measurements are to be taken after at least three (3) borehole volumes have been pumped from the well and as close to the time of sampling as is possible.
 - 5.2 Field measurements to determine temperature, conductivity, and pH are performed on a water sample as it is being collected and are to be recorded on Attachment 16.1.10.2.
 - 5.3 Alkalinity is determined on an aliquot of the raw water sample and prior to shipment of the sample to the laboratory. Titration volumes at a specified pH are observed and recorded on Attachment 16.1.10.2, page 1 of 4. Known alkalinity solution data are observed and recorded on Attachment 16.1.10.3.
 - 5.4 Total Acid determinations are performed on selected groundwater or lysimeter (pore water) samples. This data is recorded in the "COMMENTS" portion of Attachment 16.1.10.2.

6.0 PROCEDURE: Temperature Measurement

A conductivity meter and electrode with a thermometer built into the electrode are strongly recommended for this procedure.

- 6.1 Clean thermometer by washing it in a solution of soap and water. Use a brush to scrub the thermometer. Rinse thermometer well with distilled water and blot dry with lint-free tissue(s).
 - 6.1.1 NOTE: Always blot dry thermometer and electrode with tissue. Water droplets remaining on these probes may dilute or otherwise affect measurement data. Rubbing probes with tissue may result in warming the thermometer or produce an electrostatic charge on the electrode and so directly alter meter readings.
 - 6.1.2 If Conductivity Meter has a thermometer built into the electrode, use instructions of 6.1.
- 6.2 A combination conductivity-temperature probe is used in the field. This thermometer must be checked against a lab grade mercury thermometer at least once a month. At the

PAGE_6___OF__17__

same time, the conductivity meter is also tested for accuracy, using the following procedure.

- 6.2.1 Pour approximately 30 mL of lab prepared KCL solution of 4.3 above into a clean beaker. Place the clean temperature- conductivity probe into the solution. Place the beaker in an ice bath, and using a lab grade mercury thermometer, chili the solution to as near 0° C as possible. Salt should be placed on the ice to reach this low temperature.
- When the KCL solution has reached a maximum of 3" 6.2.2 C, read and record the temperature from the lab thermometer and the meter. Also read and record the conductance.

If meter has an automatic temperature compensator NOTE: (ATC), turn off ATC or turn dial to "0." Do not use ATC for this test.

- 6.2.3 Remove the beaker from the ice bath to allow it to warm. Continue reading and recording temperature pairs as well as conductance at least every 2° C up to 25° C. Remove thermometer from he beaker, clean with distilled water, pat dry with tissue(2) and store.
- 6.2.4 Remove temperature-conductivity probe from beaker, rinse with distilled water, and blot dry with tissues. Pour approximately 20 mL of the other three KCl standards (at room temperature) into four separate clean beakers. Measure and record the conductance of each solution, being sure to rinse and dry the probe between measurements.
- 6.2.5 Remove temperature-conductivity probe from the final beaker, rinse with distilled water, dry with tissue(s) and store.
- 6.2.6 Using the data from 6.2.2, 6.2.3, and 6.2.4, plot lab thermometer temperature versus meter temperature on arithmetic graph paper. Plot meter temperature versus meter conductivity and meter conductivity versus known conductivity on arithmetic graph paper. Submit the data and graphs to the water sampling manager.
- The meter is now ready for measuring temperature 6.2.7 and conductance in the field.



PAGE_7____OF__17

- 6.3 Insert the dry and clean thermometer into the flow-through cell. Allow the temperature to stabilize. Record the temperature measurements periodically on Attachment 16.1.10.2 throughout the time of pumping.
 - 6.3.1 Place the flow-through cell close to the wellhead. Be sure to keep the cell and discharge hose/tube out of direct sunlight.
 - 6.3.2 Use the temperature data to adjust the pH temperature compensator.
- 6.4 When the sample bottle is 1'12, stit off the pump, and carefully withdraw the thermometer f. m the flow-through cell. NOTE: If water sample looks "dirty" and contains res due that will not be removed with distilled water, clean thermometer probe as stated in 6.1.
- 6.5 Rinse thermometer probe with distilled water and blot dry with lint-free tissue(s). The probe is now ready for another water sampling activity or storage.
 - 6.5.1 If the probe is to be stored, place it in its carrying case. Label with "clein", date, and initials.
- 6.6 Clean equipment as described in 8.8 through 8.10 of Section 16.2.1.
- 6.7 If Conductivity, pH, Alkalinity, or Total Acidity determination is required, proceed with 7.0, 8.0, 9.0, or 10.0.
- 6.8 To prepare sample for filtration (if required), preservation (if required), and shipment of samples to the analytical laboratory, proceed with 12.0 of Section 16.2.1.

7.0 PROCEDURE: Conductivity Measurement

A Conductivity Meter and electrode with a thermometer built into the electrode is strongly recommended for this procedure. Because conductivity measurement instruments and instrument operation instructions

vary with each manufacturer, the analyst is advised to use the specific instructions for instrument operations that are provided with the conductivity meter to be used by the analyst.

If not already performed, test conductivity meter as described in Section 6.2.

7.1 Remove electrode and cable from carrying case, rinse well with distilled water, and blot dry with tissue.

PAGE_8___OF_17_

- 7.1.1 Always blot dry electrode with tissue. Water droplet(s) remaining on the electrode may dilute or otherwise affect conductivity measurement(s). Rubbing electrodes with tissue may also produce electrostatic charges on the electrode and thus directly alter meter readings.
- 7.2 Insert clean and dry electrode/cable into the flow-through cell before water sampling is to begin. Read sample conductivity and record value(s) periodically throughout the sampling period and especially just after sampling is completed.
- 7.3 Withdraw electrode and cable, rinse well with distilled water, and blot dry with tissue paper. Electrode/cable now ready for another sampling activity or storage.
 - 7.3.1 Label with "clean", date, and initials if electrode is to be stored.
- 7.4 If pH, Alkalinity or Total Acidity determination is required, proceed with 8.0, 9.0, or 10.0. If pH, Alkalinity or Total Acidity determination is not required, proceed with 12.0 of Section 16.2.1.
 - 7.4.1 To prepare water sample(s) for filtration (if required), preservation (if required), and shipmont to the analytical laboratory, proceed with 12.0 of Section 16.2.1.
- 7.5 Check Attachment 16.1.10.2 to make sure that all of the required sampling data has been recorded.
- 7.6 Proceed with 12.9 of Section 16.2.1 to complete required duties.

8.0 PROCEDURE: pH Measurement

A pH meter and electrode with a temperature compensator and a slope adjustment feature are strongly recommended for this procedure. Because pH measurement instruments and instrument operation instructions vary with each manufacturer, the analyst is advised to use the specific instructions for instrument operations that are provided with the pH meter to be used by the analyst. The meter and electrode can be readily adjusted to the temperature of the sample solution and can be calibrated with pH buffer solutions so that the analyst may easily and quickly work with samples having a pH range of 1-4, 4-7, 7-10, or 10.

8.1 Rinse the pH electrode with distilled water and blot dry with lint-free tissue(s).

ALBUQUERQUE OPERATIONS

PAGE 9 OF 17

- 8.1.1 Always blot dry pH electrode with tissue. Water droplet(s) remaining on the electrode may dilute or otherwise affect measurement data. Rubbing electrodes with tissue may result in producing an electrostatic charge on the electrode and so directly alter the meter readings.
 - 8.1.2 Wher not using pH meter/electrode, always place pH electrode in a container of pH 4 buffer solution and seal with parafilm or its equivalent.
- 8.2 Prior to performing pH measurements in the field or in the lab, at ist once per quarter, test the pH meter and Electrode: Determine Temperature and pH. The description below is for a meter with an Automatic Temperature Compensator (ATC). For meters not equiped with an ATC, the procedure is the same.
 - 8.2.1 Prepare a pH 5.0 buffer solution as described on the pH tablet bottle and label as "pH 5.5 Buffer Solution", add date, and write initials on bottle.
 - 8.2.2 Pour 30 mL of pH 5.0 buffer solution into each of three (3) beakers labelled as "A. H 5.0" (beaker A), "B-pH 5.0" (beaker B), and "C-pi 5.0" (beaker C).
 - 8.2.3 Place beaker A in an ice water bath, beaker 3 on the lab bonch, and beaker C in a hot water bath. Allow beakers 10-15 minutes to acclimate to their media. Swirl beakers to mix solution.
 - 8.2.4 With automatic temperature compensator (ATC) off or dialed to "0", place a clean and dry pH electrode and a clean and dry lab thermometer into beaker A. Read pH to the nearest 0.01 pH units and temperature to the nearest 0.1° C. Record pH and temperature on Attachment 16.1.10.1.
 - Withdraw electrode and thermometer, rinse each with 8.2.5 distilled water, and dry with tissue.
 - 8.2.6 Repeat 8.2.4 and 8.2.5 with beaker B.
 - 8.2.7 Fepeat 8.2.4 and 8.2.5 with beaker C. Store clean electrode as stated in 8.1.2. Store clean thermometer as stated in 6.5.
 - Plot pH and temperature data on semi-log graph paper (pH versus temperature). Attach this data to 8.2.8 Attachment 16.1.10.1.

0 2

JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.10

8.2.9 Unless foregoing data indicates that pH meter/electrode are not operating properly, pH meter/electrode is ready for sample collection trip in the field or laboratory work.

- 8.3 Calibrate the pH meter and electrode with standardized pH buffer solutions. NOTE: See 4.13 to 4.21.
 - 8.3.1 If ph meter has a slops adjustment feature, calibrate pH meter and electrode with buffer solutions which bracket the sample pH.
 - A. Set ATC on pH meter to the temperature of the water sample. Use temperature data from 6.3. Alternatively, if the pH meter does not have an ATC, chill the pH buffer solution(s) to equal the water sample terperature.
 - B. To calibrate pH meter/electrode for pH 4-7 range, in. .t clean and dry pH electrode into a labelled beaker containing a fresh solution of pH 7 buffer. Measure pH. Adjust pH reading to 7.0. Withdraw electrode, rinse with distilled water, and blot dry with tissue. Insert clean and dry pH electrode into a labelled beaker containing a fresh solution of pH 4. Measure pH. Adjust the slope to give a pH reading of 4. Withdraw electrode, rinse with distilled water, and blot dry with tissue.
 - o To calibrate pH meter/electrode for pH 1-4 range, use pH 1 and pH 4 buffer solutions. Use pH 4 buffer solution instead of pH 7 buffer so wion and pH 1 instead of pH 4 as stated in B. alove.
 - o To calibrate pH meter/electrode for pH 7-10 range, use pH 7 and pH 10 buffer solutions. Use pH 7 buffer solution and pH 10 buffer solution instead of pH 4 as stated in B. above.
 - o To calibrate pH meter/electrode for pH 10-12.45 range, use pH 10 and pH 12.45 buffer solutions. Use pH 10 instead of pH 7 buffer solution and pH 12.45 buffer solution instead of pH 4 as stated in B. above.
 - C. Insert clean and dry pH electrode into a labelled beaker containing a fresh solution of one of the buffers which bracket the sample. Measure pH. Withdraw electrode, rinse with distilled water, and blot dry with tissue. Repeat with the other buffer which brackets the sample.

ALBUQUERQUE OPERATIONS

PAGE_11_OF_17

o If pH is within ±0.02 units, depending on certification of pH buffer solution, proceed with 8.4 and with pH measurements of water samples.

o If pH is not within ±0.02 units, depending on certification of pH buffer solution, repeat A, B, and C

until pH meter/electrode gives repetitive readings. Repeat at least twice. If pH meter/electrode does not work satisfactorily, see Troubleshooting Instructions for pH meter and electrode.

- 8.3.2 If the pH meter does not have a slope adjustment feature, set ATC on pH meter to the temperature of the water sample. Use temperature data from 6.3. Alternatively, if the pH meter does not have an ATC, chill the pH buffer solution(s) to be used for the calibration or calibration checks to equal the water sample temperature.
 - A. Insert a clean and dry pH meter into a pH 7 buffer solution. If necessary, adjust pH meter to 7 or pH reading on pH 7 buffer bottle. Record pH meter reading. NOTE: Record all pH readings on Attachment 16.1.10.2.
 - B. Withdraw probe, rinse probe well with distilled water, and blot probe ury with tissue paper.
 - C. Repeat A and B.
 - D. Insert clean and dry pH probe into the sample solution, obtain pH reading, and record pH reading.
 - E. Repeat B.
 - F. Insert clean and dry pH probe into a pH buffer solution that best approximates the pH of the sample solution. Obtain and record pH reading.
 - G. Repeat B.
 - H. Repeat A and B. NOTE: Analyst should determine when and how often 1) pH meter should be recalibrated with pH 7 buffer solution, and 2) pH buffer solution that best approximates the sample pH should be used.
 - I. Analyst should continue with sample(s) until all of pH data has been obtained and recorded. Use A through H.

JEG-AL-QA-21(8/87)

PAGE_12___OF_17_

- 8.4 Insert clean and dry pH electrode into the flow-through cell soon after well pumping has begun. This will eliminate or reduce pH meter drift.
- 8.5 If the pH meter has an ATC, adjust the temperature setting on the pH meter to the temperature of the sample solution. NOTE: Use temperature measurement data from 6.3.
 - Alternatively, if the pH meter has no ATC, reduce 8.5.1 the temperature of the pH buffer solutions when calibrating the pH meter.
- 8.6 Read pH and record measurement within five (5) minutes or after drift has ceased, whichever is sooner. Rinse and blot dry electrode. If possible, perform pH measurement at least three (3) times during the purging of the well.
- 8.7 Periodically during purging, check accuracy of the pH meter with buffer solution(s) that bracket the sample solution and adjust pH meter as needed according to 8.3.1 or 8.3.2.
- 8.8 When purging is completed, withdraw the electrode from the flow- through cell, rinse the electrode with distilled water, and blot dry with tissue(s). Measure the final pH of each of the two (2) bracketing buffers, being sure to clean the electrode between use of the buffer solutions.
 - If the meter is accurate within 0.10 units of each 8.8.1 buffer, record the results on Attachment 16.1.10.2. Proceed to 8.9.
 - If the meter is inaccurate (greater than 0.10 8.8.2 units variance with either buffer) recalibrate the meter. Continue purging and recording pH of the sample for at least four (4) to eight (8) liters of sample. Return to 8.7 and repeat until consistent pH reading within 0.10 units is achieved.
- 8.9 When a sufficient volume of sample has been collected in the sample bottle, remove hoses and pumps from the well hole.
- 8.10 Clean pH electrode: rinse with distilled water and blot dry with tissue. pH electrode now ready for another sampling activity or storage. To store electrode, see 8.1. Label with "clean", date, and initials.
- 8.11 Clean equipment as described in 8.8 through 8.10 of Section 16.2.1.
- 8.12 Check Attachment 16.1.10.2 to make sure that all of the sampling data has been recorded on the form.

PAGE_13____OF_17_

8.13. If Alkalinity or Total Acidity determination is required, proceed with 9.0 or 10.0 respectively.

8.14 To prepare sample for filtration (if required), preservation (if required), and shipment to the analytical laboratory, proceed with 12.0 of section 16.2.1.

9.0 PROCEDURE: Alkalinity Measurement

Alkalinity determination (total cerbonate) is performed on an aliquot of the raw, unfiltered, and unpreserved water sample by titration of the sample with an acid and measured by a pH meter.

- 9.1 Fill in all sample information on Attachment 16.1.10.2, page 3 of 4. NOTE: Record #11 procedure information on Attachment 16.1.10.2, page 3 of 4.
- 9.2 Wash the Erlenmeyer flask(s) and volumetric flask(s), to be used for this determination, with soap and water. Rinse the flasks THREE (3) times with 20 mL of distilled water. Discard rinses. Set flasks upside down to drain. After 10 minutes, remove remaining water droplets by shaking flasks vigorously.
- 9.3 Connect clean pH electrode to the pH meter and rinse electrode well with distilled water and blot dry with tissue.
- 9.4 Calibrate the pH meter/electrode as described in 8.3 and store electrode as described in 8.1.
- 9.5 Without too much splashing or other vigorous agitation of the water sample, carefully pour the water sample into a labelled 100 mL volumetric flask. Add water sample to the 100 mL mark. NOTE: Pour sample down the side of the flask.
- 9.6 With as little agitation as possible, carefully pour the water sample from the flask into a labelled 125 mL Erlenmeyer flask. If available, add a magnetic stirring bar, turn magnetic stirrer on, and stir sample slowly. Record sample aliquot volume on Attachment 16.1.10.2, page 3 of 4.
- 9.7 Insert the pH electrode into the sample and suspend the electrode in the water sample. Adjust the ATC on the pH meter to the temperature of the water sample. Determine sample pH and record value.
- 9.8 Place a 1.6 N H2SO, cartridge in the titrator. Record the titrant lot number of the cartridge (on Attachment 16.1.10.2). Eject a few drops of titrant from the titrator

JEG-AL-QA-21(8/87)

PAGE 14 OF 17

tip, wipe tip with a clean tissue, and reset the counter to "0000".

- 9.9 While gently stirring the sample, titrate the sample with acid by turning the titrator delivery knob. Obtain pairs of pH and alkalinity readings at each pH value listed on Attachment 16.1.10.2, page 3 of 4. NOTE: The alkalinity value at the inflection point between pH 5 and pH 4 will be determined by the Site Hydrologist. The volume in mL equals digital titrator reading divided by 800.
 - 9.9.1 Alkalinity titration endpoints depend upon the total alkalinity concentration. The following endpoints will be used:

Alkalinity	End Point
0-30 mg/L	pH 5.1
31-150 mg/L	pH 4.8
150 mg/L	pH 4.5

- When the titration is complete, check pH meter 9.9.2 calibration (See 8.3). If the meter is accurate within 0.2 units of each buffer value, record the results on Attachment 16.1.10.3. If the meter is inaccurate (greater than 0.2 units of each buffer value), recalibrate the meter and repeat the alkalinity test.
- 9.10 Repeat this procedure with a second Tech Rep performing the titration on a fresh aliquot of water sample.
- 9.11 Calculate the relative error (R.E.) as:

R.E. (%)
$$\frac{X_1 - X_2 \times 100\%}{X_8}$$

Where X, is the first alkalinity measurement at pH endpoint, X2 is the second alkalinity measurement at pH endpoint, and X, is the lesser of X, or X2.

- 9.12 If the percentage error is greater than ten percent (10%), repeat the foregoing procedure until two (2) consecutive measurements are within 10% relative error.
- 9.13 Perform alkalinity determinations on three (3) known solution(s) submitted to JEG by a quality control laboratory and record the pH and alkalinity values on Attachment 16.1.10.3. When performing this procedure, the standard solution identification, site identification, alkalinity test kit identification, PH meter identification, date, and Tech Rep name must be filled out on Attachment 16.1.10.3. Send this attachment to the Quality Control Laboratory

PAGE_15___OF_17__

9.13.1 Send a copy of Attachment 16.1.10.3 to the Data Manager who will record this data on a form similar to Attachment 16.1.10.3 provided by a quality control laboratory.

- 9.13.2 Field alkalinity values and known solution values must agree within 10% of each other or the field test will have to be repeated until the values are accurate.
- 9.14 To prepare the sample(s) for filtration (if required), preservation (if required), and shipment, proceed with 12.0 of Section 16.2.1.
- 9.15 If Total Acidity is required, proceed with 10.0.

9.16 To complete duties, proceed with 12.9 of Section 16.2.1.

10.0 PROCEDURE: Total Acidity

This procedure is performed on selected groundwater or lysimeter (pore water) samples to determine the total acidity of the solutions and to determine the acid-neutralizing potential of carbonate materia's within silty soils or sediments.

Digital Titration with HACH Cartridge

- 10.1 Record site identification, location identification, date, and all titration data under "Comments" on the Soil Water Sampler Data Record, under "Notes" of Attachment 16.1.10.2, or on a separate data sheet.
- 10.2 Calibrate the pH meter/electrode as described in 8.3 and store pH electrode as escribed in 8.1.
- 10.3 Rinse a 100 mL graduated cylinder and a 125 mL Erlenmeyer flask with distilled water. Place cylinder and flask upside down to allow excess water to drain. Remove residual water droplets by vigorously shaking the cylindar and flask. NOTE: If a 100 mL cylinder is unavailable, use a 100 mL volumetric flask.
- 10.4 Carefully pour a 100 mL aliquot of the unfiltered water sample into the graduated cylinder and then carefully pour the sample into the 125 mL titration flask. NOTE: Pour sample down the side of the cylinder and flask.

10.4.1If necessary, use a sample aliquot of less than 100 mL to perform a second titration. NOTE: Two (2) sample aliguots of 50 mL each with titrations on each aliquot are preferred to one (1) sample aliquot of 100 mL and one (1) titration.

PAGE 16 OF 17

- 10.5 Place clean pH electrode into the water sample, add a magnetic stirring har, place sample on a magnetic stirrer. and <u>Gently</u> stir sample. Record the initial sample pH. Withdraw pH electrode, rinse electrode with distilled water and blot dry with tissue.
- 10.6 Place a 1.6 N NaOH cartridge in the titrator. Record the HACH titrant lot number of the NaOH cartridge. Eject a few drops of titrant from the titrator tip, wipe tip with clean tissue, and reset the counter to "0000".
- 10.7 While gently stirring the sample, turn titrator delivery knob and start titration. Record sample pH and corresponding volume of NaOH titrant after every 1-2 turns of the titrator delivery knob (volume in mL = digital titrator reading divided by 800). If pH endpoint of 8.3 is obtained after less than eight (8) turns, then re-do the titration.
 - 10.7.1 Do not use color indicator tablets to determine the pH endpoint.
 - 10.7.2 Use a 0.16 N NaOH cartridge to re-do the titration. Record use of 0.16 N NaOH on the data sheet Form 10.1.
- 10.8 Rinse all glassware with distilled water. Set glassware upside down to allow excess water to drain.
- 10.9 Calculate the total acidity as follows:

Acidity, u equiv/liter
$$\frac{V_t \times N_t \times 10}{V_a}$$

Where $V_t = Volume$, in mL, of NaOH titrant used to obtain pH 8.3, $N_t = Normality$ of NaOH used to titrate the sample, and $V_{\rm m}$ = Volume, in mL, of sample aliquot (should be 50 or 100 mL, unless lysimeter produces very little water). NOTE: Be sure to specify and record which NaOH and Volume of NaOH are used for the titration.

- Repeat the titration procedure with a second Tech 10.10 Rep performing the titration unless there is insufficient sample volume produced by the well or lysimeter.
- 10.11 Calculate the relative error in percent (R.E., %) as previously described in 9.11.
- 10.12 If the relative error is greater than 10%, repeat the procedure a third time, unless there is insufficient sample volume available.

JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.10

PAGE_17___ OF 17_

1

. 16

10.1	13 Check Att required	achment 16.1.10.2 to see the all of the data has been recorded on the re. s.
10.1	14 See 12.9 duties.	of Section 16.2.1 to complete required
11.0	ATTACHMENT (S): Supporting Document(s)
16.1	1.10.1 Equipment	Checklist
16.1	1.10.2 Water Qua	ality Sampling Record Form
16.1	1.10.2 Alkalinit	y Titration Form (page 3 of 4)
16.3	1.10.3 Standard	Alkalinity Test

12.0 REFERENCES

- 12.1 APHA/AWWA/WPCF, 1985. <u>Standard Methods for the Examination of Water and Wastewater</u>, 16th Edition. American Public Health Association, Washington, D.C. 20005.
- 12.2 Korte, N. and P. Kearl, 1984. "Procedures for the Collection and Preservation of Ground Water and Surface Water Samples and for the Installation of Monitoring Wells," U.S. Department of Energy, Grand Junction, Colorado.

	INCOLD RE REVIEW AND APPROVAL
Prepared	By: paletith 8-30-90
Reviewed	By: Jaule to the spino
Reviewed	By: Bul & Beula 9-7-90
Approved	All Anna Anna ManagerDate
	Project Manager/Deputy Project ManagerDate

PROCEDURE DEUTEN AND ADDROUPS

-

13.0

JACOBS ENGINEERING GROUP INC. SECTION_ JE

]

OF

Ö

ATTACHMENT 26 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2					
DH BUFFER	Yes	No	Comments		
sufficient volume 4 sufficient volume 7 sufficient volume 10					
REAGENTS & BOTTLES					
Litmus paper Nitric acid (pres.) Sulfuric acid (pres.) not required * Other reagents not required					
Required sample bottles:		Number of for A1, M Pb210, Ra22	1000 ml plastic bottles 2, Po210, Th230, CN, S, 6, Ra228		
		Number of for Gross A	500 ml plastic bottles lpha-Beta, Nl		
		Number of tottles for	500 ml amber glass		
		Number of bottles for	1000 ml amber glass TOX		
Sufficient bottles	-	-			
Alkaiinity kit ck reagent volumes ck glass for breakage	-	100/06/00005 100/0000			

* If required, must explain under <u>Comments</u>.

JE JACOBS ENGINEERING GROUP INC. SECTION _____

	1.1	ATTACHMENT	16.1.10.1	
		EQUIPMENT	CHECKLIST	
(F111	out all	blanks pri	or to leaving	for field)

	Comments
squeeze bottle	
acid dispensette	
deionized H ₂ O quantity gallons	
distilled H ₂ O gallons	
water level sounder hand tape cloth towels or wipes	batt
ELD FORMS AND MISCELLANEOUS EQUIPMEN	<u>ت</u> :
Expected no. of samples	No. of forms
Clipboard with cover	
Maps marked with well location	5
Well information (completion, dept	h, etc.)
Field instruction book	
Key(s) to well(s)	To be picked up at
Clean pails	Transparent tape
WD-40 (for locks)	Strapping tape (nylon)
Marking pen	
Coolers	
Shipping address of lab(s):	



PAGE____3 CF___3

ATTACHMENT 16.1.10.1 EQUIPMENT CHECKLIST (Fill out all blanks prior to leaving for field)

Power cord

Water hose

Logic unit cord

Comments

Phone numbers and contact:

PUMPS AND FILTERS

Logic unit _____

Regulator

Wrenches _____

Air hose _____

PERISTALTIC PUMP

Battery cable Rechargeable battery _____ Clean? yes ___ no ____ Battery charger _____ Tubing ____ Length ____ Fiîter system: Inspect filter housing for cracks Filter unit leos Number of filters _____ Filter size

ATTACHMENT 16.1.10.2

			JE LACOS	BS ENK	BINEERING GRO	DUP INC			Page 1 of a
		V	VATER QUA	LITY	SAMPLIN	NG RE	COF	D	
SITE	ID:				FINA	L FIELD	VALL	E AT T	HE SURFACE
LOCA	ATION	D:	and the second state of the local difference of the lo						
SAME	PLE ID:			-	pH(SU)				
STAT	IC WA	TER LEVI	EL (FT)		Ec (µmhos/	cm)			
SAMP	LE DE	PTH (FT)			TEMP. (*C)				And a subscription of the subscription of the
SAMP	PLING	DATE			ALKALINITY	(mg/l	CaCO)3) 3 p	H 4.5
SAMP	LING .	TIME:			TITRANT:	1.6 N H	1,50	LOT	#
STA	ART								
cor	MPLETE								FILTER (S) USE
00111	IENTO							Can li	and a subject to be a subject
COMM	NENIS				and a start of the second s				
1000		SCRIPTION		eren in er die ander	And a star of the second star and the second star and the second star star and the second star star star and the				
ALR T	END	C D	O. (air) r	ng/l	Eh soln.(.1M	1)	m \	/ at	. *c
EIEL D	0000	an mar than an ann an thair	a an	in the second		And some of	aitaini		an a manufacture of the second data on a car on
r ib bb	1161150		T. T		Turning	T		<u>г г</u>	
DATE	TIME	JATOT WITH	DRAWN	рH	Ec	TEMP.	Eh	D.O.	COMMENTS
		(Gals)	(Bore Volumes)		(µmnos/cm)	(*¢)	(my)	mgrin	
		0.0	0.0	**	-	-			START PUMPING
		COMPANY OF A COMPANY							
					and an other states and the second states an				Construction of the second
		ar an interaction of							and the second se
		ala kana ka kana kana ka na mana ka							
						1			and which an an and a start of the second start of the
				Record State 2 and 2					
		and a subject of the state of the sub-							an a share an ann an

and the second s



SAMPLE ALIQUOT IS 100 ML UNLESS STATED OTHERWISE.

NOTES

1.44	APL	64.8	kine 1	3.6	a	131	5
MIL	ALA	mp.	IN I	10.	4.4	10,	i de

4 01 4

JE JACOBS ENGINEERING GROUP INC.

	EXH	IBIT	C			
AT	TAC	HME	NT	4		
SUBCONTRACT	NO.	34-	67	05-	S	-

SILE IV:	LOCATION 1D:	SAMPL	E 1D:	
SAMPLE SHIPMENT LIS	I	SAMPLE MATRI	X (Check one)	
L07#		WATER ()	SOIL ()	
SOTTLE 1D AMO A1 A1 - A K1 K2 - A F5 210 Po 210 Po 210 Ra 225 Ra 223	<u> </u>	DOTTLE 1D Th 230 GA/B N1 CN S TOC Alkalinity	<u>AMDUNT</u>	(
ATE SHIFFED:	MET	THOD OF SHIPMENT		
OMMENTS:				
	ubcontracting laborate			-
<pre>be completed by s hereby scknowledge elivery Order No ubcontractor:</pre>	e receipt of the foll Work Ord	lowing on: der Plan dated:	(date)	
To be completed by s I hereby acknowledge Delivery Order No Subcontractor: Name: Title:	e receipt of the foll Work Ord	lowing on: der Plan dated:	(date)	
To be completed by s I hereby acknowledge Delivery Order No Subcontractor: Name: Title: DMMENTS:	e receipt of the foll Work Ord	lowing on: der Plan dated:	(date)	
To be completed by s I hereby acknowledge Delivery Order No Subcontractor: Name: Title: OMMENTS:	e receipt of the foll Work Ord	lowing on:	(date)	
To be completed by s I hereby acknowledge Delivery Order No Iubcontractor: Name: Title: OMMENTS: n accordance with E Vater Quality Analytic	e receipt of the foll Work Ord	lowing on: der Plan dated: he Subcontract :	(date)	

JE		RING GROUP INC.	
S	TANDARD ALKAL	INITY TEST	
STANDARD SOLUTION	KNOWN VALUE (mg/I CaCO ₃)	ALKALINITY	TEST VALUE
		рH	mg/I CaCO3
REQ#			
LAB#			-
REQ			
LAB#		-	-
REQ.#			
LAB#			
SITE ID:		IELD REP ID	
ALKALINITY TEST KIT	ID:		
PH METER D:			
DATE:		SEND TO:	
CONVENTS			

JEG-AL-ENG-50 (9/80)

JE

ALBUQUERQUE OPERATIONS

AL RUOLEROUE OPERATIONS MANUAL	SECTION	16.1.11
ALDOQUERQUE OFERATIONS WANDAL	DATE	02/07/90
SAMPLE COLLECTION FOR ORGANIC SUBSTANCES	EFFECTIVE	04/25/90
	SUPERSEDE	S
	PAGE 1	OF 7

1.0 PURPOSE

To provide a procedure for the collection and shipment of water samples for analysis of organic hazardous constituents listed in Appendix I of 40 CFR 192. To satisfy the requirements of Appendix I, we will analyze for constituents listed in 40 CFR 264, Appendix IX. This procedure is to be used when sampling for dissolved organics, not immiscible phases.

2.0 SCOPE

This procedure is applicable to members of the TAC and its contractors and subcontractors with responsibilities and duties for the collection, preservation, and submittal of groundwater samples for laboratory analysis(es) of Appendix I organic substances. Section 16.1.10 entitled "Field Measurement of Water Samples for Temperature, pH, Conductivity, Alkalinity, and Total Acid" is also applicable to this procedure.

3.0 EQUIPMENT AND SUPPLIES

- 3.1 Bailer(s): use a teflon (tetrafluoroetnylene) bailer that can be completely disassembled for the purpose of cleaning thoroughly. The bailer design should be such that it minimizes loss of volatile organics, and contamination by headspace air during sampling. An example of this type of bailer is shown in A comment 16.1.11.1.
- 3.2 pH Meter and Electrode
- 3.3 Conductivity Meter and Electrode
- 3.4 Thermometer
- 3.5 Volumetric Flask(s), Class A: 250, 500, 1000 mL (Available at Laboratory). Required only if preparing solutions listed in 4.0.
- 3.6 Sample Bottle: use precleaned, oven-dried, certified containers as per EPA specifications.
- 3.7 Sample Bottle for Organic Substances Sampling: Use only glass containers with teflon-lined caps. Precleaned, certified containers will be supplied by the analytical laboratory to the water sampler.
- 3.8 Plastic Sheet(s) and Aluminum foil
- 3.9 Parafilm or its equivalent
- 3.10 Fen, pencil, magic marker, paper, and masking tape

PAGE 2 OF 7

3.11 Rope

3.12 Sampling Data Forms: Water Quality Sampling Record, Chain-of-Custody Forms, etc.

4.0 CHEMICAL AND REAGENTS

NOTE: whenever required by the procedure, JEG purchases and uses prepared, calibrated, and certified solutions from chemical supply firms. The following are listed in the event certified solutions are not readily available or are used whenever procedure parameters do not require certified solutions. In the event that solutions require preparation, they will be prepared at the laboratory before going into the field.

- 4.1 Hydrochloric Acid: HCl. Reagent Grade, 35-38%: ~12 M (~12 N). Purchased from chemical supply firms.
- 4.2 ~2 N HC1: Add 167 mL of concentrated Hydrochloric Acid to a 1000 mL volumetric flask containing 750 mL of distilled water, and mix solution. Add distilled water to the 1000 mL mark, stopper flask, and mix solution well. Store solution in a clean polyethylene bottle labelled with " ~ 2 N HCl", "167 mL HCl/1000 mL distilled water", date, and preparer's initials. Shake solution prior to use.
- 4.3 Sodium Hydroxide: NaOH. Reagent Grade. In pellet form. Purchased from chemical supply firms.
- 4.4 ~1 N NaOH: Dissolve 4.0 grams of Sodium Hydroxide in a 100 mL volumetric flask containing ~75 mL of distilled water, and mix solution to dissolve NaOH pellets. Cool solution in an ice bath. Add distilled water to the 100 mL mark of the flask, stopper flask, and mix solution well. Store solution in a polyethylene bottle labeled with "~1 N NaOH", "4.0 g NaOH/100 mL distilled water", date, and preparer's initials. NOTE: Prepare fresh solution every 30 days and add a new date.
- 4.5 Hexane: Reagent Grade. Purchased from chemical supply firms.
- 4.6 pH paper (1-10): Hydrion or its equivalent.

5.0 PROCEDURE: SAMPLE COLLECTION FOR ORGANIC SUBSTANCES

5.1 Sampling Principles

5.1.1 A consideration in organics sampling is what materials occur in the well from which the samples are obtained. Where PVC well casing is used to construct monitoring wells, casing sections should be connected via screw threading, not with organic cementing compounds, glues, or any solvents. This requirement should be specified in all Scopes of Work for well installation. Also, because the PVC casing may leach or adsorb trace organics, identically constructed PVC wells in

JEG-AL-OA-21 (8/87)

JACOBS ENGINEERING GROUP INC. SECTION.

upgradient background study areas are needed to provide sample "blank(s)" for PVC wells in contaminated areas.

16.1.11 OF 7

- 5.1.2 When sampling for analysis of organic parameters (i.e., TOC, TOX, pesticides, and EPA Appendix IX organics), special procedures and equipment may be required. Many plastics such as PVC and polyethylene adsorb organic contaminants and also desorb other trace compounds from their surfaces. Teflon or stainless steel is preferred for organics sampling. The following should, therefore, be strictly observed and practiced.
 - A. Grab samples to be analyzed for volatile organic acids (VOA) are best obtained by use of a teflon bailer. NOTE: All sample tubing and pump parts should be made of either teflon or approved silicon. See 8.5 for additionalguidance.
 - B. Samples should be refrigerated at the time of collection.
 - C. All organic compounds should be collected in opaque bottles and stored in the dark.
 - D. Specific needs for organic sampling should be determined in consultation with the support laboratory. See Sections 8.2 and 8.5 for additional guidance in sampling for organic substances. See Section 10.0 entitled "Sample Collection with a Bailer" of Section 16.2.1.
 - E. If possible, purge about three (3) pore volumes but not less than one (1) pore volume from well before collecting sample.
 - F. Use only glass or teflon sample containers with teflonlined caps. See Section 3.7.
 - G. Review equipment cnecklist before leaving for the field (Attachment 16.1.11.2).
- 5.2 The site hydrologist (SH) is responsible for contacting the analytical laboratory at least two weeks prior to sampling. The laboratory will provide bottles for organic sampling.
 - 5.2.1 The SH should request the required number of bottle sets and inform the laboratory of the location numbers so the bottles will be properly labeled.
 - 5.2.2 One trip blank per site should also be requested. The bottle sets and trip blank should be shipped either to the UMTRA Project Office and transported to the field from there, or shipped directly to the motel the water samplers are staying at.

- 16.1.11 HAGE 4 OF 7
- 5.2.3 The trip blank will be transported to each of the sampling locations and returned to the analytical laboratory when sampling has been completed.
- 5.3 Unwrap a clean teflon bailer, rinse inside of bailer with distilled water, and discard rinse. Repeat TWICE more with distilled water. Shake bailer vigorously to remove excess water. NOTE: If bailer s not clean, clean bailer as described in Section 5.10.
 - 5.3.1 Rinse outside of bailer several times with distilled water and discard rinse. Dry outside of bailer with tissue(s). Bailer now ready for sampling activity.
- 5.4 Purge well until the temperature, conductivity, and pH of the purged water stabilize. They may not stabilize within 10 percent of initial reading if the initial readings are taken on water that has been standing in the well for a long time. NOTE: To measure and record data for this and other steps of this procedure, use Sections 6.0. 7.0, and 8.0 of Section 16.1.10
- 5.5 After the well stabilizes, slowly submerge the bailer and fill the sample chamber completely. NOTE: Do not agitate water sample by splashing, etc. Volatile organics could escape due to the agitation.
- 5.6 Bring the bailer to the surface and, with a minimum of agitation. slowly drain the water into a labelled sample bottle until it overflows by about 1.5 bottle volumes. If possible, obtain and record temperature, conductivity, and pH of the sample on Attachment 16.1.11.3. NOTE: Drain sample down the side of the container. Avoid direct contact with a dirty surface; always rest the bailer(s) on a clean plastic sheet. Use a new bailer rope for sampling each well to avoid cross contamination.
- 5.7 Immediately cap the bottle, making sure that no headspace or bubbles exist within the bottle.
- 5.8 Place the sample(s) in a cooler that contains "blue ice" or its equivalent and ship to the analytical laboratory immediately. NOTE: Samples must arrive at the laboratory within 24 nours of collection.
 - 5.8.1 Separate field samples into sample lots for packaging and shipment to the analytical laboratory.
 - 5.8.2 A sample lot should consist of no more than nine (9) field samples, four (4) replicates of one of the field samples, and at least one (1) known solution.
 - 5.8.3 Every field sample and known solution will also be split into a sample submitted for lab analysis(es) and a sample to be archived. NOTE: Archived samples are stored by the analytical laboratory and may be analyzed by that laboratory and, at the discretion of JEG, used for re-analysis(es) or other purposes.

- 5.9 Check Attachment 16.1.11.3 to make sure all of the required sampling data has been entered, comments made, any deviation(s) from the sampling procedure recorded, and reason(s) for such deviation(s).
- 5.10 Clean and decontaminate the bailer(s) used for sampling as soon as possible, and do not allow bailer(s) to air dry. NOTE: It is difficult to remove the contaminants adhering to the bailer when it is dry.
 - 5.10.1 Thoroughly scrub and wash the bailer with warm tap water and detergent. Use a brush. Rinse bailer several times with tap water to ensure complete removal of the detergent.
 - 5.10.2 Rinse the bailer at least three (3) times with hexane or ~2 N HCl, followed by at least three (3) rinses with distilled water.
 - 5.10.3 Dry bailer exterior with a lint-free towel. Bailer is now ready for another sampling activity or storage.
 - 5.10.4 If bailer is to be stored, wrap bailer with a clean plastic sheet and seal with tape. Label with "clean", date, and initials.
 - 5.10.5 In cases where gross organic contamination or an immiscible phase is present, it may be necessary to dedicate a bailer to each sampling station.
- 5.11 To prepare the sample(s) for snipment to the analytical laboratory, proceed with 6.0.
- 5.12 Perform the following POST-C. LLECTION DUTLES:

5.12.1 Clean up field equipment.

- 5.12.2 Replace expended items, filters, reagents, reagent solutions, etc.
- 5.12.3 Repair broken equipment or report/request replacement.
- 5.12.4 Make photocopies of all field forms associated with a given sample lot and distribute copies to the Site Hydrogeologist, Data Manager, Water Sampling Manager, and the BOA Tech Rep within two (2) weeks of the date of sample collection. Submit any additional information to the Data Manager.
- 5.12.5 Give original field form(s) containing the sampling information to JEG Document Control.
- 5.13 Notify the Site Manager, by a memorandum, of the collection of the field sampling data and delivery of this information to JEG Document Control.

- 6.0 PROCEDURE: PREPARATION OF WATER SAMPLES FOR PACKAGING, LABELLING, AND TRANSPORT
 - 6.1 Label bottles with the TAC's preprinted labels. Fi i in all appropriate information in the blank spaces.
 - 5.2 Immediately put the samples requiring refrigeration into an insulated cooler containing "blue ice" of ice. Samples which do not require refrigeration should be placed in a box or other suitable container, filled with absorber material, sealed, and shipped to the analytical laboratory. NOTE: Keep <u>all</u> samples out of direct sunlight.
 - 5.3 Put the address and telephone number of the UMTRA Project Office and the name of a JEG contact in the insulated container. Include a list giving the number of samples and sample identifications in each container. Use Attachment 16.1.11.4. This sample list will be completed by the laboratory and sent to the BOA Tech Rep so that records will be kept on the list of samples received. This list can be compared to a duplicate sample list which is retained with the water sampling data to assure that no samples were lost or misplaced during shipment.
 - 6.4 Tape the container with strapping tape for shipping. Inform the laboratory of the sample arrival time and date.
 - 6.5 Schedule sample collection/shipment so that the samples do not arrive at the lab after 4 p.m. on a Friday, or on a Saturday or on a Sunday. Whenever possible, ship the samples so that they arrive at the laboratory no later than 72 hours after they were collected.
 - 6.6 Have the laboratory document sample lot arrival times, and report this information to the BOA Tech Rep.
- 7.0 ATTACHMENT(S): SUPPORTING DOCUMENT(S)
 - 7.1 Schematic Drawing entitled "Point Source Bailer"
 - 7.2 Equipment Checklist
 - 7.3 Water Quality Sampling Record Form--3 Forms
 - 7.4 Sample Receipt Acknowledgment Form
- 8.0 REFERENCES
 - 8.1 Barcelona, Michael, John A. Helfrich, and Edward E. Garske, 1985. "Sampling Tubing Effects on Groundwater Samples." <u>Anal. Chem.</u>, 57: 460-464.

PAGE 7 OF 7

- 8.2 Korte, N. and P. Kearl, 1984. "Procedures for the Collection and Preservation of Groundwater and Surface Water Samples and for the Installation of Monitoring Well." U.S. Department of Energy, Grand Junction, Colorado.
- 8.3 Miller, G. D., 1982. "Uptake and Release of Lead, Chromium, and Trace Level Volatile Organics Exposed to Synthetic Well Casings." Proc. 2nd Nat'l Symp. Aquifer Restoration and Ground Water Monitoring, 1982, 236-245.
- 8.4 Nielsen, D. M. (ed.), 1983. "Chemical Problems in Groundwater Monitoring Programs." Proc. 3rd Nat'l. Symp. Aquifer Restoration and Ground Water Monitoring, 1983: 263-271.
- 8.5 Pettyjohn, Wayne A., W. J. Dunlap, R. Cosby, and J. W. Keeley, 1981. "Sampling Ground Water for Organic Contaminants." Groundwater, 19 (2): 180-189.
- 8.6 USEPA, 1979. Methods for Chemical Analysis of Water and Wastes. EPA-500/4-79-020.
- 8.7 USEPA, 1985. Practical Guide for Ground Water Sampling. EPA 600/2-85/104
- 9.0 PROCEDURE REVIEW AND APPROVAL

Prepared	By:	Tharold W. Too (Harcia W. Tso)
Reviewed	By:	Task Manager
Reviewed	By:	Quarity Assurance Manager
Approved	By:	Project Manager fr 5. HILL

ATTACHMENT 16.1.11.1



POINT SOURCE BAILER allows sampling at specific point within a monitoring well

Threaded and press fitted construction allows for ease of decontamination

٠

.

Additional 1 foot midsection(s) may be added at the center to increase volume

Available in polyvinyl chloride (PVC) or all virgin polytetrafluordethyiene (PTFE), LEXAN[™] or stainless steel

SIZES AVAILABLE



0

¥

0

Ċ

Mark.

25.1

DANE S

75171.06 # 75171.06 #41 MC KODED HERI

> PTFE: 1 inch O D thru 4.5 inches O D. (254 mm - 114.30 mm) Lengths: 2 thru 5 ft

TIMCO[™] Bailers

JE JACOBS ENGINEERING GROUP INC. SECTION

	EQUIPMENT	CHECKLIST	
(Fill out all	blanks pri	or to leaving	for field)

DH BUFFER	Yes	No <u>Comments</u>
sufficient volume 4 sufficient volume 7 sufficient volume 10		Anna Anna Anna Anna Anna Anna Anna Anna
REAGENTS & BOTTLES		
Litmus paper Nitric acid (pres.) Sulfuric acid (pres.) not required * Other reagents not required		
Required sample bottles:		Number of 1000 m1 plastic bott1 for A1, M2, Po210, Th230, CN, Pb210, Ra226, Ra228
		Number of 500 ml plastic bottl for Gross Alpha-Beta, Nl
		Number of 500 ml amber gla bottles for TOC
		Number of 1000 ml amber gla bottles for TOX
Sufficient bottles	-	And and the second second
Alkalinity kit		

* If required, must explain under <u>Comments</u>.

ATTACHMENT 16.1.11.2

JE ALBUQUERQUE OPERATIONS JACOBS ENGINEERING GROUP INC. SECTION ______

OTHER EQUIPMENT	<u>Comments</u>
squeeze bottle	
acid dispensette	
deionized H ₂ O quantity gallons	
distilled H ₂ O gallons	
water level sounder hand tape cloth towels or wipes	battery check
FIELD FORMS AND MISCELLANEOUS EQUIP	MENT:
Expected no. of samples	No. of forms
Clipboard with cover	
Maps marked with well locat	ions
Well information (completion, de	epth, etc.)
Field instruction book	
Key(s) to well(s)	To be picked up at
Clean pails	Transparent tape
WD-40 (for locks)	Strapping tape (nylon)
Marking pen	
Coolers	
Shipping address of lab(s):	
	and a second sector of the second sector is and descend on the data share been a difference of the second sector

JE JACOBS ENGINEERING GROUP INC. SECTION ______

	Comments
Phone numbers and contact:	
	References in the contract of the second sec
	When the second
PUMPS AND FILTERS	
Logic unit	Power cord
Regulator	Logic unit cord
Wrenches	Water hose
Air hose	
PERISTALTIC PUMP	an die west werden der der einen werden einen die der einen die der die der die der die der die der die der der
Battery cable	Rechargeable battery
Clean? yes no	Battery charger
Tubing Length	
Filter system:	
Inspect filter housing for cracks	
Filter unit legs	
Number of filters	
Filter size	

ATTACHMENT 16.1.11.2

3

Ç.
ATTACHMENT 16.1.11.3

			JE JACOS	ES ENG	BINEERING GRC	OUP INC.			Page 1 of				
		M	ATER QUA	LITY	SAMPLIN	IG RE	COR	D					
SITE	ID:				FINAL	FIELD	VALU	EACT	HE SURFACE				
OCA	TION	D:	(mean-mailed me "Million and merry										
AMP	LE ID:			-	pH(S.U.)				and for the state of the distance of the				
STAT	IC WAT	TER LEVE	EL (FT)		Ec (µmhos/c	pm)			Contract designed of the second				
SAMP	LE DE	PTH (FT)			TEMP. (°C)								
AMP	LING D	ATE			ALKALINITY	(mg/1	CaCO	3) @ pH 4.5					
SAMP	LING T	IME:			TITRANT:	1.6 N H	2500	LOT	#				
STA	RT												
001	APLETE				-				FILTER(S) USI				
	time? We down in a		adamentary county, "an manager, and										
NNIOC	ENTS												
			articul constants in the second second						enterneter in the entertainty in the second states				
OCAT	TION DE	SCRIPTION	1										
AIR T	EMP.	0 D.	0, (air)	ma/I	Eh soin.(.11	1	mV	at	•0				
FIELD	REP(S).		THE REPORT OF THE PARTY OF THE	Contract + House - And	the state of the state of the state of the location	and day a court form	and delay we do not		The second s				
FIELD	REP(S)	TOTAL	VOLUME	рH	Ec	TEMP.	Eh	5.0.					
TELD	TIME	TOTAL	VOLUME DRAWN	рH	Ec (µmhos/cm)	TEMP. (*C)	Eh (mV)	D.O. mg/l	COMMENTS				
ATE	TIME	TOTAL WITH (Gals) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	рH	Ec (µmhos/cm)	TEMP. (*C)	Eh (mV)	D. O. mg/i	COMMENTS START PUMPING				
ATE	TIME	TOTAL WITH (Gals) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	рH	Ec (µmhos/cm) -	TEMP. (°C)	E.h (m.V)	D.O. mg/l	COMMENTS START PUMPING				
TELD	TIME	TOTAL WITH (Gals) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	рH	Ec (µmhos/cm) -	TEMP. (*C)	E.h (m.V)	D.O. mg/l	COMMENTS START PUMPING				
FIELD ATE	TIME	TOTAL WITH (Gals) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	ρH	Ec (µmhos/cm) -	TEMP. (°C)	Eh (mV)	D. O. mp/l	COMMENTS START PUMPING				
ATE	TIME	TOTAL WITH (Gals) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	PH -	Ec (µmhos/cm) -	TEMP. (*C)	Eh (mV)	D, O, mg/l	COMMENTS START PUMPING				
ATE	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	Hq 	Ec (µmhos/cm) -	TEMP. (°C)	En (mV)	D, O. mg/l	COMMENTS START PUMPING				
DATE	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	н -	Ec (µmhos/cm) -	TEMP. (°C)	Eh (mV)	D.O. mg/l	COMMENTS START PUMPING				
DATE	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	P H	Ec (µmhos/cm) -	TEMP. (°C)	Eh (mV)	D.O. mg/l	COMMENTS START PUMPING				
DATE	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	H q	Ec (µmhos/cm) -	TEMP. (°C)	Eh (mV)	D.O. mp/l	COMMENTS START PUMPING				
ATE	TIME	TOTAL WITH (Gals) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	pH -	Ec (µmhos/cm) -	TEMP. (*C)	En (mV)	D, O, mg/l	COMMENTS START PUMPING				
	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	pH =	Ec (µmhos/cm) -	TEMP. (*C)	En (mV)	D, O. mp/l	COMMENTS START PUMPING				
DATE	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	pH 	Eo (µmhos/om) -	T E M.P. (* C)	£h (mV)	D. O. mp/l	COMMENTS START PUMPING				
PIELD	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	p H	Ec (µmhos/cm) -	T E M.P. (° C)	Eh (mV)	D.O. mp/i	COMMENTS START PUMPING				
DATE	TIME	TOTAL WITH (Gais) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	p H	Ec (µmhos/cm) -	T E M.P. (* C)	Eh (mV)	D, O, mg/l	COMMENTS START PUMPING				
DATE	TIME	T O T A L WITH (G & Is) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	pH	Ec (µmhos/cm) -	T E M.P. (* C)	£h (mV)	D, O, mp/l	COMMENTS START PUMPING				
	TIME	T O T A L WITH (G & Is) 0.0	VOLUME DRAWN (Bore Volumes) 0.0	pH 	Ec (µmhos/cm)	T E M.P. (* C)	£ h (m V)	D, O. mp/l	COMMENTS START PUMPING				

JEG-AL-ENG-44(2/90)

-

·

(

SITE ID EPTH TO WATER (h2) (FT.) EPTH OF WELL (h1) (FT.) VELL DIA (d) (FT.) ORE VOL. (FT.) ³ CREENED INTERVAL (FT.) CREENED INTERVAL (FT.) ATE OF LAST EC CALIBRATION IME OF PH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	LOCATION ID
EPTH TO WATER (h2) (FT.) EPTH OF WELL (h1) (FT.) VELL DIA (d) (FT.) ORE VOL. (FT.) ³ ORE VOL. (FT.) ³ CREENED INTERVAL (FT.) ALIBRATION INFORMATION ATE OF LAST Ec CALIBRATION IME OF pH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	WITHDRAWAL METHOD
EPTH OF WELL (h1)(FT.) (ELL DIA (d) (FT.) ORE VOL. (FT.) ³ CREENED INTERVAL (FT.) CREENED INTERVAL (FT.) ATE OF LAST EC CALIBRATION IME OF PH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	FILTER SIZE45µ THERMOMETER ID EC METER ID PH METER ID PUMP ID Eh METER ID D.O. METER ID FOR STANDARD PH FOR STANDARD PH
ALIBRATION INFORMATION ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF pH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	THERMOMETER ID EC METER ID PH METER ID PUMP ID EN METER ID D.O. METER ID D.O. METER ID FOR STANDARD PH
ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF PH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	EC METER ID PH METER ID PUMP ID Eh METER ID D.O. METER ID D.O. METER ID FOR STANDARD PH FOR STANDARD PH
ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF PH CALIBRATION HAFTER MEASUREMENT HAFTER MEASUREMENT	PUMP ID PUMP ID Eh METER ID D.O. METER ID D.O. METER ID FOR STANDARD PH FOR STANDARD PH
ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF PH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	PUMP ID Eh METER ID D.O. METER ID D.O. METER ID FOR STANDARD DH FOR STANDARDH
ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF pH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	EN METER ID D.O. METER ID FOR STANDARD DH FOR STANDARDH
ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF pH CALIBRATION HAFTER MEASUREMENT HAFTER MEASUREMENT	D.O. METERID FOR STANDARD DH FOR STANDARDH
ALIBRATION INFORMATION ATE OF LAST EC CALIBRATION IME OF pH CALIBRATION HAFTER MEASUREMENT HAFTER MEASUREMENT	FOR STANDARD PH FOR STANDARDH
ATE OF LAST EC CALIBRATION IME OF PH CALIBRATION H AFTER MEASUREMENT H AFTER MEASUREMENT	FOR STANDARD PH FOR STANDARD _H
IME OF PH CALIBRATION HAFTER MEASUREMENT HAFTER MEASUREMENT	FOR STANDARD PH
HAFTER MEASUREMENT	FOR STANDARD DH
HAFTER MEASUREMENT	FOR STANDARD _H
HIPPING INFORMATION	
AB(S) SHIPPED TO:	
ATE(S) SHIPPED:	
AETHOD OF SHIPMENT:	
NOTES	

.

i.

JEG-AL-ENG 44 (8/87)

ð -

ہ م ا 國



1000

NOTES

JE JACOBS ENGINEERING GROUP INC.

4

-

4 01 4

		an a				
SITE 10:	L	OCATION ID:	S.KMPL	.E 1D:		
SAMPLE SHIPH	ENT LIST		SAMPLE MATR	IX (Check of	ne)	
LOT#	-		WATER ()	SOIL ()	
DOTTLE	AMOUNT	<u>REC'D</u> (~)	BOTTLE	AMOUNT	RECID	(~)
A1 A1-A M1	-	{ }	Th 230 GA/B N1		{ }	
M1-A M2		{ }	C N S	-	$\{ \}$	
M2-A Fb 210 Po 210 Ra 225 Ra 223			TOC Alkalinity		{ }	
DATE SHIPPED):	MET	HOD OF SHIPMEN	1:		
COMMENTS:						_
To be comple	eted by subcor	ntracting laborato				-
I hereby acl	knowledge rece	tipt of the foll	owing on:			
Delivery Ord	der No	Work Ord	er Plan dated:	(date)		_
Subcontracto Nar Tit	pr: ne: le:					
COMMENTS:						

JEG-AL-ENG-44 (1/90)

1

1

	JE ALAUGULPEUL OFICATION	1
ACKN	OWLEDGEMENT	OF RECEIPT OF SAMPLES
	FOR ORGA	NICS ANALYSIS
SITE ID	LOCATION ID_	SAMPLE NO.
BOTTLE	REC'D(V)	DELIVERY ORDER NO
		DATE SHIPPED
I GAL. AMBER	and the second second	METHOD SHIPPED
VOA VIAL #1	annersenaer	COMMENTS
VOA VIAL #2		man a substant of the same a substant over a substant of the substant of the substant of the substant of the substant over the substant of the
SITE ID	LOCATION ID_	SAMPLE NO
BOTTLE	REC'D(/)	DELIVERY ORDER NO
BOTTLE	REC'D(/)	DELIVERY ORDER NO
BOTTLE 1 GAL. AMBER	<u>REC'D(/)</u>	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2	<u>REC'D(/)</u>	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2	<u>REC'D(/)</u>	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2 I HEREBY ACH	REC'D (/)	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS EIPT OF THE ABOVE SAMPLE(S) ON
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2 I HEREBY ACH	REC'D (/)	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS EIPT OF THE ABOVE SAMPLE(S) ON
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2 I HEREBY ACH	REC'D (/)	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS EIPT OF THE ABOVE SAMPLE(S) ON
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2 I HEREBY ACH	REC'D (/)	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS EIPT OF THE ABOVE SAMPLE(S) ON CONTRACTOR ORATORY
BOTTLE 1 GAL. AMBER VOA VIAL #1 VOA VIAL #2 I HEREBY ACH	REC'D(/)	DELIVERY ORDER NO DATE SHIPPED METHOD SHIPPED COMMENTS EIPT OF THE ABOVE SAMPLE(S) ON CONTRACTOR ORATORY E



JE JACOBS ENGINEERING GROUP INC.

ALPHOLIEDOLIE OPERATIONS MANUAL	SECTION 16.1.15
ALBOQUERQUE OPERATIONS WANDAL	DATE 8-29-90
	EFFECTIVE 8-30-90
NEUTRON PROBE ACCESS HOLE CLOSURES	SUPERSEDES
	PAGE 1 OF 3

1.0 PURPOSE

To provide guidelines for neutron probe access hole abandonment and closures on UMTRA Project sites.

2.0 SCOPE

The scope of this procedure will encompass only the neutron access holes that are located on DOE UMTRA sites and are installed on remediated uranium mill tailing piles.

3.0 AUTHORITY AND NOTIFICATIONS

- 3.1 The DOE shall have the ultimate authority to designate abandonment and clusure of neutron probe access holes. Recommendation to abandon and close these access holes shall be made by the technical staff in writing through the TAC Project Manager to the DOE and only in cases where data gathering activitie are determined to be complete or no longer necessary.
- 3.2 When DOE has authorized the avandonment and closure of neutron probe access holes, the TAC shall take the necessary steps to ensure that abandonment and closure of access holes are appropriately performed in accordance with the prescribed methods listed below and related standard operating procedures.
- 3.3 The TAC shall notify all state and tribal government agencies in addition to the Nuclear Regulatory Commission and Department of Interior, as appropriate.

4.0 ENVIRONMENTAL, HEALTH AND SAFETY

4.1 All reasonable precautions shall be taken in the performance of this procedure to protect the environment and assure the health and safety of employces and the public. Work under this procedure shall be conducted in compliance with the established health and safety regulations of the Occupational Safety and Health Administration (OSHA), the Department of Energy (DOE), and other applicable Federal, state, and local regulations and requirements. Because this procedure involves excavation, particular attention is called to the recently amended OSHA rule on excavations (29 CFR 1926, Subpart P). The sides of any excavation will need to be adequately sloped or shored. The recommended means of cutting off the tube is the use of a pipe cutter, if feasible.



- 4.2 The personnel performing this procedure shall attend a fieldwork briefing before starting any work and shall use the following personnel protective equipment:
 - o Hard hat
 - o Safety boots
 - o Eye protection
 - o Hearing protection (if noise generating equipment is used)
 - o Coveralls
 - o Work gloves
- 4.3 Adequate fire extinguishers and first aid kits shall be available at the work site.
- 4.4 Accidents and/or injuries shall be reported to the TAC Environmental, Health and Safety Manager promptly. A verbal report by phone should be made as soon as possible, but in any event within 24 hours. Written reports of incidents resulting in lost work time or medical treatment shall be submitted within 48 hours after the complete information on the incident is known.

5.0 PROCEDURE

The Procedure to be used to close the hole shall be in accordance with the details provided in Figure 1 and in the following steps. Please note that in some instances a protective casing may not have been installed. Therefore references to the protective casing can be ignored when implementing the procedure.

- 5.1 Fill access tube with bentonite chips to the top of the radon barrier and add water to hydrate chips. One hour is required for complete hydration of chips.
- 5.2 Excavate access pit by hand using shovels, picks, and buckets. Ensure access pit sidewalls have low enough angle to avoid slumping and caving. Access pit must be of sufficient size to allow adequate working space. Stockpile excavated material into separate piles for easy reuse. Use buckets or plastic sheeting to contain and separate excavated material. The bottom of the access pit is to be excavated two to three inches into the filter bedding material. Do not excavate pit below the base of the filter bedding material. Be sure to stabilize protective casing and inner access tube during removal of material by attaching a durable guideline(s) for support.
- 5.3 Cut protective casing and inner access tube using torch or cutoff saw as close as possible to the bottom of the access pit. While cutting the casing and tube, lateral tension must be applied to the guideline(s) to prevent the protective casing and inner access tube from collapsing. As the cutting proceeds, the upper portion



(protective casing/inner access tube) can be slowly lowered to one side of the access pit using the guidelines. Take particular care to avoid disturbance to cover components.

- 5.4 Remove the upper portion from the access pit. Depending on the installation procedure employed, this upper portion may be extremely heavy and cumbersome. Therefore, take necessary safety precautions to avoid undue risk to personnel.
- 5.5 Inspect the cement grout anchoring the protective casing into the radon barrier to ensure it is completed to the top of the radon barrier. If necessary add bentonite-cement grout to achieve the desired level. At least 24 hours are required for the cement grout to cure. During the curing process inspect and add as required additional grout to compensate for any settling and cracking that may have occurred.
- 5.6 Place additional bentonite chips in the inner access tube bringing the level of chips up to within 0.2 feet of the top of the tube and hydrate with water. After one hour is allowed for hydration place clean sand, comparable to the filter bedding material, within the annulus between the protective casing and the inner access tube. Fill the entire annulus and allow the clean sand to fill the 0.2 feet in the top of the inner access tube.
- 5.7 Replace filter bedding material and recompact material as close as possible to design specifications for moisture content and dry density.
- 5.8 Finally, replace rock riprap cover carefully to avoid damage to filter bedding materiai.
- 5.9 If any additional material is needed for the two layers excavated, the material must be as close as possible to design specifications.

6.0 ATTACHMENT

A schematic diagram is attached showing the idealized specifications for neutron probe access hole installation. The diagram is provided to assist in the abandonment procedure.

Prepared by: Revise Louborth -Reviewed by: Janua Task Mahager Reviewed by: 006 Repell Quality Assurance Manager Albert - this Approved by:-Project Manager





JE JACOBS ENGINEERING GROUP INC.

ALBUQUEBOUE OPERATIONS MANUAL	SECTION 16.2.1
ALDOGOLINGOL OF LITATIONS WANDAL	DATE 02/07/90
SAMPLE COLLECTION, PRESERVATION,	EFFECTIVE 06/05/90
AND SHIPMENT OF WATER SAMPLES	SUPERSEDES 08/31/87
	PAGE 1 OF 18

1.0 PURPOSE

To provide a procedure by which field programs will conduct and document the collection of water samples for water quality determinations.

2.0 SCOPE

This procedure is applicable to members of the TAC and its contractors and subcontractors with responsibilities and duties for water quality determinations and for the collection, preparation, preservation, and submittal of surface water and groundwater samples for laboratory analysis(es). Appropriate portions of Section 16.1.10 as stated herein are also applicable to this procedure.

3.0 EQUIPMENT AND SUPPLIES

- 3.1 Pumps
- 3.2 Hoses/Tubes
- 3.3 Filter Apparatus
- 3.4 Filters
- 3.5 Bailer(s): Use a teflon (tetrafluoroetnylene) bailer for sampling organics that can be completely disassembled for the purpose of cleaning thoroughly. The bailer design should be such that it minimizes loss of volatile organics, and contamination by headspace air during sampling. An example of this type of bailer is shown in Attachment 16.1. A PVC bailer may be used for sampling inorganics.
- 3.6 Flow-Through Cell
- 3.7 Volumetric Flask(s), Class A: 250, 500, 1000 mL (Available at Laboratory) Required only if preparing solutions listed in Section 4.0.

1884

- 3.d Sample Bottle: Carboy or its equivalent
- 3.9 Sample Bottle for Organic Substances Sampling: Use only glass containers with teflon-lined caps. Precleaned, certified containers will be supplied by the analytical laboratory to the water sampler.

JE JACOBS ENGINEERING GROUP INC. SECTION 16.2.1 ALBUQUERQUE OPERATIONS PAGE 0 OF 18

> to collect a full container of water). Criteria for exact location and number of grab samples should be based on the known tendency of wastes to mix slowly in surface water. See 17.2. Sample locations unlikely to be contaminated, situated upstream from contaminated sites, or classified as "Background Samples" should be collected <u>first</u> in order to prevent cross-contamination of the water samples.

- 6.1 Use a carboy for a sample container. Immerse the carboy into the surface water and fill the carboy with water to 1/10 capacity. NOTE: If the surface water is frozen, <u>do not</u> collect ice samples in lieu of water samples.
- 6.2 Swirl and rinse the carboy interior with the water. Go downstream about 5-10 feet and discard the water. Return upstream to the original sampling location. NOTE: For a lake or other standing surface water, discard the water at a site away from the sampling location.
- 5.3 Repeat Steps 5.1 and 5.2 TwICE more with fresh water.
- 6.4 Fill the carboy with raw water sample and mix the sample well. Obtain temperature, conductivity, and pH data on the sample solution as described in 6.0, 7.0, and 8.0 of Section 16.1.10. Record data on Attachment 16.2.1.6.
- 6.5 If alkalinity determination is required, carefully pour an aliquot of the sample from 6.4 down the side of and into a sample bottle labelled with "For Alkalinity Analysis". Save bottle for 5.9.
 - 6.5.1 If alkalinity determination is not required, proceed directly with 6.6.
 - 6.5.2 Record site location, sample number, sample volume, date collected, and collector name on Attachment 16.2.1.6.
- o.o If sample is to be filtered, preserved (if required), and shipped to an analytical laboratory, proceed with 12.0.
- 6.7 Clean and store thermometer, and conductivity and pH electrodes as described in 6.5, 7.3, and 8.1 of Section 16.1.10.
- 6.8 Check Attachment 16.2.1.6 to make sure that all of the required sampling data has been entered on the form.
- 6.9 If alkalinity determination is required, take sample bottle from 6.5, and proceed with 9.0 of Section 16.1.10.
- 6.10 Proceed with 12.9 to make sure all of the required duties are completed.

JACOBS ENGINEERING GROUP INC. SECTION 15.2.1 ALBUQUERQUE OPERATIONS

PAGE 7 OF 18

7.0 PROCEDURE: GENERAL GROUNDWATER SAMPLING RULES

- 7.1 Collect sample first from wells which are likely to be uncontaminated. Then collect samples from wells which could be contaminated.
- 7.2 Take water from domestic wells as close to the wellhead as is practical and downhole, if possible. NOTE: Do not take a sample downstream of any device which alters water quality (e.g., a water softener).
- 7.3 Whenever there is sufficient wellwater, pump water from the well and purge at least three (3) well volumes. Record on Attachment 16.6 when this purging cannot be performed.
- 7.4 Record any odors, colors, etc. present in the water sample on Attachment 16.2.1.6. Mark well location on a map. NOTE: when sampling domestic wells, request and record information regarding well depth, completion interval, use, diameter, etc.
- 7.5 Record any deviation(s) from or modification(s) to these general procedures as well as the rationale for these changes.

8.0 PROCEDURE: DIRECT SAMPLING OF A GROUNDWATER SAMPLE

A water sample is pumped directly from the well to a proper sample container. Dissolved constituent samples are filtered through a 0.45-mm filter and total constituent samples are not filtered.

- 8.1 Unwrap clean hose and pump. Connect hose to the clean pump.
- 8.2 Place moses and downhole pump into the wellhole. Connect a flow-through cell and place outlet hose into a clean container to allow measure ant of flow rate. Insert thermometer, conductivity, and pH electrodes into the flow-through cell. NOTE: Position pump intake a few feet above the screened interval and as near as possible to the well screen.
- 8.3 Start pump and begin purging well. Perform periodical temperature, conductivity, and pH measurements on water in flow-through cell as described in 6.0, 7.0, and 8.0 of Section 16.1.10 and record data on Attachment 16.2.1.6. These data are used to verify that temperature, pH, and conductivity have stabilized prior to sample collection. NOTE: Adjust pH temperature compensator, if necessary.
- 8.4 Begin collecting the wellwater sample after the temperature. conductivity, and pH of the discharging water has stabilized (i.e., within 0.1 units, for at least 10 minutes). Collect the required amount of sample. Close bottle and mix the water sample well.

JACOBS ENGINEERING GROUP INC. SECTION ______ 10.2.1 ALBUQUERQUE OPERATIONS

PAGE 8 OF 18

- 8.4.1 Collect the unfiltered water samples upstream of the flow-through cell and preserve and store as required. See 12.0 and 15.0.
- 8.4.2 Collect the filtered water sample upstream of the flow-through cell. See Section 12.0.
- 8.5 If an alkalinity test is required, obtain an unfiltered sample upstream of the flow-through cell. Rinse the volumetric flask two times with well water, obtain a fresh sample, and proceed with 9.0 of Section 16.1.10.
- 8.6 Check Attachment 16.2.1.6 to make sure that all sample collection data has been entered on the form and that all sample bottles are appropriately labelled.
- 8.7 Clean and store thermometer, conductivity, and pH electrodes as described in 6.5, 7.3, and 8.1 of Section 16.1.10.
- 3.8 Dissassemble sampling apparatus. Drain hoses and pumps to ensure that all of the previous water sample is expelled. Clean hoses and pumps.
 - 8.8.1 Pump 2000 mL of distilled water through the hoses and the pump. Continue pumping until all of the distilled water is expelled. Clear outside of noses/pumps with distilled water and dry the noses/pumps with tissues. The hoses/pumps are now ready for another sampling activity or storage.
 - 8.8.2 If noses and pumps are to be stored, place unit in a clean plastic pag and seal with tape. Lapel with "clean", date, and initials.
- 8.9 Clean flow-through cell with distilled water and discard rinse. Set cell to drain and snake cell vigorously to remove water. The cell is now ready for another sampling activity or storage.
 - 8.9.1 If the cell is to be stored, wrap unit in a clean -lastic bag and seal with tape. Label with "clean", date, and tials.
- 8.10 If a filter has been used, remove filter from holder and discard. Clean apparatus several times with distilled water. Make sure inlet and outlet openings are well rinsed. Rinse outside of filter apparatus and dry with tissue. Filter apparatus is now ready for a fresh filter and another sampling activity or storage.
 - 8.10.1 If apparatus is to be stored, wrap unit in a clean plastic bag and seal with tape. Label with "clean", date, and initials.

JACOBS ENGINEERING GROUP INC. SECTION 16.2.1 ALBUQUERQUE OPERATIONS

PAGE 9 OF 18

9.0 PROCEDURE: GROUNDWATER SAMPLING FROM A PRODUCTION WELL

Raw samples from production wells will be collected by the "grab" sample method from public, private, and TAC water supply wells during pump tests or regular pumping.

- 9.1 Unwrap and connect clean hose(s) to the clean pump(s).
- 9.2 Connect one end of the sampling hose(s)/pump(s) to the well and the other end to a flow-through cell. Place cell outlet hose into a clean container to measure flow rate. Check all connections to make sure there are no leaks. Insert thermometer, conductivity, and pH electrodes into the flow-through cell.
- 9.3 Open flow regulator to allow water to pass through flow-through cell. Periodically, perform temperature, conductivity, and pH measurements as described in 6.0, 7.0, and d.0 of Section 16.1.10 and record data on Attachment 16.6. This data is used to verify that temperature, pH, and conductivity have stabilized prior to sample collection. NOTE: Adjust pH temperature compensator, if necessary.
- 9.4 Begin collecting the water sample after the temperature, conductivity, and pH of the discnarging water has stabilized (i.e., within 0.1 units, for at least 10 minutes). Collect the required amount of sample. Close bottle and mix the water sample well,
 - 9.4.1 Collect the unfiltered water samples upstream of the flowthrough cell and preserve and store as required. See 12.0 and 15.0.
 - 9.4.2 Collect the filtered water sample upstream of the flowthrough cell and proceed with 12.0.
- 9.5 If an alkalinity test is required, obtain an unfiltered sample upstream of the flow-through cell. Rinse the volumetric flask two times with well water, obtain a fresh sample, and proceed with 9.0 of Section 16,1.10.
- 9.6 Cneck Attachment 16.2.1.6 to see that all of the required sampling data has been entered.
- 9.7 Clean and store thermometer, conductivity, and pH electrodes as described in 6.5, 7.3, and 8.1 of Section 16.1.10.
- 9.8 Clean filter apparatus: remove filter paper and discard (if required), wash filter apparatus with detergent and water, and rinse well with distilled water. Dry filter apparatus with tissue. Filter apparatus now ready for another sampling activity or storage. NOTE: Be sure inlet/outlet hoses are also cleaned and rinsed.

JACOBS ENGINEERING GROUP INC. SECTION 16.2.1

- PAGE 10 OF 18
- 9.8.1 If filter apparatus is to be stored, wrap unit with clean plastic bag and seal with tape. Label with "clean", date, and initials.
- 9.9 Clean flow-through cell: rinse well with distilled water. Be sure to rinse inlet and outlet hoses well with distilled water. Shake cell vigorously to make sure that all of the excess water has been expelled. Cell now ready for another sampling activity or storage. NOTE: If cell has been discolored by water substance(s), wash cell with soap and water, and rinse well with distilled water.
 - 9.9.1 If cell is to be stored, wrap cell with a clean plastic bag and seal with tape. Label with "clean", date, and initials.

10.0 PROCEDURE: GROUNDWATER SAMPLING WITH A BAILER

Water samples should only be collected with a bailer whenever it is impossible to pump from the well. Bailers made of tetion, PVC, or stainless steel should be used whenever possible.

Samples to be collected and analyzed for EPA Appendix VIII/IX organic substances may preferably be collected by the bail method. See Section 16.1.11. Teflon is preferred for collection of samples containing organics.

- 10.1 If the bailer requires cleaning, clean bailer as described in 10.14.
 - 10.1.1 NOTE: If pailer has previously been washed, rinse bailer thoroughly with distilled water. Drain and discard rinse. Repeat TWICE more with distilled water.
 - 10.1.2 Rinse outside of bailer with distilled water. Dry outside of bailer with tissues. The bailer is now ready for sample collection.
- 10.2 Lower a clean bailer into the well and fill bailer 01/3 full with water. NOTE: Omit 10.2 and 10.3 if well volume is insufficient for rinses. Proceed directly with 10.4.
- 10.3 Retrieve bailer, mix water in bailer, and discard water at a site away from the well. Repeat TWICE more with fresh wellwater.
- 10.4 Lower bailer into the well, fill bailer with water sample, and retrieve bailer. NOTE: Avoid direct contact with a dirty surface; always rest bailer(s) on a clean plastic sheet. Use a new bailer rope for sampling each well.

JACOBS ENGINEERING GROUP INC. SECTION 16.2.1 ALBUQUERQUE OPERATIONS

PAGE 11 OF 18

- 10.4.1 If pailer is used because well is "unpumpaple," proceed with 10.7 and then to 10.14. Omit all other steps.
- 10.5 Drain bailer into a clean sampling container such as a carboy. Also, fill the flow-through cell with sample water. Unwrap and connect clean hose and pump. Place one end of hose into carboy and remaining end of nose into a sample bottle. Check connections to make sure that they are tight. Insert clean thermometer, conductivity, and pH electrodes into the flow-through cell and tighten connections.
- 10.6 Perform temperature, conductivity, and pH measurements on each carboy as described in 6.0, 7.0, 8.0 of Section 16.1.10 and record data on Attachment 16.2.1.6.
- 10.7 Pump the unfiltered water samples from the carboy into the sample bottles and preserve and store as required. See 12.0 and 15.0.

10.7.1 Next collect the filtered water sample according to 12.0.

- 10.8 If an alkalinity test is required, obtain an unfiltered sample from the carboy. Rinse the volumetric flask two times with well water and optain a fresh sample and proceed with 9.0 of Section 16.1.10.
- 10.9 If the uncollected sample is insufficient, repeat 10.4 and until the required sample volume has been collected.
- 10.10 Check Attachment 16.2.1.6 to see that all of the sampling data has been entered.
- 10.11 Clean and store thermometer, conductivity, and pH electrodes as described in 6.5, 7.3, and 8.1 of Section 16.1.10.
- 10.12 Clean filter apparatus several times with detergent and water and rinse with distilled water. Rinse outside of filter apparatus with distilled water and dry with tissue. The filter apparatus is now ready for another sampling activity or storage.
 - 10.12.1 If filter apparatus is to be stored, wrap unit with a clean plastic sheet and seal with tape. Label with "clean", date, and initials.
- 10.13 Disassemble sampling apparatus, drain hoses/pumps to ensure that all of the previous water sample is expelled, and clean moses and pumps.
 - 10.13.1 $Pump \sim 2000$ mL of distilled water through the hoses and pumps. Continue pumping until all of the distilled

JACOBS ENGINEERING GROUP INC. SECTION

SECTION 16.2.1 PAGE 12 OF 18

water is expelled. Clean outside of hoses/pumps with distilled water and dry with tissues. The hoses/pumps are now ready for another sampling activity or storage.

- 10.13.2 If noses/pumps are to be stored, wrap unit in a clean plastic sheet and seal with tape. Label with "clean", date, and initials.
- 10.14 If possible, disassemble bailer and wash with detergent and warm water. Use a brush. Rinse well with distilled water to remove detergent.
 - 10.14.1 Rinse bailer at least three (3) times with hexane or ~ 2 N HCl, followed by at least three (3) rinses with distilled water.
 - 10.14.2 Reassemble bailer and rinse again with distilled water. Rinse outside of bailer with distilled water and dry with tissue. Snake bailer vigorously to remove water. Bailer now ready for another sampling activity or storage.
 - 10.14.3 If bailer is to be stored, wrap unit with a clean plastic sheet and seal with tape. Label with "clean", date, and initials.

11.0 PROCEDURE: GROUNDWATER SAMPLING FROM SEEPS AND SPRINGS

when seeps or springs are present within a study area, it is desirable to obtain a water sample. If the seep or spring discharge is small, it will be difficult to obtain a large sample. For large discharge springs, the sample should be collected by the grab method just as a sample is collected from a discharging well or collected with a peristaltic pump. In either case, the steps taken to collect and prepare the sample should be well documented. A typical procedure to collect samples follows, but it may be necessary to improvise to obtain a sample.

- 11.1 If the seep discharge is small, place a clean and unused bottle where the seep will drip into the container. If necessary, dig out the soil matrix to increase the flow, but document this on Attachment 16.2.1.6. NOTE: if possible, determine temperature, conductivity, and pH data as described in 6.0, 7.0, and 8.0 of Section 16.1.10.
 - 11.1.1 Be sure to note and record now long it takes to obtain a sample.

JACOBS ENGINEERING GROUP INC. SECTION 16.2.1 ALBUQUERQUE OPERATIONS

PAGE 13 OF 13

- 11.2 Pump sample through a filter into a clean, unused, and labelled bottle. Record sample volume on Attachment 16.2.1.0. Discard filter unless directed by the SH to save the filter for lap analysis.
 - 11.2.1 A small sample volume of as little as 10 mL can be analyzed for uranium. NOTE: If seep flow is small, collect a small sample and consult SH for further instructions.
- 12.0 PROCEDURE: FILTRATION AND PRESERVATION OF WATER SAMPLE(S)

whenever water samples are to be analyzed for soluble (or dissolved) analytes, insoluble matter must be removed or this material may interfere with the analysis(es) and could seriously affect detection instrument operations. In addition, the insoluble material in a water sample may sometimes require analysis(es) and must be separated from the dissolved analytes. The soluble and insoluble analytes are separated by filtration. The soluble analytes pass through a filter while the insolubles are retained by the filter. Then, if required, the insoluble matter with the filter is saved and submitted for laboratory analysis(es). The JEG water Samplers should consult with the Site Hydrogeologist prior to sampling to determine if sample(s) should be filtered and whether the insoluble material should be saved for analysis or discarded. This should be recorded on Attachment 16.2.1.6.

- 12.1 Assemble clean hose(s) and clean pump(s). Insert a filter apparatus containing a new filter between the water source and the sample bottle. Check all connections to make sure that they are tight. NOTE: If the equipment has not been cleaned, clean the equipment as described in 12.7.
- 12.2 Before collecting any samples, run a few hundred mL of the water sample through the filter. Discard this water at a site away from the water source. NOTE: If filter becomes clogged during this step or during Step 12.3, replace filter as described in 12.8.1. Do not clean filter or back flush filter with the water sample.
- 12.3 Fill the sample bottles. Allow no dirt or dust to blow into the bottles or onto the bottle caps. Rinse the inside of the bottle caps with filtered water and screw cap onto pottle. Snake the bottle to mix the sample. NOTE: While filling the sample bottle, and once the sample is in the bottle, do not allow the water sample to touch anything but the bottle walls. Never stick anything into the bottle. Do not smoke near open bottles.
- 12.4 If equired, add an appropriate amount of preservative after filling the sample bottles. Refer to Attachment 10.2.1.7 for volumes required and preservation techniques. Record the type of preservative on the sample label. If preservative is not required, proceed with 13.0. NOTE: Keep the water samples out of direct sunlight.

JACOBS ENGINEERING GROUP INC. SECTION 10.2

15.2.1 18

- 12.4.1 If Nitric Acid (HNO3) is to be added, add 5 mL of ~3 N HNO3 per liter of sample, mix sample well, and check pH of sample solution. pH should be <2. If pH is >2, add ~ 8 N HNO3, mL by mL, with mixing after each addition until pH of sample solution is <2. NOTE: Concentrated HNO3 may be used as an alternate: add 3 mL of HNO3 per liter of sample and mix sample solution well.
- If Hydrochloric Acid (HCl) is to be added, add 5 mL of $\sim \sigma$ 12.4.2 N HCl per liter of sample, mix sample well, and check pH of sample solution. pH should be <2. If pH is >2, add NO N HCl, mL by mL, with mixing after each addition until pH of sample solution is <2. NOTE: Concentrated HC1 may be used as an alternate: add 3 mL of HCl per liter of sample and mix sample solution well.
- 12.4.3 If Sulfuric Acid (H2SO4) is to be added, add 5 mL of ~9 M HpSO4 per liter of sample, mix sample well, and cneck sample pH. pH should be < 2. If pH is >2, add ~9 M H2SO4, mL by mL, with mixing after each addition until pH of sample solution is < 2. NOTE: Concentrated H2SO4 may be used as an alternate: add 2 mL of H2SO4 per liter of sample and mix sample solution well.
- 12.4.4 If Sodium Hydroxide is to be added, add 2 mL of ~18 N NaOH per liter of sample. Mix sample well and check sample pH. If pH is <10, add NaOH mL by mL until pH is >10. Mix sample solution well after each addition of NaOH.
- If Sodium Hydroxide and Zinc Acetate solution are to be 12.4.5 added, add 2 mL of ~18 N NaOH per liter of sample and mix sample solution well. Check sample pH. If pH is < 10, add NaOH mL by mL until pH is >10. Add 2 mL of Zinc Acetate solution (220 g/L) per liter of sample and mix sample solution well.
- 12.5 Pour a few drops of the preserved water sample onto the bottle cap and dip the tip of the litmus paper into the bottle cap to check the pH of the water sample. If the pH is >2, add acid, or <10, add base as appropriate. Discard cap contents, place cap on bottle and tighten. NOTE: Keep sample(s) out of direct sunlight.
- 12.0 Check Attachment 16.2.1.6 to make sure that all sampling data has been correctly entered. This should include any comments or deviation(s) from the procedure.
- 12.7 Clean equipment by draining hoses and pumps to make sure that all of the water sample is expelled.

JACOBS ENGINEERING GROUP INC. SECTION 13.2.1 ALBUQUERQUE OPERATIONS

PAGE 15 OF 18

- 12.7.1 Pump ~2,000 mL of distilled water or its equivalent through the hoses/tubes and pump. Continue pumping until all of the distilled water is expelled.
- 12.7.2 Rinse outside of hoses and pumps with distilled water and dry noses/pumps with tissue(s). Hoses/pumps now ready for another sample or storage. NOTE: Dry noses/pumps before inserting them into the well, before connecting them to another pump, or before putting them into a carboy.
- 12.7.3 If hoses and pumps are to be stored, place apparatus in a clean plastic bag and seal with tape. Label with "clean", date, and initials.
- 12.3 If using disposable filters, disregard 12.8 and proceed to 12.9. Clean the filter apparatus by disassembling the unit, discard the filter (if required), and rinse all surfaces which come into contact with the water sample with distilled water. Dry filter apparatus with lint-free tissue(s). Filter apparatus now ready for another sampling activity or storage.
 - 12.8.1 with clean hands, install a new filter, touching the filter only along its perimeter. Re-assemble the filter apparatus. NOTE: Use a pair of tweezers, if necessary, to handle the filter. Allow no dirt or dust to plow onto the cleaned apparatus or filter.
 - 12.8.2 If the filter apparatus is to be stored, place apparatus in a clean plastic bag and seal with tape. Label with "clean", date, and initials.
- 12.9 Perform the following POST-COLLECTION DUTIES:
 - 12.9.1 Clean up field equipment.
 - 12.9.2 Replace expended items, filters, reagents, reagent solutions, etc.
 - 12.9.3 Repair broken equipment or report/request replacement.
 - 12.9.4 Make Xerox copies of all field forms associated with a given lot of samples and distribute copies to the Site Hydrogeologist, Data Manager, water Sampling Manager, and the BOA Tech Rep within two (2) weeks of the date of sample collection. NOTE: Submit the fina! field values for pH, temperature, conductivity, and alkalinity to the Data Manager who will include this data in the water quality data pase.

JACOBS ENGINEERING GROUP INC. SECTION_ ALEUQUERQUE OPERATIONS PAGE 10

SECTION 16.2.1 PAGE 10 OF 18

- 12.9.5 Give original field form(s) containing the sampling information to JEG Document Control.
- 12.10 Notify the Site Manager, by memorandum, of the collection of the field sampling data and delivery of this information to JEG Document Control.

13.0 PROCEDURE: PREPARATION OF SPLIT SAMPLES

- 13.1 Thoroughly rinse the split jug with about one (1) gallon of distilled water or its equivalent. Drain rinse through both the inlet and outlet hoses.
- 13.2 Pump ~10 L of the filtered water sample into the split jug and rinse jug thoroughly with the sample. Drain rinse through both the inlet and outlet noses/tubes.
- 13.3 Whenever possible, pump more than enough filtered water sample into the split jug to fill all of the sample sets. As a minimum, pump more than enough filtered water sample to fill all of the bottles used for identical analyses. Swirl the jug contents thoroughly to mix the water sample.
- 13.4 Dispense sample types consecutively, i.e., fill one anion bottle immediately after the other, one metals bottle immediately after the other, etc.
- 14.0 PROCEDURE: PREPARATION OF SAMPLE SPLITS, KNOWN SOLUTIONS, AND ARCHIVED
 - 14.1 Separate field samples into sample lots for packaging and shipment to the analytical laboratory.
 - 14.1.1 A sample lot should consist of no more than nine (9) field samples, four (4) replicates of one of the field samples, and at least one (1) known solution.
 - 14.2 Every field sample and known solution may also be split into a sample submitted for lab analysis(es) and a sample to be archived. NOTE: Archived samples are stored by the analytical laboratory and may be analyzed by that laboratory and, at the discretion of JEG, used for re-analysis(es) or other purposes.

JACOBS ENGINEERING GROUP INC. SECTION_ PAGE 17 OF 18 ALBUQUERQUE OPERATIONS

- 15.0 PROCEDURE: PREPARATION OF WATER SAMPLES FOR PACKAGING, LABELLING, AND TRANSPORT
 - 15.1 Lapel pottles with the TAC's pre-printed lapels. Fill in all appropriate information in the blank spaces.

16.2.1

- 15.2 Immediately put the samples requiring refrigeration into an insulated cooler containing "blue ice" or ice. Samples which do not require refrigeration should be placed in a box or other suitable container, filled with absorber material, sealed, and snipped to the analytical laboratory. NOTE: Keep all samples out of direct sunlight.
- 15.3 Put the address and telephone number of the UMTRA Project Office and the name of a JEG contact in the insulated container. Include a list giving the number of samples and sample identification in each container. Use Attachment 16.2.1.4. This sample list will be completed by the laboratory and sent to the BOA Tech Rep so that records will be kept on the list of samples received. This list can be compared to a duplicate sample list which is retained with the water sampling data to assure that no samples were lost or misplaced during snipment.
- 15.4 Tape the container with strapping tape for shipping. Inform the laboratory of the sample arrival time and date.
- 15.5 Scnedule sample collection/snipment so that the samples do not arrive at the lab after 4 p.m. on a Friday, Saturday or on a Sunday. Whenever possible, ship the samples so that they arrive at the laboratory no later than 72 hours after they were collected.
- 15.6 Have the laboratory document sample lot arrival times, and report this information to the BOA Tech Rep.
- 13.0 ATTACHMENT(S): SUPPORTING DOCUMENT(S)
 - 16.2.1.1 Schematic Drawing entitled "Point Source Bailer"
 - 16.2.1.2 Site Hydrologist to water Sampling Manager memo
 - 16.2.1.3 Site Hydrologist Work Order Plan
 - 16.2.1.4 UNC Cover Letter
 - 16.2.1.5 Equipment Cnecklist
 - 16.2.1.6 Water Quality Sampling Record Form--4 Forms
 - 16.2.1.7 Potential Sample Overview

JACOBS ENGINEERING GROUP INC. SECTION __ 16.2.1 ALBUQUERQUE OPERATIONS

PAGE 18 OF 18

17.0 REFERENCES

- 17.1 Barcelona, Michael, John A. Helfrich, and Edward E. Garske, 1985 "Sampling Tubing Effects on Groundwater Samples." Anal. Chem. 57: 460-464.
- 17.2 Korte, N. and P. Kearl, 1984. "Procedures for the Collection and Preservation of Groundwater and Surface Water Samples and for the Installation of Monitoring Well." U.S. Department of Energy, Grand Junction, Colorado.
- 17.3 Miller, G. D., 1982. "Uptake and Release of Lead, Chromium, and Trace Level Volatile Organics Exposed to Synthetic Well Casings." Proc. 2nd Nat'l Symp. Aguifer Restoration and Ground Water Monitoring, 1982, 236-245.
- 17.4 Nielsen, D. M. (ed.), 1983. "Cnemical Problems in Groundwater Monitoring Programs." Proc. 3rd Nat'l. Symp. Aquifer Restoration and Ground Water Monitoring, 1983: 263-271.
- 17.5 Pettyjonn, Wayne A., W. J. Dunlap, R. Cosby, and J. W. Keeley, 1931. "Sampling Ground Water for Organic Contaminants." Groundwater, 19 (2): 180-189.
- 17.6 USEPA, 1963. Methods for Chemical Analysis of Water and Wastes. EPA-600/4-79-020. Revised, March 1983.
- 17.7 USEPA, 1985. Practical Guide for Ground Water Sampling. EPA 600/2-85/104
- 18.0 PROCEDURE REVIEW AND APPROVAL

Prepared By:	TAC Representative	5-30-90 Date
Reviewed By:	Task manager	5-181/90
Reviewed By:	Quality Assurance Manager	5/8//// e
Approved By:	Project Manager Deputy Project Manager	5/31/50 Date

ATTACHMENT 16.2.1.1

Point Source Bailer

POINT SOURCE BAILER allows sampling at specific point within a monitoring well

Threaded and press fitted construction allows for ease of decontamination

.

٠

Additional 1 foot midsection(s) may be added at the center to increase volume

.

Availzible in polyvinyl chloride (PVC) or all virgin polytetrafluoroethylene (PTFF), LEXAN* or stainless steel

SIZES AVAILABLE

(25.4 mm · 114.30 mm) Lengths: 2 thru 5 ft

PTFE: 1 inch OD thru

4.5 inches O.D.

: Arranac

0

0

Ċ

HE ' A INFINE

BALL CHECK

BAND I

1 FOOT

A ANI CHICA

TIMCO[™] Bailers



ATTACHMENT 16.2.1.2

JEGG/JEG/1188-0620

JACOBS ENGINEERING GROUP INC., ALBUQUERQUE OPERATIONS

TO: JFritts

FROM:

RHeydenburg

DATE : November 14, 1988

SUBJECT: Water Quality Sampling at Spook, WY

Water quality sampling of monitor wells and lysimeters at the Spock site should begin on November 28, 1988.

LOT 1

SPK01	- 910	
	- 911	
	- 913	
	- 914	
	- 915	
	- 919	
	- 920	
	- 921	
	- 922	
	- 923	
	- 938	
	- 941	
	- 702	(lysimeter)
	- 991	UNC Known

LOT 2

SPK.Ø1	- 912	
	- 918	
	- 927	
	- 928	
	- 935	
	- 936	
	- 937	
	- 939	
	- 940	splits 951, 952, 953, 954
	- 942	그 아들 것 것 같은 것 같은 것 같아. 소리가
	- 993	UNC Known

LOT 3

SPKRI	- 916						
	- 917						
	- 924						
	- 925	spl	its 95	55.	956.	957.	958
	- 931					1111	
	- 932						
	- 933						
	- 934						
	- 943						
	- 944						
	- 995	UNC	Known	5			

Slug tests will be performed in monitor wells 913, 928, and 933 through 944.

Attempt to collect samples from all lysimeters. See attached memo of October 5, 1988 regarding location and sample bottle priorities. If water is obtained from lysimeters in addition to 702, include samples in LOT 1.

Note that monitor wells 929 and 930 will not be sampled.

RH:rc Attachments

cc: BBearden EBanks MGilbert (w/out attachments) FTitus Document Control

			PAGE 1													_	_							-	-	
	SAMPLE I.D										E ASSIGNED BY: SAI															
Gross Alpha/3eta	Co	Cr	Cd	Be	Ba	As	Sb	AI	Br	PO4	\$102	NO3	NO3	NH4	TKN	s	CN	F	8	Са	Mg	ĸ	Na	S04	CI	lottle Iame
									_							-	-									-
																							_			
							-	_																		
									_											_					-	_
									-																	
																										otal of ests
			- 1		1	- 1	- 1	1																	NTS:	MME
																							-		NTS:	otal of ests DMME

				SITE:		PAGE 2 OF 2																
	RK ASSIGNED BY:																		8Y:	GNED	ASSIG	NORK
and the second second																					13510	
Th-23	Ra-228	Ra-226	Po-210	Pb-210	TOC & DOC	TDS	Zn	v	U	τi	Sn	Sr	Ag	Se	Ni	Mo	Hg	Mn	Pb	Fa	Cu	Bottle Name
						_																_
																						tal
																						of
																						fotal # of fests

ATTACHMENT 16.2.1.4

JEGA/UNC/0889-0043

August 1, 1989

Mark Hollenbach UNC Technical Services, Inc. 2597 B 3/4 Rd. P.D. Box 14000 Grand Junction, CO 81502

Dear Mr. Hollenbach:

Please prepare known solutions as indicated below. Labeled set is will be sent to you by our sampling team manager. The known solutions should be bottled as indicated on the attached Work Order Plan. Please send the known solutions to Core Laboratories, 1300 South Potomac, Suite 130, Aurora, Colorado 80012.

> Very truly yours, JACOBS ENGINEERING GROUP INC.

Kent Boshik

Frank B. Titus Manager, Hydrological Services

FBT:jw Enclosure

cc: MGilbert (w/attachments) (w/attachments)

SITE: Gener

APPROXIMATE DATE REQUIRED:

SAMPLE IDENTIFICATION NUMBER

APPROXIMATE TDS DESIRED (mg/1)

699		
975		

JE JACOBS ENGINEERING GROUP INC. SECTION

DH BUFFER	Yes	No	Comments
sufficient volume 4 sufficient volume 7 sufficient volume 10		Material College	
REAGENT & BOTTLES			
Litmus paper Nitric acid (pres.) Sulfuric acid (pres.) not required * Other reagents not required			
Required sample bottles:		Number of for Al, M2 Pb210, Ra226	1000 ml plastic bottle , Po210, Th230, CN, S , Ra228
		Number of for Gross Al	500 ml plastic bottle pha-Beta, Nl
		Number of bottles for	5CO ml amber glas
		Number of bottles for	1000 ml amber glas TOX
Sufficient bottles			

* If required, must explain under <u>Comments</u>.

ATTACHMENT 16.2.1.5

JE JACOBS ENGINEERING GROUP INC. SECTION _____

OTHER EQUIPMENT	Comments
squeeze bottle	
acid dispensette	
deionized H ₂ O quantity gallons	
distilled H ₂ 0 gallons	
water level sounder hand tape cloth towels or wipes	battery_check
FIELD FORMS AND MISCELLANEOUS EQUIP	MENT :
Expected no. of samples	No. of forms
Clipboard with cover	
Maps marked with well locat	ions
well information (completion, d	epth, etc.)
Field instruction book	
Key(s) to well(s)	To be picked up at
Clean pails	Transparent tape
WD-40 (for locks)	Strapping tape (nylon)
Marking pen	
Coolers	
Shipping address of lab(s):	

JEG-AL-QA-21 (8/87)

JE JACOBS ENGINEERING GROUP INC. SECTION _____

(Fill out all blank	(s prior to leaving for field)
Phone numbers and contact:	Comments
PUMPS AND FILTERS	
Logic unit	Power cord
Regulator	Logic unit cord
Wrenches	Water hose
Air hose	
PERISTALTIC PUMP	ander som er se som en
Battery cable	Rechargeable baltery
Clean? yes no	Sattery charger
Tubing Length	
Filter system:	
Inspect filter housing for cr	acks
Filter unit legs	
Number of filters	
Filter size	

ATTACHMENT 16.2.1.5

	CONTRACTOR DO NOT			A I I	ALMMENI 10.	2.1.6			and the foreign over the same of the same of the
			JE JACC	DBS EN	KGINEERING GR	OUP INC			Page 1 p
		1	WATER QU	ALIT	Y SAMPLI	NG RI	ECO	RD	
SITE	ID:		An of the local states and the second		FINA	L FIELD	VAL	JE AT	THE SURFACE
LOCI	ATION	ID:					been and the		
SAM	PLE ID		The second second second second		pH(S.U.)				
STAT	NC WA	TER LEV	EL (FT)		Ec (µmhos/	cm)			
SAMP	PLE DE	PTH (FT))		TEMP. (*C)				
SAMP	PLING	DATE			ALKALINIT	Y (mg/l	CalCo	3) @	pH 4.5
SAMP	LING	TIME:			TITRANT:	1.6 N I	1250	4 LO	τ#
001	MPLETI	E	national constantion of the						FILTER(S) US
OCA	TION DE	SCRIPTIO: C D	V O. (air)	mg/l	Eh soln.(.1)		m	7	at. *C
FIELD	REP(S)	na o rate da constante da la constante da la constante da la constante da constante da constante da constante m	an a			14			
DATE	TIME	TOTAL VOLUME WITHDRAWN (Gals) (Bore Volum	VOLUME DRAWN	pH umes)	Ec (µmhos/cm)	TEMP. (*C)	En (mV)	D. O. mg/l	COMMENTS
			(Bore Volumes)						
		0.0	0.0	*	-				START PUMPING
-									
0		***					*******		
0									

(d/2) = T (h1-h2)*	SAMPLING INFORMATION
SITE ID	LOCATION ID
PEPTH TO WATER (h2) (FT.)	WITHDRAWAL METHOD
DEPTH OF WELL (h1)(FT.)	FILTER SIZE45.
WELL DIA (d) (FT.)	THERMOMETER ID
BORE VOL. (FT.) ³	EC METER ID
SCREENED INTERVAL (FT.)	PH METER ID
	PUMP ID
	Eh METER ID
	D.O. METER ID
CALIBRATION INFORMATION	
DATE OF LAST EC CALIBRATION	
TIME OF PH CALIBRATION	
DH AFTER MEASUREMENT	FOR STANDARD PH
PH AFTER MEASUREMENT	FOR STANDARD PH
SHIPPING INFORMATION LAB(S) SHIPPED TO: DATE(S' SHIPPED: METHOD OF SHIPMENT:	



NOTES

JE JACOBS ENGINEERING GROUP INC.

4 cf 4

EXHIBIT C ATTACHMENT 4 SUBCONTRACT NO. 34-6705-S

ACKNOWLEDGEMENT OF RECEIPT OF SAMPLES FOR DELIVERY ORDER NO. 0

\$1TE 1D:	LOCATION ID:	SAMPL	E 1D:		
SAMPLE SHIPMENT LIST		SAMPLE MATRI	X (Check of	ne)	
LOT#		WATER ()	SOIL ()	
BOTTLE AMOUNT A1	REC'D (~)	BOTTLE 1D Th 230 GA/B N1 CN S TOC Alkalinity		REC'D	(~
DATE SHIPPED:	MET	THOD OF SHIPMENT	1		
COMMENTS:					
To be completed by sub- I hereby acknowledge re Delivery Order No Subcontractor: Name: Title:	contracting laborato eccipt of the foll Work Ord	ory lowing on: der Plan dated:_			
COMMENTS:			997 - 1000 (1997) 1997) - 100 (1997) - 100 (1997) 1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) - 100 (1997) -	Constantinum de la constantinum de	
In accordance with Exh Water Quality Analytical	ibits B and C of t	the Subcontract			
Resu	ilts due to JEG by:				
		(08	(ce)		

JEG-AL-ENG-44 (1/90)
PAGE _____ CF ____

POTENTIAL SAMPLE OVERVIEW

Bottle 1.D.	Bottle type	Volume (ml)	Preservation techniques	Analytes
A	Polyethelene	2000	Filtered, 4 ⁰ C	F, C1, S04, PO4 S102, B, TDS, Br
м	Polyethylene	1000	Filtered, HNO ₃ to pH<2	Na, K, Mg, Ca, Sr, Ba, Mo, As, Cd, Cr, Hg, Pb, Sb, Se, Al, Fe, Mn, Ag, Co, Cu, Ni, Sn, V, Zn, U, Be, Tl
NI	Polyethylene	500	H2504 to pA<2, 40C	NH_4 , $NO_2 + NO_3$, NO_3
CN	Polyethylene	1000	NaOH to pH>12, 4°C	CN"
s	Polyethylene	1000	NaOH to pH>9, 2.5 ML of 220g/L zinc acetate solution per liter	s ²⁻
TOC	Amber glass w/ Teflon liner	500	4° C H ₂ SO ₄ to pH<2	тос
тох	Amber glass w/ teflon liner and septum	1000	4°C	тох
Pb210	Polyethylene	1000	Filtered, HNO ₃ to pH<2	Pb210
Po210	Polyethylene	1000	Filtered, HNO ₃ to pH<2	Po210
Ra226	Polyethylene	2000	Filtered, HNO ₃ to pH<2	Ra226
Ra228	Polyethylene	2000	Filtered, HNO3 to pH<2	RA228
Th230	Polyethylene	1000	Filtered, HNO3 to pH<2	Th230 .
GA&GB	Polyethylene	500	Filtered, HNO_3 to $pH<2$	GA&GB

Any deviations from the above table will be noted on the Acknowledgement of Receipt of Sample form which accompanies the samples to the laboratory.

ATTACHMENT 16.2.1.7

JEM- 11-MA-21 (8/87)

ACOBS ENGINEERING GROUP INC.

ALBUQUERQUE OPERATIONS

AL PLIQUEDOUE OPERATIONS MANUAL	SECTION 16.2.2		
ALBOQUERQUE OFERATIONS IVANUAL	DATE 06/09/89		
	EFFECTIVE 11/20/89		
WATER SAMPLING FOR TRITIUM ANALYSIS	SUPERSEDES 05/28/87		
	PAGE 1 OF 3		

1.0 PURPOSE

To provide a procedure for the collection of water samples from soil water suction samplers or from pumped wells for tritium analysis.

2.0 SCOPE

This procedure is applicable to members of TAC and its contractors and subcontractors with responsibilities and duties for the collection and submittal of water samples for tritium analysis.

- 3.0 EQUIPMENT
 - 3.1 Sample Bottles (4 pottles per sampling site): Glass bottles; 250 mL, amber, new, certified clean, and dry
 - 3.2 Polyseal or similar caps
 - 3.3 Argon Gas; tank of compressed gas. Tank should contain sufficient amount of gas to purge glass bottles with argon gas
 - 3.4 Regulator: for Argon Ges tank
 - 3.5 Tubing: rubber. To transport argon gas from tank into glass bottle(s).
 - 3.6 Clamps: for hose/sube
 - 3.7 Control valve: to regulate argon gas flow from tank to sample bottle(s).
 - 3.8 Snipping Container: with packaging material
 - 3.9 Sample Labels
 - 3.10 Address Labels
 - 3.11 Strapping Tape
 - 3.12 Tooi Box
 - 3.13 Sampling Record Form(s): Water Quality Sampling Record Forms

JE JACOBS ENGINEERING GROUP INC. SECTION 16.2.2 ALBUQUERQUE OPERATIONS PAGE 2 OF 3

4.0 PROCEDURE

- o Water is recovered from the soil zone with a suction water sampler and from most wells with a bladder-type pump. Refer to the separate Standard Operating Procedures on soil water suction samples (16.1.7) and well sampling (16.2.1) for information on the use of these devices. Sampling water for tritium analysis is similar to sampling for other analytical purposes, except that contact of the solution with the open atmosphere should be minimized to limit contamination by water vapor in the air.
- o CAUTION: Due to the use of tritium in modern luminous dial watches, samplers should make sure that these watches and other self-luminescent devices are not in the vicinity of open sample containers or during sample collection activities
- o A minimum of 100 mL of water is needed from each sampling site to provide enough solution for tritium measurement at the required detection limit of 3.2 pCi/mL (0.1 Tritium Units) NOTE: Actual value is 3.231 pCi/mL based on tritium half-life of 12.30 years.
- o if possible, collect four (4) 250-mL samples for analytical and statistical purposes.
- 4.1 Using the appropriate procedure, prepare the suction sampler or bladder-pump to deliver a water sample.
- 4.2 Purge an amber bottle with argon to remove air.
- 4.3 Insert clean tube from the water sampler to the bottom of the bottle.
- 4.4 Allow water to fill bottle until there is a one-inch head space.
- 4.5 Remove the tube from the bottle and fit with polyseal cap.
- 4.6 Label bottle with site identification, location identification, sampler number, date, and time of sampling (use military time), and name of sampler. State that the sample is for tritium analysis and note any special sampling circumstances (e.g., color, turbidity of solution, prolonged sampling period, etc.).
- 4.7 Fill out the same information on Attachment 5.1 or its equivalent. If the tritium water sample is taken during regular sampling of a well, then the information on the sampling record pertinent to purging the well need not be repeated. However, if the tritium sample is the only one taken from the well, then the data on well-purging should be recorded on the tritium sampling record. If the sample was taken with a soil water suction sampler, fill out only the information listed in 4.6 on the appropriate soil water sampling record.

JACOBS ENGINEERING GROUP INC. SECTION 16.2.2 ALBUQUERQUE OPERATIONS

PAGE 3 OF 3

- 4.8 Place sample in shipping container and protect from breakage with packaging material.
- 4.9 when samples are tightly capped there are no special time or temperature requirements during storage or transit; however, temperature extremes should be avoided and holding time(s) should be kept to a minimum.
- 4.10 Obtain additional 250-mL samples from this samuling site at this time. If this is not possible because of insufficient water supply, prepare the suction sampler for later sampling.
- 4.11 When sampling is completed at all sites, send samples by the fastest available means to the analytical laboratory. Include a packing sheet with the samples that lists all the samples and states that they are for tritium analysis by enrichment and low-level counting. If possible, include a purchase requisition number with the samples.
- 4.12 Clean, wrap (if necessary), and store equipment. Label with "Clean", date, and initials.
- 4.13 Inform the Site Hydrologist of the time at which the samples were sent to the analytical laboratory and their expected arrival times at the laporatory.
- 4.14 Make copies of the sample collection field forms for document control and for the Site Hydrologist.
- ATTACHMENT(S): Supporting Documents 5.0

5.1 Attachment 5.1 water Quality Sampling Record Form

5.2 Attachment 5.2 Acknowledgement of Receipt of Samples

Prepared By:	Whows
Reviewed By:	TAC Jecnnical Representative
Reviewed By:	But Research
Approved Byr	Muero Derhiert Manager
C	Jridgett Hanager

			ATTA	CHMEN	7 5.1: Page	1 of 2	
			JE JACO	BS ENK	GINEERING GR	OUP INC	Page 1 of
		١	WATER QUA	LITY	Y SAMPLI	NG RE	CORD
SITE	ID:				FINA	L FIELD	VALUE AT THE SURFACE
LOCA	ATION	ID:			March Management	dina ang sa	an a fa a san an an ann an ann ann an an an an an a
SAM	PLE ID	1			pH(SU)		
STAT	TIC WA	TER LEV	EL (FT)		Ec (µmhos/	cm)	Ministrating second representation
SAM	PLE DE	PTH (FT)		TEMP. (°C)		
SAM	PLING	DATE		automater.	ALKALINITY	Y (mg/l	CaCO3) @ pH 4.5
SAM	LING	TIME:			TITRANT:	1.6 N H	2504 LOT #
STA	ART						
CO	MPLETI	Ε			Alter factor (10 constant)		FILTER(S) US
		<u> </u>	Same Same			-	an in a state of the
COMM	AENTS					Contraction Contraction	
C Critici							
LOCA	TION DE	SCRIPTIO	N				
					and a state of the second second		
FIELD	REP(S)						and the set of the state of the
DATE	7.1.1.5	TOTAL	VOLUME	рH	E¢	TEMP.	CONVENTE
DATE	TIME	(Gais)	(Bore Volumes)	'n	(µmhos/cm)	(*C)	COMMENTS
	And States	0.0	0.0				START PUMPING
and an other of the							
and the second se							
						-	
			1				

JEG-AL-ENG-44 (8/87)

ATTACHMENT	5.1: Page 2 of 2
WATER QUALITY	SAMPLING RECORD
BORE VOL CALCULATION (d/2) ² T (h1-h2)*	SAMPLING INFORMATION
SITE ID	LOCATION ID
DEPTH TO WATER (ba) (FT.)	WITHDRAWAL METHOD
DEPTH OF WELL (h.) (FT.)	FILTER SIZE .45.0
WELL DIA (d) (FT.)	THERMOMETER ID
BORE VOL. (FT.)3	EC MET R ID
SCREENED INTERVAL (FT.)	PH METER ID
	PUMP ID
CALIBRATION INFORMATION	
DATE OF LAST EC CALIBRATION	
TIME OF DH CALIBRATION	
DH AFTER MEASUREMENT	FOR STANDARD PH
PH AFTER MEASUREMENT	FOR STANDARD PH
SHIPPING INFORMATION	
LAB(S) SHIPPED TO:	an a second and a second s
DATE(S) SHIPPED:	anna anna anna anna anna anna anna ann
METHOD OF SHIPMENT:	
NOTES	
* The value of $\left(\frac{d}{2}\right)^2$ to is : .022 for a 2' well .086 for a 4' well	

		ATTA	CHMENT D.C		анарына арталык калыры
	WATER	QUALITY	SAMPLING P	RECORD	
	61157AU757	CT 10			
Antoniadoneon	SUDLUNIKA	* AF Samples 1	for Delivery Or	far No. 0	
SITE ID:	L DI NECEIP	OCATION ID:	SAM:	PLE 1D:	
		SAMPLE SHIPMEN LOT#	NT LIST		
BOTTLE ID	AMOUNT	RECID	BOTTLE	AMOUNT	<u>REC'D</u>
A1 A1-A M1-A M2-A F5 210 Po 210 Ra 225			Th 230 GA/B N1 CN S TOC TOX S102		
Ra 228 DATE SHIPPED: COMMENTS:		() ME	THOD OF SHIPMEN	NT:	
I hereby ackno	wledge rece	ipt of the fol	llowing on:		
() Delivery	Order No.	(Nork Ore	ler Plan) dated	t anter verannen anter verannen anter verannen	
Subcontractor: Name: Title: Date:	-		the state of the s	CONTRACTOR OF A	
Subcontractor: Name: Title: Date: Complete Analy	sis Due: _				

.

JEG-AL-ENG-44 (8/89)

JE

ALBUQUERQUE OPERATIONS

AL RUOUEROUE OPERATIONS MANUAL	SECTION 16.2.3
ALBOQUERQUE OFERANDING WANDAL	DATE 5/28/87
EVALUATION OF CHEMICAL ANALYSES OF WATER SAMPLES	EFFECTIVE 08/18/88
	SUPERSEDES 8.3.9 5/10/85
	PAGE 1 OF 3

1.0 PURPOSE

To define the procedure used in the UKTRA Project to evaluate and document the precision and accuracy of laboratory analyses of water samples.

2.0 SCOPE

This procedure outlines the criteria for the systematic submission of water samples for laboratory analysis and the evaluation and documentation of the resultant data. This procedure is applicable to all members of the UMTRA Project for submitting water samples to laboratories and evaluating results of laboratory analyses of water samples.

3.0 PROCEDURE

- 3.1 Water samples contained in plastic or glass bottles will be shipped to laboratory subcontractors by Jacobs Engineering Group, Inc. representatives.
- 3.2 Before and after analysis, samples will be stored in a manner approved by Jacobs so as to minimize the precipitation of unstable constituents. Any remaining portions of samples after analyses, and all archive samples, shall be retained by the laboratory until direction is given by the Jacobs Technical Representative for disposal of the samples or shipment to another laboratory.
- 3.3 All laboratory analyses will be specified in writing by Jacobs Engineering Group (JEG) from the list of constituents on Figure 1. Figure 1 lists the acceptable detection limit for each water chemistry parameter.
- 3.4 The subcontractor laboratory shall maintain approved quality assurance/control programs, including appropriate chain-ofcustody procedures, to ensure adequate sample traceability, that the data transmitted are correct, and that the laboratory analyses were performed in accordance with EPA approved methods, or other JEG approved methods, which result in verifiable accuracy ranges as stated below.
- 3.5 Pre-established accuracy criteria and cation-anion balance checks will be used to evaluate the acceptability of laboratory

JEQ-AL-CA-DDIA/87)

JACOBS ENGINEERING GROUP INC. SECTION _ 16.2.3 ALBUQUERQUE OPERATIONS

PAGE ____ OF ___ 3

analyses. Instructions of the details of methods and calculations to be used are specified in Technical Memo No. 3 "Instructions for evaluating the accuracy, cation-anion balance, and precision of istoratory analyses of water samples."

- 3.6 JEG representatives will group the water samples into lots. A lot is defined as a group of field samples associated with at least one solution of known composition. Lots will consist of not more than nine field samples plus known solutions and splits. Known solutions will be shipped to the laboratory with the field samples within that lot.
- Instructions for initiating water sampling, bottling, labeling, 3.7 and preservation of samples are provided in Technical Memo No. 2, Revision 3, by George Rice.
- 3.8 Instructions regarding preparation of splits, blanks, and alkalinity solutions for submittal to laboratories are given in Technical Memo No. 4, Revision 1, by George Rice.
- 3.9 Technical Memo No. 3, Revision 5 (referenced above), will be followed by JEG hydrologists in evaluating the accuracy and precision of each lot of samples. The actual evaluation of the lab data will include the following:
 - 1. Computation of the acceptable range of each constituent from the known solution.
 - 2. Values reported by the laboratory will be compared with the ranges compited above.
 - 3. Each analysis shall satisfy the cation-anion balance criterion. The balance must be within plus or minus.
- 3.10 When the reported value of any constituent in a known solution falls outside its acceptable range, the laboratory will be instructed to re-analyze the entire lot for that constituent. When a sample analysis does not satisfy the cation-anion balance criterion, the laboratory will be instructed to re-analyze that sample until the anion-cation balance criterion is met.
- 3.11 DOCUMENTATION OF WATER QUALITY RESULTS
 - 3.11a The JEG water quality analytical results form (Figure 2) shall be utilized by each subcontractor laboratory to report the results of all analyses.
 - 3.11b The evaluation of accuracy and precision by the UMTRA hydrologists shall be documented according to Technical Memo No. 3 on a form that is filed with the UMTRA Document Control Center.

JEG-AL-0A-21(8/87)

JE JACOBS ENGINEERING GROUP INC.	SECTION
Prepared By: Hydrologist	
Reviewed By:	
Approved Bys Augurance Manager	
Coject Manager	

JE JACOBS ENGINEERING GROUP INC. SECTION 16.2.3

PAGE _____ OF ____

FIGURE 1

REVISED LIST OF CONSTITUENTS AND DETECTION LIMITS FOR WATER ANALYSES

Constituent	Detection Limit (mg/l)	Constituent	Detection Limit (mg/1)
Chloride (C1) Sulfate (SD) Sodium (Na) Potassium (K) Magnesium (Mg) Calcium (Ca) Boron (B) Flouride (F) Cyanide (CN) Hydrogen Sulfide (H ₂ S) Ammonium (NH ₄) Nitrite (NO ₂) Nitrite (NO ₃) Nitrite (NO ₃) Nitrate & Nitrate Silica (SiO ₂) Phosphate (FC ₄) Iron FE	1.0 0.1 0.002 0.01 0.01 0.1 0.1 0.1 0.1 0.1 0.1 0.1 0	Lead (Pb) Manganese (Mn) Mercury (Hg) Molybdenum (Mo) Nickel (Ni) Selenium (Se) Silver (Ag) Strontium (Sr) Tin (Sn) Uranium (U) Vanadium (V) Zinc (Zn) Total Dissolved Solids (TDS) Total Organic Carbon TOC TOX	0.01 0.01 0.0002 0.01 0.04 0.005 0.01 0.005 0.005 0.005 0.005 10 1.0
Aluminum (A1) Antimony (Sb) Arsenic (As) Barium (Ba) Cadmium (Cd) Chromium (Cr) Cobalt (Co) Copper (Cu)	0.1 0.003 0.01 0.1 0.001 0.001 0.05 0.02	Radionuclides Lead 210 (PD-210) Polonium 210 (Po-210) Radium 226 (Ra-226) Radium 228 (Ra-228) Thorium 230 (Th-230 Gross Alpha Gross Beta	pC1/1 1.5 1.0 1.0 1.0 1.0 1.0 1.0

186-AL-CA-21(8187)

		JE		8 & Teber 8			
	WATE	RQU	JALITY	ANAL	TICAL	RESULT	5 Sheet of
SITE ID:			LOCATION	D:		- SAMPLE	ID:
LOT CONTR	OL NO						
SAMPLE TY	PE				DATE SAM	APLED:	
LAB NAME	-					DATI	E REC'D:
LAB SAMPL DATE A VAL SAMPLE CO	E ID:	N REC	EIPT):		CHECKED) BY:	
COMMENTS);						
PROPERTY OF THE PARTY OF THE PA		RES	ULTS		DETECTION	LIMIT	
PARAMETER	PRIMARY	UNITS	DUPLICATE	UNITS	VALUE	UNIT	COMMENIA
Geografie							
	an 174						
			and the logist distribution of the C. Some				A Des a l'Altra de la construcción
					1		and the second

JACOBS ENGINEERING GROUP, INC., ALBUQUERQUE OPERATIONS

TO:	Hyperd	Group and Water Samplers	
	12.1	1 Dutel JRH	

- FROM: BRICEYEDeutsch/DMiller
- DATE: January 28, 1986
- SUBJECT: [Technical Memo No. 2, Rev. 3 Instructions for Initiating Water Sampling and Bottling, Labeling, and Preservation of Samples

These instructions are to be followed by site Geohydrologists, the Water BOA TR, and the Water Sampling Teams.

DUTIES OF THE SITE GEOHYDROLOGIST

No later than three weeks before sampling is to begin the site Geohydrologist will:

- (1) Determine the number of samples to be collected and group them into lots. A lot will consist of no more than nine field samples, plus four splits of one of the field samples, plus at least one solution of known composition. As far as is practicable, lots should be composed of chemically similar samples. At least two lots will contain splits. One set of splits should be for a sample with a high TDS content, the other set for a sample with a low TDS content. The TDS content of the known solution should be similar to that of the lot with which it is associated.
- (2) Determine which constituents each sample is to be analyzed for. All "OX analyses will be performed by ACCU-LABS.
- (3) Tell the water BOA TR how many known solutions are to be prepared and the approximate TDS values desired. In most cases, and solutions will be prepared for nearly all of the constituents chosen in Step 2. If Bendix is to analyze any samples, tell the TR how many and what types of analyses they are to perform. Prepare required paperwork (Attached).
- (4) Determine the number of sample bottles required. Notify the water sampling team manager.
- (5) Prepare a work order plan and requisition. Give them to the water BOA TR (see attached example). See section titled Duties of Water Samplers for Information Concerning Bottle Types, Constituents, Labeling, etc.

(6) Prepare maps for the water samplers and the Health and Safety Group showing all sample locations. Tell samplers the type of sample to be collected at each location. Make sure samplers have all required keys.

PUTIES OF THE WATER BOA TR

- Review work order plans to assure they are appropriate and complete. Upon approval, sign attached requisition and deliver to contracts within two days.
- (2) Tell Bendix how many known solutions are to be prepared and what analyses they are to perform, as per step 3 above.
- (3) Order sample bottles as requested by the manager of the water sampling teams. The bottles are to be pre-cleaned and oven dried as per EPA recommendations. They are not to contain any preservatives. A source of bottles is:

1-CHEM Research 23787 Eichler St., Bldg. F Hayward, California 94545 415-782-3905

DUTIES OF THE WATER SAMPLERS

Samples will be bottled as follows.

BOTTLE 1.D.	BOTTLE	VOLUME (m))	PRESERVATION TECHNIQUES	ANALYTES
A1	Polyethylene	2000	Filtered 4°C	B. F. C1. NO ₃ ,SO ₄ , S102'TDS PO ₄ ,TDS
M2	Polyethylene	1000	Filtered HND ₃ to pH<2	Ba, As, Cd, Cr, Hg, PD, Se, Al, Fe, Mn, Ag, Co, Cu, Ni, V, Zn, U, Ca, Mg, Na, K, Sr, Mo, Sb

282.00

Sn

1

SOTTLE 1.D.	BOTTLE	(m1)	PRESERVATION	ANALYTES
Pb210	Polyethylene	1000	Filtered, HND3 to pH<2	Pb210
Po210	Polyethylene	1000	Filtered HNO3 to pH<2	Po210
Ra225	Polyethylene	2000	Filtered HND3 to pH<2	Ra226
Ra228	Polyethylene	2000	Filtered HNO3 to pH<2	Ra228
Th230	Polyethylene	1000	Filtered HND3 to pH<2	Th230
62 2 6 <i>8</i>	Polyethylene	500	Filtered HNO3 to pH<2	6 o 6/
NI	Polyethylene	500	H 504 to ph (2, Arc	N02+N03 NH4
CN	Polyethylene	1000	NaOH to pH>12. 4°C	CN
5	Polyethylene	1000	NaOH to pH>9, 2.5 M of 220g/1 zinc acetate solution per lite	S er
TOC	Amber glass w/ teflon liner	500	4°C H SO4	TOC
TOX	Amber glass w/ teflon liner and septum	1000	4°C	TOX

Each set of samples sent to a laboratory will contain a table listing the number of samples and sample 1.D.'s contained in the shipment. Any departure from the above procedures will be noted on the table.

. . .

Known solutions will be handled as follows. Sample bottles will be labeled according to the site geohydrologist's instructions and sent to Bendix. Bendix will fill them and send them back to the sampling team. In cases where Bendix does not provide known solutions for certain constituents the sampling team will make up phony solutions and press ve them just as real solutions are preserved. Known solutions are to be sent to the laboratory at the same time as their associated lots are sent.

Keep an inventory of sample bottles. When more bottles are required, notify the water BDA TR.

GR/BD/DM/cfm

Ron Chessmore Bendix Field Engineering Corporation P.O. Box 1369 Grand Junction, Colorado B1502

values a merelamentaria. Available a an east of a second term of any size a

Dear Mr. Chessmore:

10

Please prepare known solutions as indicated below. Labeled bottles will be sent to you by our sampling team manager. The known solutions should be bottled as indicated on the attached work order plan.

> Very truly yours. JACOBS ENGINEERING GROUP INC.

John Thackston Manager, Hydrological Services

SITE :

APPROXIMATE DATE REQUIRED:

OF KNOWN SOLUTIONS

APPROXIMATE TOS DESIRED (mg/1)

WORK OPDER RANS \$ 1 - For worker samples sent to E.D.A or Accu-LABS.

STIMMENT BELIVE AN STATEMENT - MAIN BITTER SHEER

	a Ba Cel Cr Co Creschrona		~ ~ ~ /													
12.02	\$	+	2	+	+	\dagger	\dagger	+	+	+	1	T	1			
1.1	B	TI	>	1				1			1	1				and a set first taxe
. [\$	11		1	1			1	1	1	1	+	1	-		
	ž	1	1	1	1	1	1	4	1	-	4	1	-	-		
	2882	1							1	1	1		1			
	8	>			1		1	1			1	1	1		11	
: 1	ũ • 1					1	1				1	1	>	1		
Ĭ	Ĩ				1	1	1	1	-	4	-	1	>	-		
	Ē										1	>				
111	~	1		1	1	1	1	T	-	>	1	1	1	1	1	
	8	1			1		1		>		1	1	1.	1	1	1
	-	17			1	1	1	1			1	1	1	1	1_	1
	-	1						1	1	1	1	1	1			
	3	1	>			1		1			1	1	1		HCSD MARK	
	£	1	>								1		1	1	1	
		1	>								-		_	+	-	
A A	2	1	>					_			_		_	-	+	
	ž	>					-					-	-		-	-
590.000	5	>				-	0	-			_			1	1	
Annual and Annual Annua	11	91	E	Phen	Pot	Rate	Y'ZY	Cool	CN	5	Toc	TOX	Z			101

. .

work order plan \$1 for worker samples sent to E.D.A or Accu-MBJ.

}

WHEN CODER PLAN - MANIFYS OF MATTER SAMPLES (Concluded)

11	8.2-43							>										
	812-618			İ		t	>											
	842-918				ŀ	-												
- 31 al 1	Pro-194				•						1						1	-
SITE :	912-94			>													1	
	¥			-						1	1	1	4				1	
	F	1						1							1			
	£	1	1	-					1	1	1		1	1	-	1	1	
	-	1	>								1				1	1		
. E		1	1				1	1	1	1	1	1	1	1	1	1	1	
1 Lune	3	1	7				1	1	1	1		1	1	1	1	1	1	
A 111	2	1	1>			-	1	1	1	1	1	1	1	1	1			
	8	1	1>			1	1	1	-				1		-			
	x	1	1			T	1	1	1	1	1	1	1	1	1	1	1	
		1	1>				1		1	1	1				1	1		
	2	1	12	1		1	1	1	1	-			1		1			
	E	1	17		1	T	1	1	1									
NA NA	1	1	1>		T	1	1											
	2	1	1>	•	1	1	1											
MSRC MSSRC		T	17	-	1	I	1										-	and the second second
Number of Street	3	T	P	-				-										
		T	B	01710	011		a la	Ra ²²³	They	(edla	S	5	100	TOX	Z			W101

......

*

WORK ORDER RANS \$2 - FOT the Bendix known solutions

STIGHTS WITH TO SIST THEF - BUT HIDDER YOUR

J

.

**

4.1

.

.

	in the	-	-	-	-		-	-	~	-	-	-	-	-	+	Cara Managara and And And
11 +	57	+	1	1	-	-	-	-	F	+	-	-	-	-	+	
Πì	3		2	1		L.				1	1	-	-	-	1	-
11 E	8		1	1					1_		1	L	1		\square	
11 [3		1>				1	1.	1	1	+	1	+	+		
L' [-	1	17		L		1				1	1	1	1	$\left \right $	
	4		17	1					1	1	1	1	1	1	1	
S I	8	17	-	L			1	1		1	1	1	1	1		
	a		17	•	1					1				-		
. [4		1	1		1.	1	1	1	1	1	+	1	-	1	
	ž	2	1		1	1	1	1	1	+	+	1	1	-		
	2005	>		1				1	1			1				
		>	1	T	1	T			1		1	1	1	1		
1	£ • £	11			1	1	1	1	1	-	>	1	1	1	1	
Ĭ	I	11	1	1	1	1	1	1	1	1	>	+	1	-	-	1
	E		1	1	1								-	1	>	
111	-	TT	1	T	T	1	1	1	1	1	1	1	>1	1	1	1
	B	11	1	1	1	T	1		1	1	1	>	1	•	1	1
111	-	121	1	T	1	1		1	I	1	1			1	1	1
111		1>1	1	1	1	1	1		1				1		1	1
	3	TI	1	>	1	T		1	1		1			1	.1	
	1	TI	1	->	1	1								1	1	1
	Rd	1		>												
1×	8	1		>												
in an	Ā	17											-			
Incard I	0	1>	1							1						
THE BUT	11	10	Ē	TE	pp	Po 200	e ut	Que 119	Bh 230	(xx/L	N	3	In	12	XOT	MIGH .

• •

WORK ORDER PLAN \$2 - FOR THE BENDIX KNOWN SOLDHONS.

0

7

)

WHEN CONTRACTION - MANY YOU'S OF MAITE SAMPLES (Concluded)

The second

-10

	RJ-61 0		-			-		1	>			-		-	1	1		
	22-98		1		-			>	_		_	-		4	4	-	noK ana	
	942-91						>											
Xett	3%-2/18					>										_		
BMIE(15)	867-64			ŀ	>	-												
	¥			1										>				
	iii	>							-									
	£	T		2			1		1	1	1				1	1	1	
		1		>		1		1	1	1	1	1	1		1	1	1	
1111		1	1	>		1	1	1	1	1	1	1	-	1	1	1	-	
	3	1	1>			1	1	1	1	1	1	1	1	1	1	+	-	
111	*	1	1	17		1	1	1	1	1	1	-	-	+	1	1	-	
	2			1			1	1	1	1	1	1	1	1	1	1	-	
	3			1	1	1	4	4	1	+	-	-	+	+		+	+	
	8		1	1>	1	1	1	_		-	1	-		1	-+	1	+	
	1	1	1	>	1	1	1					1	-	-	-	+	-	annanii Anton
	8		1	17	-		_		_		•		_	-	_	1	-	
A Mai	1		1	1	-									_		_	-	
10 0.00	8	1	1	13	-		_		1				_	_				
MS NG MS STO	1	:	1	1	>		-	-	-	-	-	-		_	-	-		
New York	1	3	1	-	*	0	1	1	R	12	-	-	-	-		x	-	¥ .
L	mile		AL	E	NE	PP	por	Q.L	Per	~F	X	12	13	5	P	5		8

,

JACOBS ENGINEERING GROUP, INC., ALBUQUERQUE OPERATIONS

TO: Hydrology Staff

June 27- 1986

DATE :

SUBJECT:

Mechnical Memo #3 Rev. 5 Instructions for Evaluating the Accuracy, Cation-Anion Balance, and Precision of Laboratory Analyses of Water Samples

This memb accompanies the SOP "Evaluation of Chemical Analyses of Water Samples." It outlines steps to be taken after laboratory analyses of Water samples are received.

EVALUATION OF ACCURACY

A solution of known composition will be associated with each sample let (see Technical Memo #2: "Instructions for Initiating Water Sampling and; Bottling, Labeling, and Preservation of Samples"). The accuracy criteria require the reported value of each constituent in a known solution to fall within an acceptable range. If the reported value is unacceptable, all analyses for that crestituent in the entire lot are unacceptable. The evaluation will be perimmed as follows.

- The acceptable range of each constituent in the known solution will be computed.
- (2) The reported values will be compared with the ranges computed above.
- (3) If all reported values fall within their acceptable ranges the lot is acceptable with respect to the accuracy evaluation.
- (4) If the reported values of any constituents fall outside their acceptable ranges, notify the water BOA TR. The laboratory will be instructed to re-analyze the entire lot for those constituents.

The above skeps will be completed no later than seven days after analyses are received.

(C) if any reported values from re-analyses do not fall within their acceptable ranges, notify the water BOA TR. The laboratory may choose to perform further re-analyses. The entire lot is unacceptable with respect to a given constituent until the re-analysis falls within its acceptable range.

CALCULATIONS

Calculate the acceptable ranges for reported values of known solutions as follows:

 Account for the uncertainty in preparing the known solution. The known concentration Registry from:

X (1-1) to (.-U)

Where:

X * Known concentration as reported by Bendix

U = Control sample uncertainty as reported by Bendix

(2) Use the allowable deviation (D, in percent) from the attached accuracy criteria to calculate the acceptable range.

Acceptable Range = $\left(\frac{X(1-U)}{\frac{D}{100}} + 1\right) t \left(\frac{D}{100} + 1\right) X(1+U)$

EXAMPLE :

Bend a reports the known concentration of C1 in a known solution ϵ_{-} 36 mg/l \pm 3%.

The known concentration may vary from:

36 mg 1 (1-0.03) to 36 mg/1 (1+0.00) = 34.9 mg/ >> 37.1 mg/1

From the accuracy criteria, Group III, for concertations greater than 1 mg/l, the allowable deviation is 10%.

The acceptable range is:

$$\left(\frac{34.9 \text{ mc}'\text{L}}{100^{+1}}\right)^{\text{to}} \left(\frac{10}{100} + 1\right) + \left(37.1 \text{ mg/L}\right) = 31.7 \text{ mg/L to } 40.8 \text{ mg/L}$$

* Any reported value out of this range is unacceptable.

WATER BOA REVISED ACCURACY CRITERIA FOR ANALYSIS OF WATER SAMPLES

Criteria for Approval of Analyses

Allowable Deviation

GROUP 1: Sb. As, Ba. Cd, Cr. Pb. Hg. Mo. Se. U. Pb-210*, Po-210*, Ra-226*, Ra-228*, Th-230*.

The allowable deviations for the Group I constituents may be derived from the following table:

Concentration (mg/L)	Allowable Deviation (percent)
<0.01	100
20.01 to<0.1	50
20.1 to <1.0	.5
21.0	10

If a constituent is incorrectly reported as being below the detection limit, that limit will be used as the reported concentration in the above equation.

*Convert Radioactivity (pCi/L) to mass (mg/L) as follows:

RADIONUCLIDE

CONVERSION

PDPRARA	0 226 228	1.31×10 ⁻¹¹ 2.23×10 ⁻¹³ 1.01×10 ⁻¹² 3.67×10 ⁻¹² 5.15×10 ⁻¹⁸	mg/pCi mg/pCi mg/pCi mg/pCi
In	230	5.15×10	mg/pul

Group II: B, CN, H₂S, NH₄, NO₂, PO₄, Co, Cu, Ni, Ag, Sr, Sn, V, Zn, TOC.

The allowable deviations for the Group II constituents may be derived from the following table:

Concentration (mg/L)	Allowable Deviation (percent)
<0.01	200
20.01 to <0.1	100
>0.1 to 1.0	50
21	20



If a constituent is incorrectly reported as being below the detection limit, that limit, will be used as the reported concentration in the above equation.

GROUP III: C1, SO4. NO3. Na. K, Mg, Ca. F. Al, Fe, Mn, SiO2. TDS.

The allowable deviations for the Group III constituents may be derived from the following table:

Concentration (mg/L)	Allowable Deviation (percent)
<0.01	200
20.01 to <0.1	100
>1.0	10

If a constituent is incorrectly reported as being below the detection limit, that limit will be used as the reported concentration in the above equation.

EVALUATION OF CATION-ANION BALANCES

- The cation-anion balance of each field sample (not known solutions) will be computed.
- (2) If the lysis satisfies the balance criterion it is acceptable.
- (3) If the analysis does not satisfy the balance criterion, notify the water BOA TR. The laboratory will be instructed to re-analyze the sample for constituents of its own choosing.

The above steps will be completed no later than seven days after analyses are received.

(4) If the re-analysis does not satisfy the balance criterion, notify the water BOA TR. The Paboratory may choose to re-analyze the sample. The entire sample analysis is unacceptable unless it satisfies the balance criterion.

CALCULATIONS

The balance may be computed using the DBMS. The steps presented by a show how to do it by hand. An example is given at the end of this sectio

- (1) Alkalinity Calculations
 - (a) Account for uncertainty in the field alkalinity measurement. This will be done using data provided by Bendix (see memo by GRice: Splits, Blanks, Alkalinity Solutions; April 1985).

- (b) Convert the alkalinity reported as mg/L CaCO₃ to meq/L by dividing the alkalinity by 50.0. The resulting value can be summed with the other anions (in meq/L) in the balance calculation (step 3).
- (2) Convert concentrations of cations and amions from mg/l to meq/l by dividing them by constituent equivalent weight. Consider only those constituents which comprise more than 0.1% of the sample's TDS. Compute for both oxidizing and reducing conditions.

EQUIVALENT WEIGHTS OF ANIONS

All Conditions		Oxidizing Conditions	Reducing Conditions
F	19.00		
SOA	48.03		
NO2	46.00		
NO3	62.00		
Br	79.90		
CN	26.02		
нs	33.07		
5	16.03		
POA (as H2POA)	94.98		
C1	35.45		
Mo (as Mo04 *)	47.97		
	EQUIVALE	NT WEIGHTS OF CATIONS	
All Conditions		Oxidizing Conditions	Reducing Conditions
Na ⁺	22.99		
к*	39.10		
Mg	12.15		

Ca 20.04

NH4	18.04		
A1 .	8.99		•
Pin		13.74	27.47
Fe		18.62	27.92
Sr	43.81		
Co		19.64	29.45
Ni		19.57	29.36
Cu		31.78	63.55
Zn	32.69		
Sb		24.35	40.58
Ea	68.67		
Cd	56.20		
Pb		51.80	103.60
Hg		100.30	200.60
Ag	107.87		
Sn		29.67	59.34

(3) Calculate Balance:

1

meq Cations - meq Anions x100
meq Cations + meq Anions

Use the values which result in the lowest absolute values, i.e., an alkalinity value and values for oxidizing conditions, or, an alkalinity value and values for reducing conditions. If the absolute values of a sample's balance is greater than 5%, the sample analysis is unacceptable.

EXAMPLE

The analysis for GUN-D1-O10, 84/07/13 will be used.

(1a) The reported alkalinity is 323 mg/1 as CaCO₃. Bendix reported the following:

Field Analysis of Kno	own Alkalinity Solutions,	Gunnison, 6/27/84
'Field Value (mg/l as CaCO ₃)	True Value (mg/l as CaCO ₃)	Uncertainty
22.7	21.1	8
362	377	4
914	920	1

The reported alkalinity of 323 mg/l is nearest to the true value of 377 mg/l therefore, an uncertainty of + 4% will be used. The alkalinities used to calculate the balance will be 310 mg/l is CaCO₃ and 336 mg/l as CaCO₃.

(1b) Convert alkalinity values to meg/L.

310	mg CaCO ₃	x 1mmole x 2meg = 6.20 r	neq
	L	100 mg mmole	L
336	mg Caco ₃	x 1meg = 6.72 meg	
	L	50 mg L	

(2) The sample's TDS is 2520 mg/1. Only those constituents with reported concentrations greater than 2.52 mg/1 will be used to calculate the balance.

100		10		in.	641	pa
1	z.			ΕМ	Ν.	×.
1.	n		*	1	1.	w

Constituent	Concentration (mg/1)	Oxidizing Conditions (meq/1)	Reducing Conditions meq/1)
Ca Fe Mg Mn	595 26.1 44.3 7.01	29.69 1.40 3.65 0.51	29.69 0.93 3.65 0.26
K Na	6.26 19.2	0.16 0.84 36.25	0.16

A		Part 1	
# N	Т.	C 184	ς.
PV-1	-41	UR1	2
S			. m

Constituent	Concentration (mg/1)	Oxidizing Conditions (meg/l)	Reducing Conditions meg/1)
C1 S0,	19 1490	0.54 31.02	0.54
Alkalinity (CaCO3)	310;336	* 6.20	6.20

#310 mg/1 was used because it will result in the lowest absolute value in the balance.

(3) The Balance for Oxidizing Conditions is:

$$\left(\frac{36.25-37.76}{36.25+37.76}\right)100 = 2.04\%$$

For Reducing Conditions:

$$\left(\frac{35.53-37.76}{35.53+37.76}\right)$$
 100 = 3.04%

The analysis is acceptable under both oxidizing and reducing conditions.

EVALUATION OF PRECISION

The analytical precision of splits will be evaluated to supplement the accuracy and cation-anion balance criteria and aid in identifying analytical problems.

Calculations

Precision will be evaluated using the coefficient of variation of the splits.

The coefficient of variation is:

CV =S

S = Standard Deviation
$$\left[\frac{1}{n-1}\sum_{i=1}^{n}(x_{i-x})^{2}\right]^{1/2}$$

X = Arithmetic Mean = $\frac{1}{n}\sum_{i=1}^{n}x_{i}$

The following table lists allowable deviations taken from the attached accuracy criteria and maximum coefficients of variation. If these coefficients are exceeded, precision is poor. Notify the water BDA TR.

llowable Deviation (%)	CV
200	1.38
100	0.84
50	0.45
25	0.23
20	0.18
10	0.09

Example

The laboratory reports the following Mn values for 5 splits 0.03, 0.007, 0.02, 0.03, and 0.009 mg/l.

Average the reported concentrations to determine the allowable deviation. In this case it's 100%.

- X = 1/5 (0.03+0.007+0.02+0.03+0.009) * 0.0192

+(0.03-0.0192)2+(0.009-0.0192)2)31/2

- . 0.011
- $CV = \frac{0.011}{0.019} = 0.7^{\circ}$

The precision of the Mn analyses is considered good.

ADVANCED SYSTEMS DIVISION, ALBUQUERQUE OPERATIONS

TO: Distribution

FROM:

GRice

DATE: May 9, 1985

SUBJECT: [Technical Memo No. 4] Rev. 1 Splits, Blanks, Alkalinity Solutions

SPLITS

At this time, it is not possible for our water sampling teams to collect true splits for analysis of precision. This is because we have been unable to find an acceptable container which can hold five samples at once. Instead of splits, five consecutive replicates will be collected. In most instances, we will be able to treat these replicates as splits. The time at which each replicate was collected will be noted on it's field form.

BLANKS

The following blanks will be submitted to the analyzing laboratory each time a site is sampled or reagents are changed. Blanks are meant to check the purity of preservatives and consist of the preservative alone in deionized water.

PRESERVATIVE	AN AL YTE
NaOH Pellets	CN
NaOH Pellets and Zn Acetate Solution	S
H2504	NO2, NO3, 1-4, TOC
HNO3	METALS
HC1	Mo,Sb,Sn

ALKALINITY SOLUTIONS

Field measurements of alkalinity are critical and we must have a way of checking their accuracy. This will be done as follows.

Each time a site is sampled, Bendix will send three solutions of known alkalinity to the sampling team in the field. The team will measure their alkalinities and send the results to Bendix.

Bendix will submit a report giving the true values along with the team's reported values.

Any inaccuracies in the reported values must be accounted for when computing cation-anion balances.

GR/11

cc: DBetsill DMiller





JACOBS ENGINEERING GROUP INC.

AL PLICIEROLIE OPERATIONS MANUAL	SECTION	16.2.4
ALBOQUERQUE OPERATIONS MANOAL	DATE	09/18/90
	EFFECTIVE	09/28/90
CANDI THE DADON IN HATED	SUPERSEDE	S
SAMPLING RADON IN WATER	PAGE 1	OF 2

1.0 OBJECTIVE

To establish a procedure for the collection of water samples for the analyses of radon in water.

2.0 <u>REFERENCE</u>

EPA/EERF - Manual-78 Radon in Water Sampling Program.

3.0 APPLICABILITY

Radon content in water analyses will be used in health risk assessment determinations.

4.0 EQUIPMENT

Water collection vials - supplied by laboratory (two 40 ml vials per location ID).

5.0 <u>PROCEDURE</u>

5.1 Samples are to be collected in duplicate.

- 5.2 If collected at laucet with small screen aerator, remove the aerator before sample collection. Do not collect from faucet with carbon filtration system.
- 5.3 Run the water for a minimum of two minutes before collecting sample to purge the line. Use cold water.
- 5.4 Reduce the flow rate to a slow stream, trying to avoid as much agitation and turbulence as possible.
- 5.5 Place the sample vial under the faucet outlet, as close as possible to the outlet. Direct the water stream down the inside wall of the glass vial and allow it to overflow. After the vial is full (with some overflow) replace the screw cap tightly. Invert the sample vial. If there are any bubbles in the vial discard the sample and collect another sample in the same vial. Repeat this process as many times as needed to obtain a sample without an air bubble.

ALBUQUERQUE OPERATIONS GROUP INC. SECTION 16.2.4

- 5.6 Record the collection time and date on the vial label. Also record pertinent information on form 16.2.4A. Note that field parameters are not required for the analysis.
- 5.7 On the day of collection, or no later than 24 hours after sample collection, return the samples by overnight mail to the laboratory for analysis. Because the laboratories are closed on weekends collect and ship samples only on Mondays, Tuesdays and Wednesdays of each week.

Prepared	By: Jaule Cenears
Reviewed	By: Mark Miller
Reviewed	By: Bulanty Assurance Manager
Approved	Br. Alun & Alin

		Bua	
	2	8/2	~
截	肥		

ALBUQUERQUE OPERATIONS PAGE _____ OF ____

r	0	DM	1	R		2		٨	٨
r	v	nn		v	۴	6.	٠	7	n

SAMPLING RADON IN WATER

Site Identification:	
Location Identification:	
Date of Sample collection:	
Time of Sample Collection:	
Date of Sample Shipped to Lab:	
/ime of Shipment:	
Method of Shipment:	
Sample Conjected by:	
Sample collected from:	
Comments:	

JE JACOBS ENGINEERING GROUP INC.

ALBUQUERQUE OPERATIONS MANUAL	SECTION 16.3.1
	DATE 06/09/89
COMPLETION OF WELL AND SPRING INVENTORY	EFFECTIVE 11/20/89
	SUPERSEDES 05/28/87
	PAGE 1 OF 6

1.0 PURPOSE

These procedures enable the use of a uniform methodology for inventory and documentation of wells and springs located in the vicinity of UMTRA sites.

2.0 SCOPE

These procedures are applicable to members of the TAC or its contractors and subcontractors who have responsibilities and duties for the identification, acquisition, and inventory of descriptive data for each well or spring located within UMTRA Project study areas.

3.0 PROCEDURE: Data Collection Preparation

- 3.1 The location of each well or spring identified by this well and spring inventory shall be identified by the relevant Site Code and Location ID in accordance with Section 17.4.1 entitled, "Standard ocation Identification for Test Borings, Test Pits, and Monitoring ocations".
- 3.2 Each Site Hydrologist (SH) shall review the computer listing for Borehole Definitions/Test Pit Definitions (BHD/TPD) to obtain previously assigned Location Identification numbers for data collection purposes. For existing and future purposes, a specific location which has been previously assigned a Location Identification number shall permanently retain and use this number, unless this number is changed in accordance with required procedure. Otherwise, a new Location Identification number shall be assigned by the SH in accordance with Section 17.4.1.
- 3.3 Each SH shall prepare and maintain an updated map of the site vicinity showing all wells and springs located in the course of the well and spring inventory.
 - 3.3.1 Each SH shall incorporate the methodology of this procedure into all well and spring inventories and permit no data gathering efforts that do not include the Standard Location Identification system.
 - 3.3.2 The SH shall review the BHD/TPD computer listing to determine previously assigned Location Identification numbers.

JACOBS ENGINEERING GROUP INC. SECTION 16.3.1 ALBUQUERQUE OPERATIONS

PAGE 2 OF 6

- 3.4 Record all field data on the attached form entitled "Well and Spring Inventory Field Data" form. See Attachment 10.1.
- PROCEDURE: Determination of Measuring Point Description 4.0

Determine the height above or below land surface of the measuring point by use of a pocket steel tape measure graduated in 0.01-foct intervals. Determine the height of the measuring point to the nearest 0.01 foot. Enter this value along with a description of the measuring point on Attachment 10.1 as the "Measuring Point Description" (MPD).

- PROCEDURE: Water Level Measurement of Non-Flowing Wells 5.0
 - 5.1 Wetted-tape Method
 - 5.1.1 Prepare the tape for measurement by coating several feet of the lower end of a steel tape with chalk.
 - 5.1.2 Attach lead weights to the lower end of the steel tape to keep it taut.
 - 5.1.3 Lower the tape into the well until a foot or two of the chalked portion is submerged. Alternatively, use tape without weight if the well opening or pump casing clearance is too small and restricts the passage of a weight.
 - 5.1.4 Determine by experimentation the proper length req'ed to lower the tape.
 - 5.1.5 Perform measurement as follows:
 - A. Lower and hold tape at an even foot mark at the measuring point and record this tape reading.
 - B. Remove the steel tape from the well. Subtract the wetted length from the even foot mark recorded in A above and record this on Attachment 10.1 as "Static Water Level Depth" (SWLD).
 - 5.2 Electric Sounder Method (M-scope)

An electric sounder consists of a contact electrode that is suspended into a water body by an insulated electric cable from a reel. The sounder has an ammeter, a buzzer, or other closed circuit indicator attached. The indicator shows a closed circuit and flow of current when the electrode touches the water surface. Calibrate the electric sounder(s) used in this survey by measuring each interval and re-marking them where necessary. Measure the water levels with an electric sounder as described below.

5.2.1 Switch on.
JE JACOBS ENGINEERING GROUP INC. SECTION 16.2.1

- 5.2.2 Lower the electric cable into the well unit until the ammeter or buzzer indicates a closed circuit. Raise and lower the electric cable slightly until the shortest length of cable that gives the maximum response on the indicator is found.
- 5.2.3 With the caule in this fixed position, note the length of cable at the reasuring point.
- 5.2.4 Since the electric cable is usually graduated in five-foot intervals, use a pocket steel tape measure, graduated in 0.01-foot intervals, to interpolate between consecutive five-foot marks. Care must be taken that the steel tape measurements are added to the nearest graduated mark footage value when the water level hold point is above the mark. The steel tape measurements are subtracted when below the nearest mark. Determine this total distance to the nearest 0.01 foot and record this value on Attachm . 0.1 as the SWLD.
- PROCEDURE: Water Level Measurement of Firming wells 6.0
 - 6.1 Preferred Method

Measure the water level (piezometric surface) in a flowing well with a manometer or a pressure gauge as described below.

- 6.1.1 Shut off all flow of the well to any uses (i.e., domestic supply, livestock watering tank, etc.) other than at the well itself. Valves are usually present near the well for this purpose. If it is impossible to shut off the flow of the well other than at the well itself, record this observation on Attachment 10.1 and skip the remaining procedures.
- 6.1.2 Measure the water level in flowing wells with a manometer. This method is shown schematically in Attachment 10.2.
- 6.1.3 Record the elevation at the top of the well casing as the "MP Elevation" (MPE).
- 6.1.4 Attach a vertical section of the flexible transparent plastic tubing to the top of the well casing using various couplings carried into the field for this purpose.
- 6.1.5 Extend the plastic tubing vertically above the well casing until water ceases to flow from the end of the tubing and stands at a fixed height in the tubing.
- 6.1.6 Record the height of the water in the plastic tubing above the measuring point as SWLD.

ALBUQUERQUE OFERATIONS

PACE 4 OF 6

6.1.7 Add SWLD to MPE and record this as a "Static Water Level Elevation" (SWLE).

6.2 Alternative Method

The water level in a flowing well could also be measured by using a pressure gauge as described below.

- 6.2.1 Record the elevation at the top of the weil casing as the "MP Elevation" (MPE).
- 6.2.2 Attach a pressure gauge to the well casing using the various couplings carried into the field for this purpose.
- 6.2.3 If the gauge is calibrated to indirate pressure in feet of water, record the gauge readings (to the nearest one-half foot) as SWLD. If the gauge is calibrated in pounds per square inch (psi), multiply the gauge reading by 2.3 to convert it to feet of water, and record this number as SWLD.
- 6.2.4 Add SWLD to MPE, and record this as a "Static Water Level Elevation" (SWLE).
- 7.0 PROCEDURE: water Level Measurement of Springs

The water level of any spring will be identified as that land elevation above mean sea level (MSL). MSL is determined from the inventory location as plotted on the relevant USGS topographic map at the intersection of the spring and the land surface.

- PROCEDURE: Discharge Measurement of wells 8.0
 - 8.1 Determine the discharge from a weil in gallons per minute (gpm) by reading a water meter, if the well is sp equipped. Alternatively, measure the time required to fill a buciet of known volume.
 - 8.2 Determine the discharge for most well; that are pumping when visited during the field survey by using a bucket or drum and a stopwatch. Fill the bucket or drum of known volume (gallons) with the water that is discharging from the well. Measure the time, t (seconds), required to fill the bucket. This discharge, Q (gpm), is calculated by :

Q = 60 V/t

where, Q = discharge in gpm, ¥ = volume of container in gallons t = time required to fill the bucket in seconds

JACOBS ENGINEERING GROUP INC. SECTION 16.3.1 ALBUQUERQUE OPERATIONS

The condition of a well at the time the water level is measured must be recorded in the "remarks" column of Attachment 10.1. Such conditions may include:

- 8.2.1 Well was pumping during water level measurements.
- 8.2.2 Well was pumping intermittently during water level measurement.
- 8.2.3 Well was pumping two hours before water level measurements.
- 8.2.4 Owner reports well has not been pumping for four days, etc.
- 8.3 If well is pumping at the time of measurement, it will be necessary to record the static water level at a later time. This would be done by taking another measurement after pumping has ceased and sufficient time has been allowed for recovery.

PROCEDURE: Discharge Measurement of Springs 9.0

The natural variability of springs as to location, ease of access, discharge, etc. precludes the establishment of rigid procedures to measure spring discharge. The following list of spring discharge measurement methods are available to the investigator. Based on field conditions, the investigator will determine the method best suited to the characteristics of the spring under investigation.

- 9.1 Erect a V-notch weir and measure spring discharge. The application of this method is covered in a number of standard references. See Reference 11.1. While this method provides reliable estimates of spring discharge, it is labor intensive, time-consuming, and not suited to all applications.
- 9.2 Determine the cross-sectional area of the spring discharge outlet. Introduce a floating object into an unobstructed reach of the spring discharge channel and time the object's progress over a fixed distance; this will yield an approximated spring discharge velocity. The product of the cross-sectional area and spring discharge velocity is an approximate measurement of the spring discharge.
- 9.3 Channel the spring discharge into a container of a known volume and record the time needed to fill the container. This will yield an approximate measurement of spring discharge.
- 9.4 Make a visual, qualitative estimate of spring discharge. This would include remarks such as very slow, flowing, seepage, moderate, etc. This is possibly the least accurate of the methods discussed.



10.1 Well and Spring Inventory Field Data (WSIFJ) Form.

10.2 Figure 1 Schematic Diagram Illustrating Water Level Measurement in a Flowing Well.

11.0 REFERENCES

11.1 National Handbook of Recommended Methods for Water Data Acquisition, 1977. Office of Water Data Coordination, U.S. Geological Survey, MS-417, National Center, Reston, Virginia 22092.

12.0 PROCEDURE REVIEW AND APPROVAL

Prepared By: eser Reviewed By: asx Manage Reviewed By: 17 4 Assurance Manager Quality Approved by: Protei Manager

WELL AND SPR	ING INVENTORY FIELD DATA FO	DRM (WSIFD)
DATE	FIELD REPRESENTATIVE (S))
SITE CODE		
LOCATION ID.	NORTHING	TOPO MAP
	EASTING	D SURVEY
DESCRIPTION OF LOCATION	(IF NEEDED)	
	METHOD OF LIFT	
DISCHARGE WELL	DISCHARGE	
D OBSERVATION WELL	HOW DETERMINED	
LAND SURFACE DATUM		D SURVEY
MEASURING POINT ELEVATIO	N D TOPO MAP	D SURVEY
MP DESCRIPTION		
STATIC WATER LEVEL DEPTH	H M-SCOPE	D WETTED TAPE
PUMPING WATER LEVEL DEP	TH D PRESSURE GAUGE	D MANOMETER
STATIC WATER LEVEL ELEVA	ATION	
PUMPING WATER LEVEL ELE	VATION [] TOPO MAP	SURVEY
WATER LEVEL REMARKS		
OWNER		
ADDRESS	n the chair die Line and C. M. M. M. M. S. State in the second strength of the second strength of the second st	
USE OF WELL DHUMAN C	ONSUMPTION D LIVESTOCK	IRRIGATION
NO. OF PEOPLE SERVED		
SPRING DATA		
DISCHARGE	HOW DETERMINED .	
WATER LEVEL ELEVATION	D TOPO MAP	SURVEY



#	na that an indicate the second state of the second of the second state of the second s	
	FIELD REPRESENTATIVE (S)	
LOCATION ID.	NORTHING	TOPO MAP
DESCRIPTION OF LOCATION (IF	EASTING	D SURVEY
D WELL DATA	METHOD OF LIFT	
D LISCHARGE WELL	DISCHARGE	
OBSERVATION WELL	HOW DETERMINED	
AND SURFACE DATUM	D TOPO MAP	D SURVEY
ALASURING POINT ELEVATION	D TOPO MAP	C SURVEY
AP DESCRIPTION		
TATIC WATER LEVEL DEPTH _	M-SCOPE	D WETTED TAPE
UMPING WATER LEVEL DEPTH	D PRESSURE GAUGE	D MANOMETER
UMPING WATER LEVEL ELEVA	TION [] TOPO MAP	D SURVEY
DDRESS		
SE OF WELL D HUMAN CON	SUMPTION D LIVESTOCK	DIRRIGATION
O. OF PEOPLE SERVED		
SPRING DATA		
DISCHARGE	HOW DETERMINED	
WATER LEVEL ELEVATION _	TOPO MAP	D SURVEY

JEG-AL-ENG-25 (5/AS)

AL PLICIEDOLE ODEDATIONS MANUAL	SECTION 17.1.1
ALBOQUENQUE OFENAINAS INANUAL	DATE 3/30/87
	EFFECTIVE 04/29/88
ANALYSIS OF DATA AND COMPILATION OF	SUPERSEDES 8.3.2 8/8/85
THE SITE CHARACTERIZATION AFFENDIA	PAGE 1 OF 3

1.0 PURPOSE

The purpose of this document is to establish procedures for the analysis and documentation of data on the UMTRA Project.

2.0 SCOPE

In order to prepare remedial action plans (RAP's) for UMTRA Project Sites, data are collected and analyzed. The data as collected, analyzed, and interpreted form the basis for the formulation of the Site Conceptual Designs and the final Site Designs of approved remedial actions. The Processing and Disposal Site data and other pertinent data are formally documented in the Site Characterization Appendix of the draft and final RAP.

This procedure delineates the responsibilities for analyzing data and compiling the Site Characterization Appendix. Also described are the methods of analyzing data, documenting the analyses, and checking, reviewing, and approving data analyses.

3.0 PROCEDURE

- 3.1 Analysis of data relevant to a particular aspect of the site shall be done by, on behalf of, or under the direction of the discipline site team member responsible for that aspect of the site.
- 3.2 Analysis of data shall be undertaken at the direction of the Site Manager, who is responsible for the schedule and budgets involved in compiling the site RAP. The Task Manager shall exercise delegated authority to control manhour budgets. In the event that the analyses do not relate to a specific site, the Task Manager of the section responsible for collecting and collating the data shall direct the analysis.
- 3.3 Data analyses for specific sites shall be done in accordance with the directions and advice of the responsible Task Manager. The Task Manager or his designated representative shall approve all analyses, shall sign the analyses, and shall be responsible for the adequacy and correctness of all data analyses.
- 3.4 The lead page of a data analysis shall indicate the Project, originator, subject, and date. The Analysis cover sheet to be used is given as Exhapit A.

JACOBS ENGINEERING GROUP INC. SECTION ______

١,

- Analyses shall be numbered in accordance with the numbering 3.5 system set out in Exhibit B, or in accordance with a numbering system specifically compiled for a particular site to accord with the particular requirements and facts at that site.
- 3.6 Calculations undertaken prior to, in conjunction with, or pursuant to data analyses shall be performed and documented as described in Standard Operating Procedure Number 17.1.2.
- 3.7 Analysis of data shall be done in accordance with the methods and approaches described in the DOE Technical Approach Document (TAD). If an appropriate or applicable method is not given in the TAD, a method that is standard in the relevant technical discipline shall be used.
- Analyses shall be checked by qualified person other than the 3.8 person who performed the analysis. The Task Manager shall designate the checker, and shall determine the amount of checking necessary. The checker shall evaluate the consistency, applicability, and defensibility of all references, technical concepts, assumptions, calculations, analyses, and conclusions. The Task Manager or his designated representative shall approve the analyses.
- At the direction of the Task Manager or at the specific request 3.9 of the Site Manager, analyses may be reviewed. A review may be requested or ordered if the analyses involve unusual approaches or methods, or pertain to significant aspects of the work. Reviews will be performed in accordance with the requirements of Standard Operating Procedure Number 9.2.6 and, if applicable, 8.5.2.
- 3.10 Conflicts between the person compiling the analysis and a reviewer or checker shall be resolved by the Task Manager and, if necessary, by the Project Manager.
- 3.11 Analyse and documentation of the checking and review process shall be filed with Project Document Control. Documents recording the analyses shall be legible and suitable for reproduction and filing.
- 3.12 When a computer is used in an analysis, only appropriately verified, validated, and documented programs approved by the TIMS Manger and the Task Manager shall be used.
- 3.13 As directed by the Site Manager, or advised by the Task Manager, the results of analyses shall be incorporated into or reported and documented in the Site Characterization Appendix of the RAP.
- 3.14 The format of the Site Characterization Appendix shall be arranged to meet the specific requirements of each site.

SECTION _______ 17.1.1 PAGE _____ OF _____

Generally, the approach adopted in previous successful Site Characterization Appendices will be adopted and modified as appropriate. The Site Manager, in consultation with Task Managers, may establish the format of the Site Characterization Appendix for reports for which he is responsible.

- 3.15 As directed by the Site Manager or advised by the Task Managers, the Site Characterization Appendix shall be reviewed by Peer Reviewers at appropriate stages of compilation of the document.
- 3.16 Information derived from calculations may be incorporated into the Site Characterization Appendix in accordance with the provisions of this Standard Operating Procedure.

	71/10/1
Prepared B	y: Aluldurth
D	74/11/1
Kevlewed b	Ask Monager
Reviewed B	y: Quality Assurance Manager
Approved B	V Project Hanager

	JE		RING G	ROUP INC.	_	Section 17	.1.1
•	<u>c</u>	ALCULATION CO	OVER S	SHEET			
CALC NO)	DISCIPLINE			NO. OF	SHEETS	
PROJECT	T:						
SITE							
FEATURE	E:						
SOURCE	S OF DATA:						
the second se		REFERENCES		an ra sense in real sector and second and	Received a state of the state o	Informer and the state of the s	
SOURCE	S OF FORMULAE &	NEFENEROED.					
SOURCE	S OF FORMULAE &	HEFERENCED.					
SOURCE	S OF FORMULAE &	HEFERENCES.					
SOURCE	S OF FORMULAE &	HEFERENCES.					
SOURCE	S OF FORMULAE &	HEFERENCES.					
SOURCE	S OF FORMULAE &						
RELIMIN	ARY CALC.	FINAL CALC.	UPERS	EDES CALC	D. NO		
RELIMIN	ARY CALC.	FINAL CALC.	UPERS	EDES CALC	D. NO		
PRELIMIN	ARY CALC.	FINAL CALC.	UPERS	EDES CALC	D. NO		
RELIMIN	ARY CALC.	FINAL CALC.	UPERS	EDES CALC). NO		

11

•

۱

JE JACOBS ENGINEERING GROUP INC. SECTION _______

Subgrade Limits of Contamination Subgrade Material Properties Subgrade Material Quantities Tailings Material Quantities Tailings Material Quantities Tailings Long-Term Moisture Content Tailings Limits of Contamination Embankment Slope Stability Embankment Liquefaction and Cyclic Mobility Embankment Settlemenc	01 02 03		01 02 03 01 02 03 04
Tailings Material Properties Tailings Material Quantities Tailings Long-Term Moisture Content Tailings Limits of Contamination Embankment Slope Stability Emtankment Liquefaction and Cyclic Mobility Embankment Settlemenc	02 03		01 02 03 04
Embankment Slope Stability Embankment Liquefaction and Cyclic Mobility Embankment Settlemenc	03		
		:	01 02 03
Radon Barrier Material Properties Radon Barrier Material Quantities Radon Barrier Long-Term Moisture Content Radon Barrier Cover Thickness Rudon Barrier Diffusion Coefficient	04		01 02 03 04 05
Erosion Protection for Top and Side Slopes Erosion Protection for Ditches Erosion Protection Due to Off Pile Flooding Erosion Protection Rock Durability Properties	05		01 02 03 04
Erosion Protection Quantities Restoration Material Quantities Restoration Material Properties	06	-	05 01 02
Waste Water Retention Basin & Temporary Utility Ditches Waste Consumption	07	-	01 02
Surface Water Flood Studies PMP Calculations	08		-1



AL PLIQUERQUE OPERATIONS MANUAL	SECTION	17.1.2
ALBOQUERQUE OFERATIONS WANDAL	DATE	05/11/90
	EFFECTIVE	06/05/90
COMPILATION OF DESIGN CALCULATIONS	SUPERSEDES 06/20/89	
	PAGE 1	OF 3

1.0 PURPOSE

The purpose of this document is to establish procedures to be used to compile Design Calculations for the UMTRA Project.

2.0 SCOPE

Formulation of a design for the UMTRA Project involves analyzing data and completing calculations to assess and support the design. The Calculations are kept on record in Document Control to support design decisions. They are also submitted to review agencies and other affected parties as needed. This procedure describes responsibilities for compiling the Calculations and also sets out the methods and approaches involved in compiling a typical Calculation.

3.0 PROCEDURE

- 3.1 The Calculations relevant to a particular aspect of the formulation of the design or disposal cell performance shall be compiled by or under the direction of the site team member responsible for that aspect of the design or disposal cell performance.
- 3.2 Calculations shall be compiled at the direction of the Site Manager, who is responsible for the schedule and budgets involved in compiling the Calculations. The Task Managers, as Work Package Managers in terms of the Cost/Schedule Control System, shall exercise delegated authority over mannour budgets.
- 3.3 Calculations shall be performed and compiled in accordance with the directions and advice of the responsible Task Manager. The Task Manager shall approve, sign, and is responsible for the adequacy and correctness of the Calculations.
- 3.4 The lead page of a Calculation shall indicate the Project, originator, subject, and date. The Calculation cover sheet to be used is given as Exhibit A.
- 3.5 Calculations shall be numbered in accordance with the numbering system set out in Exhibit B, or in accordance with a numbering system specifically compiled for a particular site to accord with the particular requirements and facts at that site.

JACOBS ENGINEERING GROUP INC. SECTION 17.1.2 ALBUQUERQUE OPERATIONS

PAGE 2 OF 3

- 3.6 Analyses of data undertaken in connection with or in support of Calculations shall be performed and documented as described in Standard Operating Procedure No. 17.1.1.
- 3.7 Calculations shall be done in accordance with the methods and approaches described in the COF Technical Approach Document (TAD). If an appropriate or applicable method is not given in the TAD, a method that is standard in the relevant technical discipline shall be used.
- 3.8 Examples of acceptable Calculations are given in Exhibits D. E. and F.
- 3.9 Calculations shall be checked by a qualified person other than the person who performed the calculations and analyses. The Task Manager snall designate the checker and shall determine the extent of checking and review that is necessary. The checker snall evaluate the consistency, applicability, and defensioility all references, technical concepts, assumptions, of calculations, and conclusions.
- 3.10 At the discretion of the Task Manager or at the specific request of the Site Manager, calculations may be reviewed. A review may be ordered or requested if the calculation involves unusual approaches or methods, or results in significant or disputed conclusions. Reviews will be performed in accordance with the requirements of Standard Operating Procedure Number 9.2.6 and. if applicable, 8.5.2.
- 3.11 Conflicts between the person compiling the Calculations and the reviewer shall by resolved by the Task Manager and, if necessary, by the Project Manager.
- 3.12 Documents recording Calculations shall be legible and suitable for reproduction and filing. Calculations and documentation of the review process shall be filed with Project Document Control.
- 3.13 When a computer is used by the compilation of Calculations, only appropriately verified, validated, and documented computer programs approved by the Task Manager shall be used. Reference Standard Operating Procedure Number 8.3.3.
- 3.14 In compiling Calculations, the methods, approaches, and layout adopted in previously accepted Calculations shall be considered or followed as appropriate.

JACOBS ENGINEERING GROUP INC. SECTION ALBUQUERQUE OPERATIONS

SECTION ___________ PAGE _______________OF __________

- 3.15 In the case of calculations compiled by staff of the Engineering Department, the Site Engineer shall collate all engineering calculations into one or more stand-alone volume(s). The Site Engineer shall be responsible (via delegated authority from the Task Manager) for delivering to Document Control, a copy of the collated Engineering Calculation Volume. Delivery shall be effected by a Document Transmittal Notice.
- 3.16 In case of calculations compiled by staff of the Hydrological Department, the Site Hydrologist shall collate all Hydrologic calculations into one or more stand-alone volume(s). The Site Hydrologist shall be reponsible (via delegated authority from the Task Manager) for delivering to Document Control, a copy of the collated Hydrologic calculation volume. Delivery shall be effected by a Document Transmittal Notice.

Prepared	ву: ()	Allalla
Reviewed	ву: _С	Task Manager - Engineering
Reviewed	ву:	But R Beaulan- Quality Assurance Manager
Approved	87:	Project Manager

Hydrology Manager Rad H&S

	JE		RING GR	ROUP INC	* .		
		CALCULATION CC	VER S	HEET			
CALC N	(0	DISCIPLINE			NO. O	F SHEETS	
PROJEC	эт:						
SITE:							*******
FEATUR	RE:			***			
SOURC	ES OF DATA:						
•							
SOURC	ES OF FORMULAE 8	& REFERENCES:					
SOURC	ES OF FORMULAE 8	REFERENCES:					
SOURC	ES OF FORMULAE 8	REFERENCES:					
SOURC	ES OF FORMULAE &	REFERENCES:	JPERSED	DES CALC	C. NO		
SOURC	ES OF FORMULAE &	FINAL CALC.	JPERSED	DES CALC	C. NO		
SOURC	ES OF FORMULAE &	REFERENCES:	JPERSED	DES CALO	C. NO		
RELIMIN	ES OF FORMULAE 8	REFERENCES:	JPERSED	DES CALO	C. NO		

JE JACOBS ENGINEERING GROUP INC. SECTION _____

EXHIBIT B DESIGN FEATURES			
Design Features	Major Design Feature Class No.	Sub-Design eature No.	
Geotechnical feature Subgrade limits of contamination Subgrade material properties Subgrade material Quantities			
Tailings material properties Tailings material quantities Tailings long-term moisture content Tailings limits of contamination Off-Pile Ra-226 source term	.02		
mbankment slope stability mbankment liquifaction and cyclic mobility mbankment settlement	03	01 02 03	
Radon barrier material properties Radon barrier material quantities Radon barrier long term moisture content Radon barrier cover thickness Radon barrier diffusion coefficient Radon barrier sensitivity analysis			
Trosion protection for top and side slopes Trosion protection for ditches Trosion protection due to off pile flooding Trosion protection rock durability properties Trosion protection quantities	05	01 02 03 04 05	
Restoration material quantities			
laste water retention basin & temporary utility ditches laste consumption	07	01	
Surface water flood studies PMP calculations			

JE JACOBS ENGINEERING GROUP INC. SECTION _____

DESIGN FEATURES (Continued)		
Design Features	Major Design Feature Class No.	Sub-Design Feature No.
Health effects from radon/radon daughter innalation Health effects from gamma radiation exposur Health effects from radioactive particulate innalation Health effects from radioactive water ingestion Health effects from radioactive food ingestion	re	
Miscellaneous topics - Subsidence Pile configuration analysis		
Disposal Cell Hydrologic and Geochemical Properties Material properties Relation of hydraulic conductivity to moisture content Hydraulic gradient Seepage flux through the cover Transient tailings drainage Seepage flux from the disposal cell Source Concentration; weighted lysimete back calculated from groundwater benear tailings, Batch-leach or column extract analyses Geochemical modeling of Speciation and saturation indices of tailings fluids.	er data,	
Soil Hydrologic and Geochemical Properties. Material properties Relation of hydraulic conductivity to moisture content Hydraulic gradient Travel time of seesage to groundwater. Geochemical modeling; speciation and saturation indices of soil solution, pr buffering or mixing of tailings fluids and soil geochemistry	1	01 02 03 04

N.C. N. CALMAN

JE JACOBS ENGINEERING GROUP INC. SECTION ______

PAGE OF

EXHIBIT B DESIGN FEATURES (Continued)		
Design Features	Major Design F∉ature Class No.	Sub-Desigr Feature No,
Geochemical attenuation of contamina by soil	tion	
Material properties Slug test analysis Pumping test analysis Dispersivity analysis Distribution coefficient analysis Hydraulic gradient Average linear groundwater velocity. Potential for vertical flow of ground well yield (Class III groundwater) Statistical analysis of background water quality Statistical aralysis of existing groundwater contamination Bivariate analysis of water quality. Trilinear or Stiff Diagram analysis water quality type Geochemical modeling; speciation and saturation indices of groundwater, pi Duffering or mixing of tailings fluid with aquifer geochemistry	dwater	01 02 03 04 05 05 05 05 07 08 09 10 10 11 12 13
Water Resources Protection Strategy Identification of hazardous constitue Concentration limits Simulation of concentration of hazard constituents at POC in uppermost aqui Simulation of concentration of hazard constituents at point of exposure	ents dous ifer dous	01 02 03
Groundwater Restoration Passive flusning Volume of contaminated groundwater	1ô	

JACOBS ENGINEERING GROUP INC. SECTION ALBUQUERQUE OPERATIONS

PAGE OF

EXHIBIT C - CONTENTS OF A STANDARD CALCULATION

1.0 INTRODUCTION

This exhibit, a part of the Standard Operating Procedure for compilation of Calculations, describes the contents of a standard Calculation. A standard Calculation should contain at least those sections described in this Exhibit. It is recognized that in specific instances some sections may not be incorporated; however, that should be done only for good and sufficient reason. Additional Sections as warranted by the nature of the Calculations being compiled may be used as necessary to compile a complete and comprehensive Calculation.

2.0 CONTENTS OF THE STANDARD CALCULATIONS

- 2.1 Problem Statement or Purpose of Calculation: This section should describe what design problem is to be addressed, how the calculation fits into the formulation of the SCD, why the calculation is performed, and in what manner the results may be used.
- 2.2 Overview of the Method and Procedures to be Adopted: This section should describe the methods and procedures, or the models that are to be used in the analyses, evaluations, or calculations. For methods that are described in the Technical Approach Document. reference to that document is sufficient. Methods not included in the Technical Approach Document require a more detailed discussion.
- 2.3 Assumptions: Provide a series of tullet items listing the assumptions used as input to the models or methods used.
- 2.4 Material Properties: List all material properties used in the calculations. This section will be necessary for a calculation involving soils, but may not be needed for a calculation involving pile layout or quantities.
- 2.5 Data Source: List the sources of field, laboratory, or reference data that are to be used in the analyses.
- 2.6 Solution Calculations and Analyses: This section should describe and detail the calculations actually performed. Sufficient detail should be given for a checker or reviewer to follow the work done without recourse to the originator.
- 2.7 Sensitivity Analyses and Parametric Evaluations: If precise parameter values are not known or are likely to vary during the design life of the facility, sensitivity analyses and/or parametric evaluations shall be undertaken. Such analyses and evaluations should consider a feasible range of values for relevant parameters. The range may be derived from statistical

JACOBS ENGINEERING GROUP INC. SECTION____ ALBUQUERQUE OPERATIONS

PAGE OF

analysis of available data or the range may be defined on the basis of judgmental assessment of the likely or feasible range of values. The effect of differing results on the design or performance of the disposal cell shall be considered in deciding on reasonable design details.

- 2.8 Conclusions and Recommendations: The results of the calculations should be clearly stated. Tables, graphs, or both should be used to summarize the results. If the results are to be used as inputs to another calculation, that fact should be noted, and the results that are to be used should be clearly identified. If the calculation includes a parametric or sensitivity analysis, this should be noted and conclusions stated. The effect that the calculation has on the design layout should be noted. Recommendations for further analyses or design evaluations should be noted.
- 2.9 Results: This should list the main, relevant results.
- 2.10 References
- 2.11 Attachments: Computer program listings or output. Laboratory data.

EXHIBIT D. EXAMPLE CALCULATION GEDTECHNICAL CALCULATION: SETTLEMENT ANALYSIS

Purpose: This pile settlement calculation evaluates the rate and magnitude of pile settlement that will result when the pile is shaped to promote positive drainage from the top surface. Reshaping will be done by placing windblown soils on the top of the pile. The rate of settlement is required so that the need for pile monitoring following remedial action construction may be evaluated. The magnitude of settlement is required for further assessment of the potential for cracking of the radon barrier or the development of an unacceptable pile top surface profile.

SECTION _____

PAGE OF

Method Overview: Standard Engineering settlement methods (as incorporated into the "CONSOL" computer program) are used to calculate the rate and magnitude of settlement of the pile for the given loadings. Sensitivity analyses are performed to assess the effect of variations in the design parameters (more specifically, the coefficient of consolidation of the tailings) on the rate and magnitude of settlement. The calculation method adopted models the tailings as a series of layers each with distinct settlement characteristics.

Assumptions: The main assumption involved in the calculations is that the alluvial soil layer beneath the tailings pile will act as a drainage layer. A second significant assumption is that all the tailings are saturated. Other assumptions are

Material Characteristics: This section lists material types and soil properties.

Data Sources: This section notes that the major sources of data about the tailings are the laboratory test results, that specified tailings properties are taken from a listed reference work, and that the layout of the pile as modeled in the analyses is based on maps obtained from the previous owner of the pile.

Solution and Analysis: In this section the detailed calculations are described. All steps are listed. Figures showing the components and layout of the pile and the layers as modeled in the analyses are given. The different cases that are analyzed are tabulated. Details of the computer program and the main equations are listed. Graphs showing the results of the sensitivity analyses are provided.

Discussion: This section discusses the consitivity analyses and the effect they have on determining whether special design measures are required to deal with settlement of the pile. The discussion focuses on whether the main conclusions would be different if there is a significant difference between actual field tarlings properties and those measured in the laboratory.



E,

4

JACOBS ENGINEERING GROUP INC. JE JACOBS ENGINEERI ALBUQUERQUE OPERATIONS

SECTION _____ PAGE_____ OF _____

Conclusions: This section concludes that the rate of settlement is such that all significant settlement will occur while the tailings and other soils that will be used to reshape the pile are being placed. The sensitivity analyses indicate that even if extreme values of the tailings properties are assumed, this conclusion remains valid. It is concluded that the magnitude of settlement is shown to be such that a further assessment of the potential for cover cracking is warranted: The calculation from which this assessment is made is referenced.

PAGE ____ OF ____

EROSION PROTECTION DESIGN

Purpose: The purpose of the calculation is to determine the size of rock required on the top of the pile to prevent erosion of the radon cover in the event of the occurrence of the Probable Maximum Precipitation on the pile.

Design Methods and Procedure: The design method used is the Safety Factor Method as described in the Technical Approach Document and other referenced publications.

Design Discussion and Assumptions: This section references a figure of the layout of the pile and describes the flow directions that will occur in the event of the PMP. The maximum topslope length is given. Note is made that the design of rock is not for flow from flooding around the pile. The assumption that adequate sources of durable rock of the required size will be available is noted.

Material Properties: The angle of friction of the rock as estimated from published data, and the rock specific gravity as assumed are listed and discussed.

Data and Data Sources: The section references the calculation in which the pile area and the site PMP are calculated. Values of these parameters to be used in the calculation are given.

Solution - Calculation and Analysis: This section, which comprises the bulk of the Calculation, describes in detail the application of the method, now the time of concentration is determined, the determination of the volume of flow within the voids of the rock layer, the estimation of the value on Manning's n factor, and a tabulation of the input parameters to the computer program.

Conclusions and Results: This section lists the design flow velocities, the discharge quantities, and the required erosion protection rock sizes and possible gradations. Suitable layer thicknesses are given.

Attachments: The RIPRAP computer program listing, data input, and resultant output printings are given.

JACOBS ENGINEERING GROUP INC. SECTION ALBUQUERQUE OPERATIONS

PAGE OF

EXHIBIT F EXAMPLE CALCULATION

DISPOSAL CELL GEOCHEMICAL PROPERTIES: SOURCE U CONCENTRATION

Purpose

In absence of lysimeter data, this calculation will be used to provide a conservative estimate of uranium (U) concentration in tailings fluid. This concentration will be used to calculate flux of U below the tailings pile. This flux is used to calculate concentration of U in groundwater of the uppermost aquifer beneath the pile in order to determine if U concentration exceeds the EPA regulated concentration limits.

Overview of Method/Procedure

Using the leachable uranium concentration in tailings and the concentration of fluid in tailings, calculate concentration of uranium in the tailings fluid. This is an alternate way to estimate the source concentration of contaminants when direct date, such as from analysis of lysimeter water samples, are not available. Because the entire amount of leachable constituents is assumed to be concentrated in the pore fluid, this calculation provides a very conservative estimate,

Assumptions

- o Cnemical conditions in the tailings fluid are similar to those in the leach solution.
- o The entire amount of leachable U is present in the tailings fluid.
- o Density of the tailings fluid is same as that of the leach water, l g/ml.

Material Properties

N/A (Can skip items not applicable)

Data Sources

The leaching and tailings fluid content data are from the reports referenced on the Calculation Cover Sheet.

Calculation

- a) Concentration of leachable uranium in tailings = 1.5 g/g.
- b) If total mass of tailings is M gram, total amount of leachable uranium = 1.5 M q.

JACOBS ENGINEERING GROUP INC. SECTION ALBUQUERQUE OPERATIONS

PAGE OF

- c) Fluid (moisture) content = 10% (wt); therefore, total amount of fluid = 0.1 Mg.
- d) Assuming fluid density = 1 g/ml, total volume of fluid = 0.10 M ml.

From b and d above, concentration of U in tailings fluid =

 $\frac{1.5 \text{ M} \text{ g}}{0.10 \text{ M} \text{ m}} = 15 \text{ g/m} = 15 \text{ mg/l}$

Conclusions and Recommendations

- 1. Estimated U concentration in the tailings fluid, when the tailings fluid concentration is 10% (wt), is 15 mg/l.
- This is a highly conservative value since all of leachable U is assumed to be concentrated in 10% pore fluid.
- 3. This estimate should be substantiated whenever possible, for example, using U concentration in lysimeter samples.

Results

See Calculation above.

References

See source of data.

Attachment

None.



AL PLIOUEDOUE OPEDATIONS MANUAL	SECTION 17.2.1
ALBOQUERQUE OFERATIONS WANDAL	DATE 3/30/87
CONCEPTUAL DESIGN	EFFECTIVE 04/29/88
	SUPERSEDES 8.5.0/8.5.1
	PAGE 1 OF 3

PURPOSE 1.0

The purpose of this document is to describe the Procedures involved in compiling the Site Conceptual Design (SCD) for the UMTRA Project Sites.

2.0 SCUPE

An SCD is prepared for each UMTRA Project site. The SCD is described in a section of the draft Remedial Action Plan that is compiled to describe remedial actions at the site. This Procedure describes the methods and the steps involved in formulating and documenting an SCD. Also described are the responsibilities of the various members of the Site Team involved in formulating and documenting the SCD.

PROCEDUKES 3.0

- 3.1 The Task Managers responsible for the Technical Departments involved in formulating an SCD assign technical staff to the Site Team.
- The Site Manager leads the Site Team and is responsible for 3.2 organizing and controlling the schedule and budget involved in formulating the SCD. The Task Managers exercise delegated authority over manhour budgets.
- Table 17.2.1 lists the Technical Disciplines normally 3.3 represented in a Site Team.
- The Site Team, as directed by the Site Manager, meets to 3.4 consider site characterization data available and makes recommendations for the collection of additional site characterization and other relevant data required to formulate an SCD. Site Characterization and other data are analyzed and documented as described in Standard Operating Procedure Number 17.1.1. As adequate site characterization and other date are obtained and reviewed, the site design team formulates the SCD.
- 3.5 The Task Managers, at appropriate times during data collection and the formulation of the SCD, consider with their technical staff activities, recommendations, and contributions to the formulation of the SCD.

JE ALBUQUERQUE OPERATIONS GROUP INC. SECTION 17.2.1

1

- 3.6 The Site Team, as directed by the Site Managers and as advised by the Task Managers, compiles an SCD by: (1) formulating alternative disposal options; (2) evaluating the alternatives; and (3) selecting the alternative option which best meets the design criteria, which is economical, and which is acceptable to the DUF.
- 3.7 The SCD is documented in the SCD section of the draft Remedial Action Plan. Text and Figures are used as appropriate to describe and document the SCD.
- The site design team members compile design calculations to 3.8 support the SCD. Design calculation compilation is described in Standard Operating Procedure Number 17.1.2 of the Albuquerque Operations Manual.
- In compiling the SCD, account is taken of: Design Criteria as 3.9 documented in NRC, EPA, and UMTRA Project related documents; UMTRA Project technical approaches; Project requirements and plans as documented in appropriate Management plan, or procedures; standard practice in the fields of expertise involved in the formulation of the SCD; and approaches and details adopted and successfully implemented on other SCD and final Remedial Action Plans and construction works.
- 3.10 The format of the SCD section of the draft Remedial Action Plan will be arranged to meet site needs. Generally, the approach adopted in previous, successful draft Remedial Action Plans will be adopted and modified as appropriate. The Site Manager, in consultation with Task Managers, may establish for his site the appropriate format for the draft Remedial Action Plan.
- 3.11 As directed by the Site Manager or advised by the Task Managers, the SCD will be reviewed by Peer Reviewers (Reference S.D.P. 8.5.2) at appropriate stages of the formulation and documentation of the SCD.
- 3.12 As directed by the Site Manager, or as advised by the Task Managers, the SCD will be considered in Value Engineering workshops at appropriate stages of the formulation of the Remedial Action Plan and the SCD.

-	-	ED4	
		L	
		See.	
181	.8	BL 1	

JACOBS ENGINEERING GROUP INC. SECTION_ ALBUQUERQUE OPERATIONS

17.2.1 PAGE _____ OF ___ 3

LIMITATIONS 4.0

Should conflicts exist between the requirements presented herein and those defined in the UMTRA Project Site Management Manual, 4.1 the requirements of the Site Management Manual shall take precedence.

Prepared	By: A CALOWELL
Reviewed	By: Aulilian
Reviewed	By: Quality Assurance Manager
Approved	By: Aroject Manager

SECTION 17.2.1

11

TABLE 17.2.1 SITE TEAM MEMBERS AND SITE CUNCEPTUAL DESIGN RESPONSIBILITIES

Site Managers Team Leader; organize teams; control budgets and schedules; interact with project management, client, public, and agencies; direct formulation of Site Conceptual Design.

Task Managers Review Site Conceptual Design; advise on specialist matters; exercise delegated authority for manhour budgets.

Civil Engineering Specify pile layout, volumes, erosion protection requirements, surface-water control features, construction schedules, and quantity estimates. Compile relevant calculations.

Geotechnical Eng. Define site soil and rock; tailings geotechnical characteristics; site geology, geomorphology, and seismicity; and pile stability and deformation analyses. Compile relevant calculations.

Genydrology Characterize ground-water and interconnected surface-water conditions at the site. Define present and potential for contamination of ground water and surface water from pile remedial action. Assess possible design options to maintain or restore ground water quality.

Radiology Define areal and vertical extent of subpile and off-pile contamination and specific health and safety requirements. Calculate radon source term and thickness of radon barrier.

Cost Confirm quantities and prepare cost estimates for the site design and alternatives.

ALDUQUEDQUE ODEDATIONS MANUAL	SECTION	17.2.2
ALBUQUERQUE OPERATIONS WANDAL	DATE	7/20/87
	EFFECTIVE	04/29/88
SITE CONCEPTUAL DESIGN	SUPERSEDE	158.5.2 9/25/86
	PAGE 1	OF 2

PURPOSE 1.0

To describe the Radiological Site Conceptual Design Procedures.

2.0 SCOPE

This procedure describes the responsibilities for the data analysis, design calculations, reviews, and other activities required to complete a Site Conceptual Design.

3.0 PROCEDURE

- The Manager of Radiological Services will assign a health 3.1 physicist to formulate the Site Conceptual Design. Other Radiological Services staff will serve as checkers on any projects to which they have not been assigned.
- The designated health physicist will use the Materials 3.2 Management computer program to make a preliminary estimate of the volume of contaminated material and the Ra-226 source term. These estimates will be presented to the Manager of Radiological Services and the Site Manager for review. Appropriate changes on the preliminary estimates will be made as directed by the Manager of Radiological Services or his designated representative.
- Upon approval of the volume and source term calculations, the 3.3 designated health physicist will evaluate the radon diffusion coefficient data, the radon emanating fraction data, and the contaminated material porosity and density data. Modeled or default value of -15 bar long-term moisture content to be provided by Engineering. The data evaluation will provide a determination of the appropriate input parameters for the RAECOM Computer Code. All input parameters will be checked and approved by a checker on the radiological services staff.
- A sensitivity analysis will be performed by the designated 3.4 health physicist to evaluate the effect of cover long-term moisture, Ra-226 concentration, radon diffusion coefficient, and radon emanating fraction on the proposed cover thickness. The results of the sensitivity analysis will be reviewed by a checker on the radiological services staff.

SECTION _______ PAGE ____ OF _____

The cover thickness estimates generated by the RAECOM and the results of the sensitivity analysis will be approved by the Manager of Radiological Services or his designated 3.5 representative.

Prepared	By:	Marte fille for I Green
Reviewed	By:	Alack Alle
Reviewed	By:	Quality Assurance Manager
Approved	By:	Project Manager



ALDINOUEDOUN ODEDATIONS MANUAL	SECTION 17.3.1		
ALBUQUEFQUE OPERATIONS MANUAL	DATE 3/31/87		
PRODUCTION OF DOCIMENTS: TECHNICAL	EFFECTIVE 04/29/88		
DEPARTMENT PARTICIPATION	SUPERSEDES 8.10.3 8/14/80		
	PAGE 1 OF 3		

PURPOSE 1.0

The purpose of this section is to describe the procedures for the participation by Technical Departments in the production of UMTRA Project documents.

SCUPE 2.0

In order to document the remedial action plans and planning for UMTRA Project sites, the following documents are produced: Discussion Papers, Option Papers, and other similar documents; Environmental Impact Reports - EA and EIS; the Comparative Analysis of Disposal Site Alternatives Report - CADSAR; and the Remedial Action Plan - RAP. The Site Managers are responsible for producing the documents. The Technical Services (Engineering, Hydrological Services, Radiological Services, and Environmental Services) produce appropriate sections that are collated to compile the complete document. This procedure describes the participation by and input from the technical staff (Site Team members and Task Managers) to the documents. It does not cover the procedures adopted by Site Managers in compiling documents.

PROCEDURES: THE CADSAR 3.0

- 3.1 The Site Manager is responsible for compiling and writing the CADSAK.
- Site team members shall assist the Site Manager in writing the 3.2 CADSAR by providing information, data, advice, text of appropriate sections, figures or diagrams as requested by the Site Manager and directed by the Task Manager.
- Site team members may, as requested by the Site Manager, and as 3.3 directed by the Task Manager, write Appendices for the CADSAR. Such Appendices shall be written in a format similar to the format of the corresponding Appendix or Section of the RAP.
- Site team members shall input to and contribute information and 3.4 data to the CADSAR either by oral discussions or the transmission of white papers, figures, diagrams, tables, analyses, calculations, or aprices.
- Site team members and Task Normal Shall review the CADSAR, in accordance with the requirement Standard Operating Procedure 3.5 Number 9.2.6, prior to issue for technical acceptability, correctness, and reasonability.

ALEUQUERQUE OPERATIONS

2 05

3.6 The Site Manager is responsible for ascertaining and confirming that Task Manager approval of all information, etc., provided by Site Team members has been obtained.

- 3.7 It is not necessary to compile calculations or analyses to support data and designs incorporated into the CADSAR.
- 3.8 The provisions of this section shall apply also to the compilation of Discussion Papers, Option Papers, and other similar documents.

4.0 PROCEDURES: EA AND EIS

- 4.1 The Site Manager and Task Manager, Environmental Services (EV) are responsible for compiling the site EA or EIS.
- 4.2 Site team members shall contribute text, tables, figures, diagrams, and information to the documents as requested by the responsible Site and Task Manager.
- 4.3 Information provided to the EV Specialist compiling the EA or EIS shall be reviewed by the responsible Task Manager as soon as practical after compilation of the first draft. This may be before or after submission of the information to the EV Specialist.
- 4.4 The EA or EIS shall be reviewed and approved by the Task Managers in accordance with the applicable Standard Operating Procedures: Peer Review 8.5.2 or Request for Review 9.2.6.
- 4.5 Responses to comments on the EA or EIS shall be compiled by site team members as requested by the Site Manager and as directed by the Task Manager. Responses to comments shall be reviewed and approved by Task Managers if possible before submission to the Site Managers, and at any rate before issue to DOE.

5.0 PROCEDURES: RAP (Draft and Final)

- 5.1 The Site Manager is responsible for compiling and collating the RAP, and he shall do this in accordance with the provisions of the UMTRA Project Site Management Manual, UMTRA-DOE/AL 40005.0000.
- 5.2 Technical Departments are responsible for compiling sections of the RAP that relate to their specific disciplines. Technical contributions to the RAP are made by way of written text and associated figures and tables.
- 5.3 Site team members shall compile the written sections of the RAP text for which they are responsible at the direction of the Site Manager and in accordance with the concurrence and knowledge of the responsible Task Manager.

10-AL-0A-21(8/87)

JE ALBUQUERQUE OPERATIONS GROUP INC. SECTION 17.3.1

PAGE _____ 3 OF ___ 3

- 5.4 Before submitting sections of the RAP to the Site Manager for incorporation into the KAP, site team members shall have that text checked and approved by the responsible Task Manager. The Site Manager is responsible for confirming that Task Manager concurrence has been obtained.
- The RAP shall be checked and approved by the site team members 5.5 and the Task Managers before publication or issue of the RAP. This shall be done in accordance with Standard Operating Procedure 9.2.6.
- Updates or revisions to the RAP shall be made by the site team 5.6 members as requested by the Site Manager and as directed by the Task Manager.
- Responses to comments on the RAP shall be compiled by the site 5.7 team members as requested by the Site Manager and as directed by the Task Manager. Responses to comments shall be checked and approved by Task Managers if possible before submission to the Site Managers, and at any rate before issue to DUE.
- 6.0 LIMITATIONS
 - Should conflicts exist between the requirements presented herein 6.1 and those defined in the UNTRA Project Site Management Manual, the requirements of the Site Management Manual shall take precedence.

	TY P	
Prepared	By: A. ALOWELL	
Reviewed	By: Aaldwell	
Reviewed	By: Quality Assurance Manager	
Approved	By: Abroject Manager	

AL DI KOLIEDOLIE ODEDATIONIS MANI LAL	SECTION	17.3.2
ALBUQUERQUE OFERATIONS WANDAL	DATE	1/14/38
TECHNICAL APPRUACH DOCUMENT (TAD) CHECKLIST FOR REVIEW OF UMTRA RAP REQUIREMENTS	EFFECTIVE	04/29/88
	SUPERSEDES	
	PAGE 1	OF 2

PURPUSE 1.0

The purpose of this procedure is to provide guidance and direction to UMTRA personne: and minimum requirements for verification that draft and final Remedial Action Plans are reviewed and completed; containing all required material and data as prescribed by the UMTRA Project Technical Approach Document (TAD).

SCUPE 2.0

The scope of this procedure will encompass personnel responsibilities and requirements for performing reviews of UMTRA Project RAPs to minimum requirements listed on the attached checklist.

PRUCEDURE 3.0

- UMTRA Project site manager or task managers will assign 3.1 technical staff to review RAPs for evaluation of technical content, conclusion, and completeness.
- Technical staff personnel shall review all assigned UMTRA RAPs and verify that said documents contain, at a minimum, all items 3.2 identified in the Technical Approach Document (TAD) checklist for document review (Attachment 1).
- As each TAD checklist item is verified as being contained in, or absent from, the reviewed document, and that the item(s) is 3.3 determined either complete or lacking data, the reviewer will check the appropriate corresponding space for yes or no. "Yes" will indicate the item(s) is incorporated into the document and that the data and information contained within the document is complete and accurate. If, however, the document does not contain the required item(s) or the data or information is determined to be incomplete, the checklist item should be marked "No." All "No" responses should be accompanied by a brief statement in the "Remarks" space explaining the negative response. If the item(s) does not apply to the document under review, check "No" and write "N/A" under remarks.
- 3.4 Upon completion of review, completed checklists will be returned to the responsible task manager.
- The task manager shall review all returned TAD checklists. Task 3.5 managers shall take appropriate action as indicated below.
JE JACOBS ENGINEERING GROUP INC. SECTION 17.3.2 ALBUQUERQUE OPERATIONS

PAGE 2 OF 2

- 3.5.1 If the document is determined to be complete, with all minimum characteristics satisfactorily addressed, it will be approved by the task manager and forwarded for final review as directed in 3.6.
- 3.5.2 If comments on the checklist indicate that an item(s) is not incorporated, or data or information is determined to be incomplete, the document and a copy of the checklist will be returned to the appropriate technical personnel for rewrite or data incorporation and completion, as necessary.
- Upon task manager approval, the document and TAD checklist will 3.6 be routed for final review and approval and/or comment to the appropriate site manager, quality assurance, and project manager.
- After the document has completed the final review cycle, it 3.7 shall be returned to the task manager for incorporation of comments and final processing. Final document final processing shall be in accordance with AOM, Section 17.3.1.

Prepared By:	Chris Rfan Jether
Reviewed By:	Task Managert
Reviewed By:	, Quality Assurande Manager
Approved By	Project Manager

PAGE 1 OF 6

TECHNICAL APPROACH DOCUMENT CHECKLIST FOR RAP DOCUMENTS

SITE:	REVIEWER:
TASK MANAGER:	DATE:
CONCURRENCE:	

	ITEM		COVERAGE		
			YES	NO	REMARKS
1.	BAC	CKGROUND WATER QUALITY			
	Α.	Literature review			
	Β.	Well and spring inventory for two-mile radius			
	С.	Background munitor wells			
		1. Uppermost aquifer			
	******	i. Three or more monitor wells			
		2. Lower aquifer			
		i. Three or more monitor wells			- 11-12-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-
		3. Sampled at least twice			
		i. Six or more sample analyses			a filiana i si sena i na second i
******		4. Well logs			
	D.	Surface waters (pile side of stream or river)			
		1. Low flow			
		i. Upstream			
		ii. Intermediate			
		iii. Downstream			
		2. High flow			
		i. Upstream			
		ii. Intermediate			
		iii. Downstream			
JEG-	AL-0/	A-22 (10/88)	l		

Page 2 of 6

	ITEM		1.14	COVER	RAGE
			YES	NO	REMARKS
11.	PRESE	NCE AND MOVEMENT OF CONTAMINANT PLUMES IN IDWATER AND DISCHARGE TO SURFACE WATERS			
	A. C	ontaminants identified			
	1	. Source term			
	2	2. Grounowater			
*****		i. Vertical extent			
	Annalder verbende	ii. Lateral extent			
	a for a survey of the grant of	iii. Constituents above standards			
	3	3. Location of groundwater discharge to surface			
	B. 1	flow directions and aquifer properties			
		1. Vertical gradients			
	1	2. Lateral gradients			A PROPERTY OF A PROPERTY OF
Constitution of the		3. Average flow velocities			
		 Travel times to background concentration or surface water discharge 			
		5. Piezometric surface map	-	an a na tar taranta	
ΪΠ.	PRED	ICTION OF EFFE. OF REMEDIAL ACTION ON GROUNDWATER			
	Α,	Flux calculations for saturated zone			the contract of the second second second
****** Autom	Β.	Infiltration through pile			
	Ç,	Ectimate dispersion/attenuation of contaminants			
	D.	Geochemical controls on contaminant movement, geochemical modeling, precipitation/dissolu- tion, speciation adsorption, other water-rock interactions, leaching studies on site specific materials, mineralogical analyses influencing geochemical processes			
Peter Striverschaft	Ę,	Predicted future concentrations			
Second and the same	Street - Anticestances				with more than a second

JEG-AL-QA-22 (10/88)

.

a

Page 3 of 6

	ITEM	COVERAGE			
		YES	NO	REMARK	
1∀.	IMPACTS ON BENEFICIAL USE OF GROUNDWATER				
	A. Present value of affected resource				
	B. Potential for human exposure				
V.	CONTROL ALTERNATIVES FOR GROUNDWATER				
menuer entries	A. Aquifer restoration			an ger dat og ter binne sin de	
ana indoné na	1. Potential restoration alternatives				
	i. Technical feasibility				
	ii. Estimated cost				
	iii. Impacts on other users				
******	B. Alternate water supplies				
	C. Institutional controls				
VI.	RADIOLOGICAL SITE CHARACTERIZATION			en addates men anne (noa)	
	A. Appropriately spaces grid points				
	B. Borehole drilling/logging/sampling for deep contamination			nana an agas ng a	
	C. Gamma exposure rate measurements				
* *******	D. Soil sampling/delta measurements for shallow contamination				
	E. Soil sampling below tailings for thorium and heavy metals		****	e militate cannot a son qu	
	F. Building surveys for salvageable buildings				
	G. Volume calculations based on reasonable, conservative assumptions				
	H. Off pile Ra-226 source				
	I. Tailings pile Ra-226 source				



Page 4 of 6

	ITEM		COVERAGE		
			YES	NO	REMARK
/11.	RAD	DN BARRIER DESIGN			
	Α,	RAECOM			and the second se
	Β.	Geotechnical data - bulk density, porosity, long-term moisture, provided by site hydrologist			
	С.	Radon emanation - show insignificant dependence on moisture, Ra-226 contents			
	D.	Radon diffusion coefficients			
		 Measurements for all materials and justifica- tion 			
		2. Least squares fitting (LSDFIT) of results			
	Ε.	Radium content - layer average and values calcu- lated for each RAECOM layer			
	F.	Ambient radon - background site radon levels			
	G.	Sensitivity Analysis			
		1. Pile Ra-226 content			
		2. Radon emanating fraction			
		3. Cover diffusion coefficient			
		4. Cover moisture content			
		5. Tailings diffusion coefficient			
	terre minera	6. Worst case			
VIII	, SUF	RFACE WATER HYDROLOGY			
	Α.	Consider effect on pile of runoff from: direct precipitation; upland watershed; large streams			
	B.	Collect available data	-		
-	С.	Do flood studies			
	D.	Assess effect of geomorphology on pile			
-	E.	Review dam failure impact (if any)			

JEG-AL-QA-22 (10/88)

With Pitter

1

いの

•

CH.

Page 5 of 6

	ITEM	TEM			COVERAGE			
		n de la centre de la constance	YES	NÒ	REMARKS			
	F.	Determine PMP and PMF						
	G.	Provide diversion facilities to direct off pile flow						
martin of t	Н.	Provide erosion protection in and around pile for design events.						
Χ.	ERO	SION PROTECTION						
	Α.	Select appropriate design methodology						
	Β.	Identify source of rock						
	Ċ.	Size rock						
	D.	Size bedding						
	E.	Check adequacy of protection for ditches, aprons						
	F,	Confirm rock durability, check laboratory test list and conformance with NUREG requirements						
-	G.	Check compatibility of filter and erosion barrier						
Χ.	GEO	DTECHNICAL						
	Α,	Review previous work						
	Β.	Characterize region and site geology						
******	С.	Define site geomorphology; gullies, mass move- ment, slope erosion, techtonism, base level/ change						
	D.	Define subsurface conditions: archived data, field boreholes, test pits, laboratory test			teri metan dapat dang salah ter			
	Ε.	Borrow pit definition - material characteristics						
	F.	Tailings characteristics						
	G.	Construction material characteristics						
and success	Н.	Field seismicity: surface and bedrock acceler- ation; seismic magnitude and intensity; fault locations						

JEG-AL-QA-22 (10/88)

......

9

.

Page 6 of 6

	ITEM			COVERAGE		
			YES	NO	REMARKS	
	Ι.	Slope stability: static and seismic; factors of safety adequacy				
	Э.	Pile settlement: settlement defined, potential effect on pile integrity considered				
	К.	Liquefaction potential: adequate soil strength soil moisture adequacy, compaction of tailings				
XI.	NON	-RADIOLOGICAL CONTAMINANTS				
	Α.	Review of existing soil and tailings data		1		
	₿.	Inventory of potentially hazardous materials				
	С.	Data gaps (additional sampling)				
	D.	Correlation between non-radiological consti- tuents and Ra-226				
to prove the second		and the second	and the second sec			

JEG-AL-QA-22 (10/88)

ACOBS ENGINEERING GROUP INC.

A LEUQUERQUE OPERATIONS

AL PLIQUEROUE OPERATIONS MANUAL	SECTION	17.4.1
ALBOQUERQUE OFERATIONS WANDAL	DATE	6/12/87
	EFFECTIVE	04/29/88
LOCATION IDENTIFICATION FOR TEST BORINGS, TEST PITS. AND MONITORING LOCATIONS	SUPERSEDE	IS 8.2.13 7/4/86
TEST FITE, AND HONITONING DOOLSTOND	PAGE 1	OF 2

1.0 PURPOSE

This operating procedure establishes a uniform test boring, test pit, and monitoring location identification system for use on UMTRA site investigations.

2.0 POLICY

All test borings, test pits, monitoring locations, and observation wells where physical, chemical, biological, or radiological measurements are made will be designated using the standard identification system described herein. This procedure includes all existing locations that are or will be used by the TAC for the completion of the site characterization tasks. Note, however, that excluded from this standard procedure are the thousands of surface locations scanned or sampled for the purpose of site characterization requiring no specific identifier. These surface locations (where the elevation (Z) is implied to be the topographic surface) will be simply referred to by a north coordinate (Y) and an east coordinate (X).

3.0 PROCEDURE

Each location defined above shall be identified by a Site Code and a Location ID. The site code will consist of the standard 3-character designation as listed in Section 8.3.5 of the AOM plus an additional two-digit number further describing a specific aspect of the site defined categorically by the site type. This two-digit number is assigned by the Data Administrator after concurrence with the Site Manager and maintained in the General Site Information (GSI) Master File.

The Location ID will consist of a 3-digit designation with a range per site as follows:

001 through 999

This three-digit number is assigned and maintained in a log by the Data Administrator.

Thus, the identification will take the form,

SSSNN-NNN



ALBUQUERQUE OPERATIONS SECTION 17.4.1

PAGE 2 OF 2

where:

SSSNN is the site code NNN is the Location ID per site.

EXAMPLES

CAN01-001 SLC02-345 Canonsburg Processing Site - Borehole Number 001 Salt Lake City (Clive) Disposal Site - Test Pit Number 345

No designation in this procedure provides for distinguishing the log type, e.g., radiological, geological, hydrological, etc. As a result, any location may be logged for multiple data types without restriction noted in the identification. Other characteristics associated with each defined location will have independent descriptions. Included are items such as: north coordinate, east coordinate, elevation, log type, log date, previous identification, etc. All of these descriptions may be used as additional identification for test borings, test pits, etc., as necessary.

4.0 RESPONSIBILITY

Each Site Manager and responsible technical personnel shall incorporate this identification system into all field investigation activities and permit no data gathering efforts that do not include the standard identification system. The Data Administrator will be responsible for maintaining the sequential numbering system and dispensing these numbers as necessary. Any existing location used in site characterization will be redesignated by technical personnel and approved by the Data Administrator to conform to the standard identification system described above.

Prepared	By:	Guid fan Bel in & Alel.
Reviewed	By:	Juida laxie Mask Manager
Reviewed	By:	Quality Assurance Manager
Approved	By:	Project Manager

JEG-AL-DA-21(8/87)

JE JACOBS ENGINEERING GROUP INC.

UMTRA PROJECT TAC HYDROLOGICAL STANDARD OPERATING PROCEDURES

- 16.0 Hydrological Standard Operating Procedures
 - 16.1 Field Operations

16.1.1	Monitor Well Installation Well Development
16.1.3	Slug Testing
16.1.4	Packer Testing Aquifer Pump Testing
16.1.6	Soil Water Sampler Installation and Use
16.1.7	Batch and Column Leach Testing

16.2 Water Sampling

16.2.1	Water	Sampling,	Preserving,	, Shipping	, and Testing
16.2.2	Water	Sampling	for Tritium	Analysis	
16.2.3	Evalua	ation of C	hemical Anal	lysis of W	later Samples

16.3 Data Control

16.3.1 Completion of Well and Spring Inventory

dan tinued

JE JACOBS ENGINEERING GROUP INC.

UMTRA PROJECT TAC TECHNICAL STANDARD OPERATING PROCEDURES

- 17.0 Technical Standard Operating Procedures Office
 - 17.1 Analysis and Calculations

17.1.1 Analysis of Data 17.1.2 Compilation of Design Calculations

17.2 Design

17.2.1	Conceptual	Design		
17.2.2	Radon Barri	er Site C	onceptual	Design

17.3 Documents

17.3.1	Production	of Documents:	Technical	Division
	Participat	on		

- 17.3.2 Technical Approach Document (TAD) Checklist for Review of UMTRA RAP Documents
- 17.4 Data Control
 - 17.4.1 Location ID for Test Borings, Test Pits, and Monitoring Locations

JACOBS ENGINEERING GROUP INC.

ALBUQUERQUE OPERATIONS

AL REPOLIE OPERATIONS MANUAL	SECTION 16.1.1
ALBOQUENQUE OFERATIONS MANUAL	DATE 10/02/90
	EFFECTIVE 11/15/90
MONITOR WELL INSTALLATION	SUPERSEDES 02/25/87
	PAGE 1 OF 8

1.0 PURPOSE

To provide quality control for monitor well installation that will enable successful completion of field drilling investigation(s) oriented toward obtaining representative geohydrological quality information, and to provide regulatory compliance for monitor well construction, modification, repair and abandonment activities.

2.0 SCOPE

This procedure is applicable to members of the TAC and its contractors and subcontractors with responsibilities and duties for the installation of monitor well(s), well development, well testing, and collection of water samples for water quality determinations.

3.0 PROCEDURES

- 3.1 All field measurements and comments shall be recorded on Attachment 5.1 or 5.2 pursient to TAC data reporting formats and protocols. See 4.0.
- 3.2 All lines on the forms shall be completed. The letter designation "NA" (Not Applicable) or "NK" (Not Known) shall be used in all blank spaces.
- 3.3 If some steps or procedures were not performed as described, the reason must be stated on Attachment 5.1 or 5.2 or submitted as an attachment to the data sheet.
- 3.4 When granular backfill is used as part of a well installation, at least one-half to one pint of sample shall be taken from each shipment of granular material, stored with the soil samples, and recorded on the sketch of each installation for which that material was used.
- 3.5 The contractor shall submit the spelifications of the proposed sand pack material to the TAC Field Technical Representative (FTR) for approval prior to use. Describe each sample in terms of lithology, grain size distribution, and source. Record both the company from whom the material was purchased and the pit or quarry of origin. This material shall be clean, inert, and siliceous. Use typically well-sorted (poorly graded) sand such as No. 8-12 or similar material with less than ten percent (10%) passing through the design slot size for medium to course grain aquifer materials and No. 10-20

JACOBS ENGINEERING GROUP INC. SECTION 16.1.1 ALEUQUERQUE OPERATIONS PAGE 2 OF 8

with less than ten percent (10%) passing through the design slot size for fine to medium grain aquifer materials.

3.6 Sketch the well installation on Attachment 5.1 and show, by depth from surface grade, the bottom of the boring, sump, screen location, coupling design and location, granular backfill, seals, grout, cavein centralizers, and height of riser above ground surface. Record the actual composition of the grout, seals, and granular backfill on each Attachment 5.1. Also include the screen slot size (in inches), slot configuration, total open area per foot of screen, and screen manufacturer.

Note: Any use of PVC solvents, glues, soap, or cleaners is prohibited below grade unless otherwise stated in the specifications. Where such material is used, it shall be recorded and described on Attachment 5.1 and shall include manufacturer and type (specification).

- 3.7 Use, whenever applicable, PVC centralizers to assure the uniform and complete annular filling by granular backfill, seal, and grout materials. Fasten centralizers to the well casing with mechanical fasteners. The centralizers shall be radially spaced about the casing at 120° or 90° intervals. On Attachment 5.2, provide a description and location(s) of the fastening device and centralizer.
- 3.8 Attachment 5.3 and 5.4 depict typical well construction. However, specific contract requirements may alter some of the components and/or values shown. Furthermore, when installed as a piezometer or all point (for water levels only), the well configuration may significantly vary from these figures. After development is complete, indicate the static water level in the well on the well construction diagram.
- 3.9 Well sketches shall include the protective casing detail.
- 3.10 Install protective casing around all monitor wells. The TAC shall determine exceptions on a case-by-case basis. Minimum elements in the protection design include:
 - 3.10.1 A five-foot minimum length of black iron pipe or galvanized pipe extending about 1.5 to three feet above the ground surface and set in cement grout. See Attachments 5.3, 5.4, and 5.5.
 - 3.10.2 The pipe diameter shall be eight inches (8" i.d.) for fourinch wells and six inches (6" i.d.) for two-inch wells (depending on approved borehole size). A 0.5-inch drain hole near ground level is permitted.
 - 3.10.3 Install a protective steel cap (as in Attachment 5.3 and 5.4) and secure the cap to the top of each protective casing by the use of a steel pin which penetrates the

casing and cap. Paint the location identification on the inside and outside of the cover with enamel type paint. The Site Manager shall designate location identification which shall conform with TAC format.

- 3.10.4 The well cover must keep precipitation out of the protective casing and shall be secured to the casing by means of padlocks.
- 3.10.5 It is desirable that all padlocks at a given site be opened by the same key. The numbers on the locks shall be noted and then rubbed off the back of the lock.
- 3.10.6 (Optional) Paint the protective casing with aluminum colored paint.
- 3.10.7 As shown in Attachment 5.5, secure those wells which are not protected by the steel surface casing with an enclosure of three (3) fenced wood or steel posts. Cement each post in place. NOTE: The installation of the posts is recommended in addition to the protective casing in areas where vehicle traffic might pose a hazard.
 - The wood posts should be 4" x 4" x -6', the steel posts should be 3" diameter (o.d.) x -6', and the T-bar driven steel posts should be 6' long.
 - Place each of the posts radially and about four feet (4") from the well, placed two feet (2') below ground surface, and have four feet (4') minimally above ground surface.
 - If high vegetation exists around the well site, flag the posts with brightly-colored plastic ribbon-like material or its equivalent.
- 3.10.8 Optional. Use three-strand barbed wire strung on four (4) posts as shown in Attachment 5.5, as livestock guards in grazing areas. For post dimensions see 3.10.7.
- 3.11 Emplace a filter pack (granular backfill; sand pack) in the annulus adjacent to the well screen in all monitor wells as follows:
 - 3.11.1 Fill the annulus between the well screen and borehole wall with approved silica sand.

When neither of the two proposed generic filter sand/screen-slot sizes are used, it will be required that representative aquifer material be collected and sieved to determine grain size distribution.

JACOBS ENGINEERING GROUP INC. SECTION ALBUQUERQUE OPERATIONS

3.11.2 The average grain-size of the filter sand shall be based on expected grain-size distributions in the screened formation and on the size of predesigned well-screened openings. Ideally, the filter pack will pass only the smallest of the grain size (40%) found within the formation. The sand shall have a gradation which will allow not more than ten percent (10%) of pack material to pass through the screen slots. Refer to Figures 16.1 1-5, 16.1 1-6, 16.1 1-7, for selecting filter pack material size and well screen slot size.

16.1.1

OF _

- 3.11.3 For wells deeper than 75 feet, or as determined by the FTR, emplace the sand pack with a tremie pipe. Gravity feed the sand through the tremie pipe into the annulus throughout the entire screened interval and a minimum of three feet over the top of the screen.
- 3.11.4 As a last resort, clean water should only be added when no other practicable method exists for proper filter pack emplacement. In such a case it is necessary to pump sufficient sand slurry through the tremie pipe to cover the screen after the sand pack has densified and settled. As a general rule-of-thumb, apply at least a seven-foot (7') thick layer of sand slurry over the top of a 15 foot-long screen. Increase the sand-pack thickness by two feet (2') for each additional five feet (5') of screen. Thus, the top of a 30-foot long screen would be 13 feet deeper than the top of the sand-pack after emplacement.
- 3.11.5 Using the tremie pipe, ascertain the depth of the top of the sand and verify the thickness of the sand pack. Additional sand shall be added to bring the top of the sand pack to the proper elevation, if necessary.
- 3.11.6 Under no circumstances shall the sand pack extend into any aquifer other than the one to be monitored. In most cases, the well design can be modified to allow for a sufficient sand pack without threat of cross-flow between producing zones through the sand pack.
- 3.11.7 The Site Hydrologist and FTR are the only individuals authorized to modify an existing well design.

The filter pack shall consist of clean silica sand and ensure continuous flow capability from the natural formation to the well bore.

3.12 Emplace a bentonite seal between all contacts of sand pack and grout to prevent infiltration of cement into the well. In special circumstances, where an open borehole has been drilled to a depth below that at which the screen will be set and if grout is used to seal off a lower aguifer or as backfill up to the proper level,

JACOBS ENGINEERING GROUP INC. SECTION 16.1.1 ALBUQUERQUE OPERATIONS

PAGE 5 OF 8

emplace a bentonite seal above the grout before casing, screen, and sand pack are introduced. Allow the grout to set up prior to placing the seal. Place any bentonite seal in the monitor well as follows:

- Fill the annulus between water-tight well casing and 3.12.1 borehole (and in the interval between the sand pack and the grout seal) with a bentonite seal at least five-feet thick (vert' 11)y). The annulus must be filled with a bentonite seal to a level above the highest seasonal groundwater fluctuation.
- 3.12.2 Emplace the seal material as bentonite pellets or graded bentonite chip. The bentonite shall have a minimum purity of 90% montgorillonite clay and a minimum dry bulk density of 75 lb/ft³ for 1/2-inch pellets, as provided by American Colloid, or its equal and 50.0 lb/ft³ for 3/4" graded chip. The pellets and graded chip should be screened prior to placement in order to separate the fines from the unbroken - intact seal material. The fines are discarded. Place a cap over the top of the casing and pour a five (5) gallon bucket of bentonite directly down the annulus. Note: Care must be taken to avoid introducing pellets into the well bore. Pour the pellets from different points around the casing to ensure even application. Use a tremie pipe to redistribute and level out the top of the seal.
- 3.12.3 For wells deeper than 75 feet, pour the bentonite as pellets, or if necessary, introduce the bentonite as a slurry. The TAC FTR shall determine the method of emplacement after evaluating the condition of the well and borehole wall.
 - 1) If there are no centralizers in the upper portions of the casing, manipulate the casing to prevent pellets from hanging up in the narrow annulus and to allow them to settle as rapidly as possible to the bottom.
- If a slurry of bentonite is used as an annular seal, 3.12.4 prepare the slurry by mixing powdered bentonite with potable water. The slurry must be of sufficiently high specific gravity and viscosity to prevent its displacement by the grout that will be emplaced above. As a precaution, regardless of depth and depending on fluid visocity, emplace a few handfuls of bentonite pellets in to solidify the surface of the bentonite slurry.
- 3.12.5 Before pumping the seal, make sure that the sand pack has ceased settling by measuring the depth of the top of the sand with the tremie pipe. The sand pack must provide an adequate cover over the screen as discussed in 3.8.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.1 ALBUQUERQUE OPERATIONS

PAGE 6 OF

- 3.12.6 Visually check the condition of the slurry by pumping into a bucket or onto the ground. Retract the tremie pipe three feet and commence pumping.
- 3.12.7 In all situations, emplace a bentonite seal a minimum thickness of five feet. The top of the seal shall be tagged by tremie pipe to verify that the proper thickness of seal has been placed in the annulus.
- 3.12.8 Repeat application and verification as necessary until the specified quantity of bentonite has been placed in the well annulus.
- 3.13 Use only Type I or Type II cement without accelerator additives. Allow 20 minutes for bentonite slurry or bentonite pellets to set before grouting commences. Place grout in the monitor well as follows:
 - 3.13.1 Fill the annulus between the well casing and borehole wall with grout.
 - 3.13.2 Place the grout with a grout tremie-pipe located just over the top of the seal.
 - 3.13.3 The tremie pipe should normally consist of 1.25-inch PVC or steel pipe. One-inch rolled thin-walled poly pipe has been used with success in some cases.
 - 3.13.4 Pump the grout through this pipe to the bottom of the open annulus until undiluted grout flows from the annulus at the ground surface.
 - 3.13.5 The grout shall consist of a neat cement mix with four pounds of commercial bentonite and approximately 7.5 gallons of water added per 94-pound bag of cement. Use only grout mixed with TAC-approved water.
 - 3.13.6 Any drill casing or hollow-stem auger shall be left in the hole during grouting to the extent practical.
 - 3.13.7 During grouting, the drill casing or hollow-stem auger may be removed as the grout level rises above the contact.
 - 3.13.8 While the grout is still green (per 3.13.5 above), add additional grout to compensate for the removed casing or auger and tremie pipe and to ensure that the top of the grout is at or above ground surface. Construct a grout apron at the base of the protective casing to facilitate surface water runoff.
 - 3.13.9 Install the protective casing.

JE

JACOBS ENGINEERING GROUP INC. SECTION 16.1.1

3.13.10 After the grout has set for 48 hours or more, fill any depression in the grout due to settlement with a grout similar to that described in 3.13.5.

3.14 Clean up equipment and work area.

4.0 MONITOR WELL REGULATORY COMPLIANCE

- 4.1 It shall be the responsibility of the UMTRA site hydrologist, site manager, or other project personnel designated by the site manager proposing to construct, modify, repair, or abandon an UMTRA monitor well to notify <u>in writing</u> the regulatory compliance staff, within Environmental Services, of the proposed action. This shall be done <u>at the same time</u> as the scope of work is developed for the proposed action.
 - 4.1.1 The notification to Environmental Services shall be recorded on Part I of the Regulatory Compliance Data Sheet, Attachment 5.8.
- 4.2 It shall be the responsibility of the regulatory compliance staff, upon being notified that a proposed monitor well action is pending, to initiate the regulatory compliance and clearance process, and update the central Monitor Well Inventory database.
 - 4.2.1 With this information, the regulatory compliance staff will open a record or modify existing records in the Monitor Well Inventory database (available via the DART) providing a tracking system of the regulatory, maintenance, and access status of all UMTRA monitor wells.
 - 4.2.2 Environmental Services will obtain necessary regulatory permits or clearances from the appropriate state, tribal, or Federal permitting agencies and provide copies to the FTR before drilling can commence.
- 4.3 In the case of new monitor wells or those wells requiring repairs, the site hydrologist/FTR must provide a status report to Environmental Services on the proposed activity within 30 days of the initial notification.
 - 4.3.1 For those states where licensed drillers are required to file completion reports, the site hydrologist/FTR must provide verification of filing, including a copy of the completion report, to the regulatory compliance specialist.
 - 4.3.2 For those states where the well owner files compliance documentation, the site hydrologist/FTR must provide copies of the completion report to the regulatory compliance staff within one week of the completion of drilling.



J COBS ENGINEERING GROUP INC. SECTION 16.1.1

- 5.0 HTTACHMEN(15): Supporting Document(s)
 - 5.1 Figure 15.1.1-1; Well Completion Record [JEG-AL-ENG-3(6/87)].
 - 5.2 Figure 16.1.1-2; Borehole/Well Construction Log [JEG-AL-ENG-1(6/87)].
 - 5.3 Figure 16.1.1-3a; Typical Construction of Alluvial Well.
 - 5.4 Figure 16.1.1-3b; Typical Construction of Bedrock Well.
 - 5.5 Figure 16.1.1-4a; Post Placement Around Well.
 - 5.6 Figure 16.1.1-4b; Post Placement Around Well
 - 5.7 Figure 16.1.1-5; Sieve Retention Graph
 - 5.8 Figure 16.1.1-6; Sieve Retention Graph
 - 5.9 Figure 16.1.1-7; Correlation Chart of Screen Openings and Sieve Sizes
 - 5.10 Figure 16.1.1-8; Regulatory Compliance Data Sheet

weir & Su. Prepared By: Representative Reviewed Ey: Manager Reviewed By: Quality Control Manager Approved By: Project Manager



		IF A	COBS	ENGINEE	RING PROUP INC.			
	В	OREHOLE	E/WI	ELL CO	NSTRUCTION	LOG		
SITE ID	-	LOCATION	N ID:		FIELD REP:			-
APPRO)	SITE COO	RDINATES	(FT.)	: N		E		
GROUND	ELEVATION	FT. MSL):		COMPLET	ON DATE	1	
	BOREHOL	E SUMMAR	Y		CONST	RUCTION	TIME LOC	3
DOULED					1.5500150	STA	RT	END
DRILLER	and the second s	A month and insult from the statement of the		and some of the standard	ACTIVITY	DATE	TIME	TIME
RIG TYP	E:	END .	1		DRILLING			-
BIT TYP	E DIA.	DEPTH	F	YPE	UNICEING			
	(11.7	<u> </u>						
					CASING			
	CASING	SUMMARY			FILTER PACK			
CASING	DESCRIP	TION	DIA.	DEPTH				
111 6 -		and repairing and the second		(11.)	SEAL			
					BACKEILL			
					DAGRICE			
					DEVELOPMENT			
		an and a fair and a second			-			
		and the state of the						
					OTHER			1.
P-Protect	ive S-Screen	B-Blank O-	Open	N-None	1			1
Depth from	m Top of Casin	9				1 DEVEL	OPHENT	Accession
	WELL CON	STRUCTIO	N	TEND	WEL	L DEVEL	OPMENI	
CODE*	DESCR	RIPTION		DEPTH (ft.)				
]			
					-			
	an ya ku su				-			
	and the statement of the statement of the statement of				COMMENTS	-		
								CONTRACT OF A DESCRIPTION
and the second s		Carl E	- Fito	Park	Contraction and the second			

JEG-AL-ENG-1 (6/87)





的你你的。" 第二章 FIGURE 16.1.1-4



PROFILE VIEW

÷

Figure a

PROFILE VIEW

Figure b



SLOT OPENING AND GRAIN SIZE, IN THOUSANDTHS OF AN INCH

FIGURE 16.1.1-5





SLOT OPENING AND GRAIN SIZE. IN THOUSANDTHS OF AN INCH

Castonia				Tyler		UA	S. Standard	
Material Grain-size Range	Johnson Slot No.	Gauze No.	Sieve No.	Size of C Inches)penings mm	Sieve No.	Size of Openings Inches	
clay & silt		1111	400 325 270 250 200	0.0015 0.0017 0.0021 0.0024 0.0029	0.038 0.043 0.053 0.061 0.074	400 325 270 230 200	0.0015 0.0017 0.0021 0.0024 0.0029	
fine sand		90 80 70 60	170 150 115 100 80 65 60	0.0035 0.0041 0.0049 0.0058 0.0069 0.0082 0.0097	0.088 0.104 0.124 0.147 0.175 0.208 0.246	170 140 120 100 80 70 60	0.0035 0.0041 0.0049 0.0059 0.0070 0.0083 0.0098	
medium sand	12 14 16 18 20 23	<u>50</u> 40	48 42 35 	0.0116 0.0138 0.0164 0.0180 0.0195 0.0232	0.295 0.351 0.417 0.457 0.495 0.589	50 45 40 	0.0117 0.0138 0.0165 (//m) 0.0180 0.0197 0.0232	
coarse sand	25 28 31 33 35 39	30 20	24 20 16	0.0250 0.0276 0.0310 0.0328 0.035 0.039	0.635 0.701 0.788 0.833 0.889 0.991	25 20 18	0.0250 0.0280 0.0310 (½) 0.0331 0.0350 0.0394	(a)
very coarse sand	47 56 62 66 79		14 12 10 9	0.046 0.055 0.062 0.065 0.078	1.168 1.397 1.590 1.651 1.981	16 14 12 10	0.0469 0.0555 0.062 (%) 0.0661 0.0787	(b)
very fine gravel	93 94 111 125 132		8 -7 -6	0.093 0.094 0.110 0.125 0.131	2.362 2.390 2.794 3.180 3.327	8 -7 -6	0.0931 0.094 (%) 0.111 0.125 (%) 0.132	
fine gravel	157 187 223 250 263 312 375		5 4 3½ 	0.156 0.185 0.221 0.250 0.263 0.312 0.371	3.962 4.699 5.613 6.350 6.680 7.925 9.423	5 4 3 1/2 3 1/2 - 1/2 3/2	0.1\$7 0.187 (Y _m) 0.223 0.250 (Y ₄) 0.263 0.312 (Y _m) 0.375 (Y ₄)	
	438 500		0.441 0.525	0.441 0.525	11.20 13.33	У њ Уз	0.438 (½) 0.500 (½)]

(a)

Correlation Chart of Screen Openings and Sieve Sizes

UMTRA specifications suggest 10-20 silica sand with 0.030 slotted screen for fine to medium a) grain aquifers.

 b) UMTRA specifications suggest 8-12 silica sand with 0.050 slotted screen for medium to course grain aquifers.



REGULATORY COMPLIANCE DATA SHEET

PART I (To Completed by Site Hydrologist/FTR)

Submit to Regulatory Compliance Specialist, Environmental Services, concurrent with initiation of the scope of work. Locate wells by township, range, and section (See back of sheet) and attach map _____.

Site Hydrologist/FTR

Site ID Code: _____ Location ID Code: _____

Action to be taken: [] Drill Wells [] Repair Wells [] Abandon Wells

No. of wells: Monitor Wells ____ Bore Holes ____ Test Pits _____

Driller (name/address):_____

State License #:

Land Owner (name/address):

Proposed Construction Start Date:

Proposed Construction Completion Date:_____

Drill Rig/Construction Type:

PART II (To be completed by Regulatory Compliance Staff)

Permit/Clearance Date:_____

Access Agreement #: ____ Date:____

Open record in Monitor Well Inventory [_] Yes [_] No Date:___

PART III (To be completed by Regulatory Compliance Staff)
Actual Construction Start Date:______ Actual Completion Date:______
Well Completion Report filed: Driller [_] Jacobs [_] Date:______
Records in files: [_] Yes [_] No
Update Monitor Well Inventory: Date

1.	N/SE/W	FT. & E/W SLBM/USM.	DEPTH	COR.	or 1/4 of SEC	TN	_ N/S
2.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC	TN	_ N/S
3.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC	IN.	N/S
4.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC.	TN	_ N/S
5.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC.	TN	_ N/S
6.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC.	TN	N/S
7.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM _	COR.	or 1/4 of SEC.	TN	N/S
8.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM _	COR.	or 1/4 of SEC.	TN	N/S
9.	N/SE7W	FT. & E/W SLBM/USM.	FT. FRM	COR. C	DIAMETER	T	N/S
10.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR. C	DIAMETER	T	_ N/S
11.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC.	T -	N/S
12.	N/SE/W	FT. & E/W SLBM/USM.	FT. FRM -	COR.	or 1/4 of SEC.	T	N/S
13	. N/SE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC. DIAMETER	TN	N/S
14	N/SE/S	FT. & E/W SLBM/USM.	FT. FRM -	COR. FT.	or 1/4 of SEC. DIAMETER	T -	N/S
15	. N/SRE/W	FT. & E/W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC. DIAMETER	T -	N/S
16	. N/SRE/V	FT. & E/W SLBM/USM.	FT. FRM DEPTH	COR.	or 1/4 of SEC. DIAMETER	T -	N/S
17	. N/SRE/	FT. & E/W SLBM USM.	FT. FRM	COR.	or 1/4 of SEC. DIAMETER	TN	N/S
18	. N/SE/	FT. & E/W W SLBM/USM.	FT. FRM	COR.	or 1/4 of SEC.	T -	N/S
19	. N/SE7	FT. & E/W SLBM/USM.	FT. FRM DEPTH	COR.	or 1/4 of SEC	T	N/S
20	. N/SE7	FT. & E/W SLBM/USM.	FT. FRM	COR	or 1/4 of SEC	TN.	N/S

FIGURE 16.1.1-8

JE JACOBS ENGINEERING GROUP INC.

AL BUOLIEROUE OPERATIONS MANUAL	SECTION	16.1.2	
ALBOQUENQUE OFENATIONS IVANUAL	DATE	5/27/87	
	EFFECTIVE	08/18/88	
WELL DEVELOPMENT	SUPERSEDES	8.2.8 6/6/86	
	PAGE 1 (DF 3	

1.0 PURPOSE

To define the procedures to be followed while developing a well in order to obtain water samples for future water-quality testing.

2.0 SCOPE

The following procedures shall be implemented after the installation of an observation monitoring well.

As attachments to this section are the applicable TAC forms:

Figure 1, Borehole/Well Construction Log JEG-AL-ENG-1 (3/84).

All field measurements and comments shall be recorded on the appropriate TAC forms as per TAC data reporting formats and protocols. All lines on the forms shall be completed. The letter designation "NA" for not applicable or "NK" for not known shall be used in all blank spaces. If some steps or procedures were not performed as described, the reason must be stated is is practicable on the appropriate form or submittee as an attachment thereto.

3.0 PROCEDURES

- 3.1 The development shall be performed as soon as practical after well installation, but no sooner than 48 hours after grouting is completed.
- 3.2 Development shall be accomplished by airlift pumping and/or bottom discharge bailer until the well water is clear and sediment within the well is removed to the fullest extent practical.
 - a) For those wells where the boring was made without the use of drilling fluid (mud, and/or water), remove approximately five (5) times the standing water volume in the well (well screen and casing plus saturated cnnulus). Should recharge be so slow that five volumes could not be removed in one day, or the water remain discolored or contain greater than one percent particulates after this five-volume removal, the TAC Office shall direct an alternate procedure based on the subjective judgment of the lead geotechnical engineer or hydrologist.

JE ALBUQUERQUE OPERATIONS GROUP INC. SECTION _______ 16.1.2

PAGS 2 OF 3

- b) For those wells where the boring was made or enlarged with the use of drilling fluid (mud, and/or water), remove approximately five times the measured amount of total fluids lost while drilling plus five times the standing water volume as above (in paragraph a.). The same procedure also applies here as above for cases of slow recharge, discolored, or particulate-laden water.
- 3.3 No dispersing agents, acids, or disinfectants shall be used to enhance development of the well.
- No water shall be added to the well to assist development 3.4 without prior TAC/TR approval. If a well cannot be cleaned of mud using airlift to produce formation water because the aquifer yields insufficient water (e.g., one gallon per minute or less), small amounts of potable water may be injected to clean up this poorly yielding well. This may be done by dumping in buckets of water or by pumping it through the air line intermittently. When most of the bentonite is out, continue development with formation water only. It is essential that at least five times the amount of water injected must be produced back from the well in order to assure that all injected water is removed from the formation. When most of the bentonite is cut, continue development with formation water only, as described in Section 3.2.b.
- 3.5 During development, the effort shall be made to remove the standing water from points near the bottom of the well screen as well as from the top of the water column. Moreover, the airlift outlet should be placed at several depths along the screened interval to ensure complete development through the sand pack. Since casing consists primarily of PVC material, the control of air-blast within the screen section is essential to avoid rupturing the screen.

To evaluate initial water quality, the FTR shall conduct pH and specific conductivity tests during development. The detection of abnormal readings should be relayed to the TAC/TR prior to completion of development.

- 3.6 If problems are encountered during development, the TAC project team shall be notified by telephone within one working day after problem detection.
- The following data shall be recorded on the Borehole/Well 3.7 Construction Log or an attachment thereof as part of development.
 - a) Well designation (location ID).
 - b) Date(s) of well installation.

JEG-AL-OA-21(8/87)

JACOBS ENGINEERING GROUP INC. ALBUQUERQUE OPERATIONS

PAGE 3 OF 3

- c) Date(s) and time of well development.
- Static water level before and after development. d)
- e) Quantity of water removed and time of removal.
- f) Depth of open hole inside well before and after development.
- g) Physical character of removed water, to include changes during development in clarity, color, particulates, and odor.
- h) Physical character of removed sediments, to include lithology and grain size.
- i) Type and size/capacity of pump and/or bailer used.
- j) Description of surge techniques, if used.

Prepared By: Reviewed By: Manage ACK Assurance Manager Reviewed By: Jua Approved By: Project Manager

	В	JE -	COBS E		NG GROUP INC.	LOG		
SITE ID:		LOCATION	1 ID:		FIELD REP:			
APPROX.	SITE COO	RDINATES	(FT.)	: N		E		
GROUND I	ELEVATION	(FT. MSL):		COMPLETI	ON DATE	-	
	BOREHOLE	E SUMMAR	Y		CONSTR	RUCTION	TIME LOG	
	an baar baar ay baar ay baar ah ay ah		0A 04.512 005.511		and the second	ST	ART	
DRILLER.		and a set of the second se			ACTIVITY -			TIME
RIG TYPE	-					DATE	TIME	
BIT TYPE	HOLE DIA. (in.)	END DEPTH (ft.)	FL	UID YPE	DRILLING			
					0.10110			
					CASING			
AND AND A DESCRIPTION OF THE OWNER OF THE OWNE	CASING	SUMMARY	ada ar ana menanin		EU TER PACK			
CASING	DECODI	DTION	DIA.	END	The fen From			
TYPE	DESCHI		(in.)	(ft.)	SEAL			
					BACKFILL			
					DEVELOPMENT			
					ATUED			
	a, managatan orang ataunan ara bit a		-		OTHER			
* P-Protecti	ve S-Screen	B-Blank O	-Open	N-None			11.1.1.2	
Depth from	Top of Casi	ng						
	WELL CO	NSTRUCTIO	N	TEND	WE	LL DEVE	LOPMENT	
TYPE CODE*	DESC	RIPTION		DEPTH				
				- a cond individuo	1			
					_			
					COMMENTS:			
	acktill C	- Seal	F - Fib.	Pr Pack				
* 5 * 8	auxim o	- dear						

FIGURE 1

-

JE JACOBS ENGINEERING GROUP INC.

THE PROVIDE OPERATIONS MANUAL	SECTION	16.1.3
ALBUQUERQUE OPERATIONS MANUAL	DATE	5/27/87
	EFFECTIVE	08/18/88
ALVA TESTING	SUPERSED	ES 8.2.10 1/30/86
SLUG TESTING	PAGE 1	OF 2

1.0 PURPOSE

To define the methodology to be followed in conducting slug tests of hydrogeologic units. A slug test is used to determine saturated hydraulic conductivity with rapid withdrawal or injection of a solid slug or slug of water from a well.

2.0 SCOPE

The following format will be used for the conducting, reporting, and analysis of all slug tests, whether conducted by JEG representatives or subcontractors.

- 3.0 PROCEDURE
 - 3.1 All data generated in performing slug tests will be entered on a JEG Slug Test Data form JEG-AL-ENG-8 (3/84).
 - 3.2 All spaces on the form shall be completed excepting: "open interval," "total depth of well," "well diameter," "residual ddn," "1/tm," and "H-h/H-Ho." The first 3 quantities are a part of the historic data base and may be completed prior to going to the field if it is desired to analyze the slug test in the field. The last three quantities may be computed from the slug test data.
 - 3.3 The depth to static water shall be measured and recorded prior to emplacement of the slug or bailer. The datum shall be the top of the well casing or, for uncased holes, the ground surface.
 - 3.4 Either slug-withdrawal or slug-injection tests will be performed. The type of test performed will be noted on the Slug Test Data Form.
 - 3.5 Prior to the start of the test, the slug or bailer shall be placed in the well or borehole so that the top of the slug or bailer is below the level of the static water level.
 - 3.6 After the slug has been emplaced, the test observers shall allow the water level to equilibrate as nearly as possible to the static water level. When in the judgment of the test observers the water level has been allowed to equilibrate as near as is practicable to the static water level, the water level will be recorded (to + .01 foot) on the form as the initial water level

JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.3

PAGE 2 OF 2

using the datum as the top of casing or, for uncased tries, the ground surface. The initial water level will also be recorded as the first entry in groundwater depth measurements.

- The test will start upon removal or addition of the slug from the water, and that moment will be designated as "to". The 3.7 elapsed time will be measured from that moment using a stop watch or other accurate and precise timing device.
- Periodically, after the removal of the slug, the depth to water 3.8 will be measured and recorded on the form. The datum will be the top of casing, or, for uncased holes, the ground surface. The depth will be recorded in feet to an accuracy of + .01 foot. The corresponding elapsed time will be recorded as elapsed time in minutes and seconds. The initial frequency of measurement shall be approximately every 30 seconds. After about 5 minutes, the frequency of measurement shall be approximately every minute. Measurement frequency will depend upon the rate of recharge in each individual well. After about 20 minutes, the observers shall make further measurements at their convenience. The test shall be conducted for at least 20 minutes or until the water level recovers to w': in 0.10 foot of the initial water level. If pressure transducers are used as the primary water-level measurement device, the values should be checked and calibrated with a sounder or tape, approximately every 20 minutes.
- Tests shall be conducted on relatively undisturbed wells. If a 3.9 test is conducted on a well that has recently been pumped for water sampling purposes, the measured water level must be within 0.10 foot of the water level prior to sampling. At least one week should elapse between the drilling of a well and the performance of a slug test.
- 3.10 Upon completion of the slug test, the completed form shall be checked by the Site Hydrologist. Upon acceptance as being complete and accurate, the data shall be stored in the technical data base as described in Section 8.3.1.

S M Theyi Prepared By: TAC Hydrologist Task Magager Reviewed By: Duality Assyrance Manager Reviewed By: 1 Mit Approved By: Project Manager



JACOBS ENGINEERING GROUP INC.

AL RUOLEROUE OPERATIONS MANUAL	SECTION 16.1.4
ALBOQUERQUE OFERATIONS MANUAL	DATE 11-2-89
	EFFECTIVE 12-08-89
PACKER (PERMEABILITY) TESTING	SUPERSEDES 05/27/87
	PAGE 1 OF 5

1.0 PURPOSE

 To provide procedures, equipment requirements, and test methods for conducting packer tests and analyzing data.

2.0 SCOPE

2.1 This procedure is applicable to members of TAC and its contractors and its subcontractors who have responsibilities and duties for the installation of well(s), well development, packer testing, data analyses, and collection of water samples for water quality determinations.

3.0 EQUIPMENT

- 3.1 Packer unit(s)
- 3.2 Extra glands
- 3.3 Strapping Tool
- 3.4 Electric Tape: several rolls to seal "A" rod joints
- 3.5 Top of "A" rod connector with pressure gauge
- 3.6 Inree (3) sections of water hose
- 3.7 In-line straight pipe sections near flowmeter
- 3.8 Two (2) flowmeters: 1+50 gpm, 50-350 gpm
- 3.9 By-pass "T"
- 3.10 Stopwatch
- 3.11 Air line
- 3.12 CO2 tank
- 3.13 CO2 regulator
- 3.14 Air line couplings
- 3.15 Thermometer
JACOBS ENGINEERING GROUP INC. SECTION 16.1.4 ALBUQUERQUE OPERATIONS

3.15 Water supply tank

3.17 Pump

3.18 Packer (Permeability) Test Forms

4.0 PROCEDURE

4.1 Record all information and data from a test on a Packer (Permeability) Test Form (Figure 4.1) pursuant to TAC data reporting formats and protocols.

PAGE 2 OF 5

- 4.2 Perform pre-test preparation. Before introducing the test equipment into the borehole, flush the boring extensively with clean potable water to rid hole of all drilling fluids and cuttings. This is best. accomplished at the end of last core run, while the tool string is still in the poring. If possible, avoid the use of revert or bentonite while drilling the hole. If drilling mud must be used, use revert and do not packer test the boring for 72 hours after drilling.
- 4.3 Set up the equipment in accordance with Figure 4.2. Introduce the packer unit, "A" rod string, and air line into the borehole. All connections should be kept as short and straight as possible with a minimum number of changes in the diameter of hose, pipe, etc. Measure the height of the water pressure gauge above ground level, the depth to the top of the test interval, and the length of the test interval. Record the data to the nearest 0.1 foot or 0.01 foot on the test form. NOTE: DO NOT USE INCHES.

The length of the test section is governed by the character of the rock, but should be between 10 and 20 feet in length. Except under the most adverse conditions, the use of test sections greater than 20 feet in length is not recommended. Longer test sections may not permit sufficient localization of permeable zones and may complicate calculations.

The test should be performed using centrifugal pumps having sufficient capacity to develop back pressure. Muitiple cylinder pumps should be avoided due to the fluctuating pressures which create difficulty in making accurate readings because the high and low readings must be averaged to determine the approximate true effective pressure. A pump with a capacity of up to 250 gpm against a total head of 160 feet is adequate for most tests.

4.4 Before inflating the packer, circulate clean potable water through the system using low flow and, if possible, establish a head at the top of the hole. This helps to check the packer seal. During this procedure check the flowmeter performance to assure reliability. Have two flowmeters available to cover a range of flow from less the gpm to 400 gpm.

JACOBS ENGINEERING GROUP INC. SECTION __ 16.1.4 ALBUQUERQUE OPERATIONS

PAGE 3 CF 5

- 4.5 Stop the water flow and inflate the packer unit(s). The pressure is a function of the depth to the test interval and rock conditions. Experience has shown that 35 to 45 psi is usually sufficient. Higher pressure can be used; however, problems with leakage may be more pronounced. During this procedure check for possible leaks (i.e., air bubbles in borehole head, leakage at tank, etc.).
- 4.6 Commence water flow to the test interval using only clean potable water. Regulate the water pressure using the by-pass valve (Figure 4.2). Avoid exceeding a maximum water pressure of 25 psi. Record the water temperature on the packer test form. In tests above the water table, pump water at the desired pressure into the test section for about 10 minutes in coarse materials or 20 minutes in fine-grained materials before making measurements. Check for poor packer seal by checking whether water is rising in the borehole.
- 4.7 Commence the actual test holding the water pressure as constant as possible. Record the water flow rate and water pressure, the air pressure of the packer, and the elapsed time, e.g., 0.0 minutes, 0.5 minutes, 1.0 minutes, 1.5 minutes, etc. Record every 30 seconds for at least 10 minutes. Test should be run until stabilization occurs; that is, until three or more readings of flow rate and pressure are essentially equal. The test can be cut short if flows are too nigh to establish pressure. However, a constant inflow value should be established. If there is no flow, check all the joints, connections, and hoses to make sure they are tight so no water loss occurs and cneck the lines for blockage.
- 4.8 when test is complete depressurize the packer and allow a minimum of seven (7) minutes for packer deflation before withdrawing equipment from borenole.
- 4.9 Attachments

4.9.1 Figure 4.1 Packer (Permeability) Test Form

4.9.2 Figure 4.2 Schematic Diagram of Packer System Set-Up

5.0 TEST METHODS

5.1 The two methods presented are described in detail in the U.S. Bureau of Reclamation (USBR) Ground Water Manual (1977). All figures and examples are from this reference. A schematic drawing of the two methods is shown in Figure 5.1.

The Ground Water Manual (1977) and the USBR Earth Manual (1974) provide additional packer (permeability) test methods for various conditions in individual drill noles in unsaturated and saturated material. These two references are readily available and provide access to other sources of literature on packer test methods which can be used to analyze test data. Values for head loss in pipes due to friction are available from Driscoll (1986). Also, other standard references are available on head loss in pipes and may be used in the computation of permeability.

JACOBS ENGINEERING GROUP INC. SECTION _______ ALBUQUERQUE OPERATIONS

PAGE_ 4 OF 5

- 5.2 Method 1: Involves the use of a single packer in consolidated rock. The hole may be cased down to the test interval or uncased depending on material stability. The hole is drilled to a given distance, the tools removed, and a packer is seated at a prescribed distance above the bottom of the hole. Water is pumped into the test section under pressure and readings are recorded. For additional tests deeper in the boring, the equipment is removed and the hole is drilled deeper. The packer is inserted again to isolate the newly drilled hole, and the test is repeated.
- 5.3 Method 2: Involves the use of two packers in consolidated rock. The material must be stable and not require casing above the test interval. The hole is drilled to the final depth and cleaned prior to testing. Two packers are spaced to isolate a desired test interval starting at the bottom of the hole and, after the test, lifted a prescribed distance to test the next section. This procedure is repeated until the entire hole is tested.
- 5.4 Examples: The equations presented in Figure 5.1 are based on the percent of unsaturated material. Three examples are provided addressing the three different zones of saturation. Figure 5.2 is to be used in delineating between Zone 1 and Zone 2. Figures 5.3 and 5.4 are to be used to determine the respective conductivity coefficients (Cu and Cs) for the equations provided in Figure 5.1.
- 5.5 Attachments
 - 5.5.1 Figure 5.1 Permeability test for use in saturated or unsaturated consolidated rock.
 - 5.5.2 Figure 5.2 Location of Zone 1 lower boundary, for use in unsaturated materials.
 - 5.5.3 Figure 5.3 Conductivity coefficients for permeability determination in unsaturated materials with partially penetrating cylindrical test walls.
 - 5.5.4 Figure 5.4 Conductivity coefficients for semispherical flow in saturated materials through partially penetrating cylindrical test wells.
 - 5.5.5 Examples 1, 2, and 3.

6.0 REFERENCES

Driscoll, F. G., 1985. Groundwater and Wells, second edition, Johnson Division, St. Paul, Minnesota, 1088 pp.

U.S. Bureau of Reclamation, 1974. Earth Manual, second edition, U.S. Government Printing Office, Washington, D.C., 810 pp.

U.S. Bureau of Reclamation, 1977. Ground Water Manual, first edition, U.S. Government Printing Office, Wasnington, D.C., 480 pp.

JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.4

	n111
Prepared	By: 11/h Comis KH-
Reviewed	By: 164 Bothell Task Manager
Reviewed	By: But Assurance Manager
Approved	By: Project Manager

FIGURE 4.1



JACOBS ENGINEERING GROUP INC.

PACKER (PERMEABILITY) TEST FORM

Fage____ of _____

DAVEPPTF.DRW

SITE ID	LOCATION ID	LOG DATE
LOGGING COMPANY		_ FIELD REP
BOREHOLE RADIUS (FT)_		PACKER TYPE: SINGLE/DOUBLE TEST
TOP PACKER LENGTH (F	T) BOT	TOM PACKER LENGTH (FT)
BOREHOLE DEPTH (FT B	ELOW DATUM)	
BOTTOM PACKER DEPTH	(BOTTOM OF HOLE (FT	BELOW DATUM)
TEST INTERVAL LENGTH	(FT)	
HEIGHT OF PRESSURE G	AUGE ABOVE DATUM (FT)
DIAMETER (IN)	LENGTH (FT)	OF INTAKE PIPE
STATIC GROUNDWATER	DEPTH BELOW DATUM	(FT)
NOTE: DATUM = GROUN	D SURFACE	

EQUIPMENT DESCRIPTION

ELAPSED TIME (MINUTES)	VOLUME (GALS)	WATER PRESSURE (PSI)	PACKER PRESSURE (S) (PSD)
			-
	THE OWNER AND A DESCRIPTION OF A		
	and the second second statement of the second s	and the second dependence of the second s	
		and the balance of the second s	
		The second s	
CONSTRUCTION OF THE PROPERTY OF THE DESCRIPTION	and the subscription to the second	and the second state of th	
The first state from the second state and the first state of the second state of the			Parallel Deputit Carlor and an and a subscription
the state of the second second second states and the		The second distribution of the second s	and the second state with a second
	and the second	and the second	
		· · · · · · · · · · · · · · · · · · ·	
The factor is an internet or the same as the factor is a second sector in the same			
The second second data as the second second second	A state in the second second second second second		
The Rest Name of Concession of			
MMENTS:			
	and the second		The second s

JEG-AL-ENG-62 (10/89)

9





REF: U.S. BUREAU OF RECLAMATION (1977)

K = coefficient of permeability, feet per second under a unit gradient Q = steady flow into well, ft²/₈

 $H=h_1+h_2-L=$ effective head. ft

 h, (above water table) = distance between Bourdon gage and bottom of hole for method 1 or distance between gage and upper surface of lower packer for method 2. ft

h, (below water table) * distance between gage and water table. ft

 h_2 =applied pressure at gage, 1 lb/in² = 2.307 ft of water

L *head loss in pipe due to friction, ft; ignore head loss for Q<4 gal/min in 1¹/₄ - inch pipe; use length of pipe between gage and top of test section for computations

 $X = \frac{H}{T}(100)$ - percent of unsaturated stratum

A=length of test section, ft

r = radius of test hole. ft

Cu² conductivity coefficient for unsaturated materials with partially penetrating cylindrical test wells

Cs=conductivity coefficient for semi-spherical flow in saturated materials through partially penetrating cylindrical test wells

U=thickness of unsoturated material. ft

S= thickness of saturated material, ft

 T_{u} = U - D + H = distance from water surface in well to water table, ft

D=distance from ground surface to bottom of test section. ft

a =surface area of test section, ft²; area of wall plus area of bottom for method 1; area of wall for method 2

Limitations:

 $Q/o \le 0.10$, $S \ge 5A$, $A \ge 10r$, thickness of each packer must be $\ge 10r$ in method 2

FIGURE 5.1 PERMEABILITY TEST FOR USE IN SATURATED OR UNSATURATED CONSOLIDATED ROCK



REF: U.S. BUREAU OF RECLAMATION (1977)



Cu = UNSATURATED CONDUCTIVITY COEFFICIENT



REF: U.S. BUREAU OF RECLAMATION (1977)

FIGURE 5.3 CONDUCTIVITY COEFFICIENTS FOR PERMEABILITY DETERMINATION IN UNSATURATED MATERIALS WITH PARTIALLY PENETRATING CYLINDRICAL TEST WELLS

ALCHER P. AND AN



A or A re

REF: U.S. BUREAU OF RECLAMATION (1977)

FIGURE 5.4 CONDUCTIVITY COEFFICIENTS FOR SEMISPHERICAL FLOW IN SATURATED MATERIALS THROUGH PARTIALLY PENETRATING CYLINDRICAL TEST WELLS

Example 1:

Zone 1—Method 1 Given: U=75 ft, D=25 ft, A=10 ft, r=0.5 ft, $h_1=32$ ft, $h_2=25$ lb/in²=57.8 ft, and Q=20 gal/min=0.045 ft³/s.

Head loss L for a 1%-inch pipe at 20 gal/min is 0.76 foot per 10-foot section. If the distance from the Bourdon gage to the bottom of the pipe is 22 ft, the total L is (2.2)(0.76)=1.7 ft.

 $H = h_1 + h_2 - L = 32 + 57.8 - 1.7 = 88.1$ ft of effective head. $T_2 = U - D + H = 75 - 25 + 88.1 = 138.1$ ft

$$X = \frac{H}{T_u} (100) = \frac{88.1}{138.1} (100) = 63.8 \text{ percent}$$
$$\frac{T_u}{A} = \frac{138.1}{10} = 13.8$$

The values for X and $\frac{T_u}{A}$ lie in zone 1 (fig. 5.2). To determine the conductivity coefficient C_u from figure 5.3 :

$$\frac{H}{r} = \frac{88.1}{0.5} = 176.2$$

$$\frac{A}{H} = \frac{10}{88.1} = 0.11 \text{ and } C_u = 62$$

Then,

$$K = \frac{Q}{C_u r H} = \frac{0.045}{(62)(0.5)(88.1)} = 0.000016 \text{ ft/s}$$

REF: U.S. BUREAU OF RECLAMATION (1977)

FIGURE 5.5 PRESSURE PERMEABILITY TESTS EXAMPLES OF METHODS 1 AND 2 Example 2:

Zone 2

Given: U, A, r, h_2 , Q, and L are as given in example 1. D=65 ft, and $h_1=72$ ft.

If the distance from the Bourdon gage to the bottom of the intake pipe is 62 ft, the total L is (6.2)(0.76) = 4.7 ft.

H = 72 + 57.8 - 4.7 = 125.1 ft $T_u = 75 - 65 + 125.1 = 135.1$ ft

$$X = \frac{125.1}{135.1}$$
 (100) = 92.6 percent and $\frac{T_u}{A} = \frac{125.1}{10} = 13.5$

The test section is in zone 2 (fig. 5.2). To determine the conductivity coefficient C_i from figure 5.4 :

$$\frac{A}{r} = \frac{10}{0.5} = 20$$
 and $C_s = 39.5$

Method 1

$$K = \frac{2Q}{(C_{\star}+4)r(T_{\star}+H-A)} = \frac{(2)(0.045)}{(39.5+4)(0.5)(135.1+125.1-10)}$$

== 0.000016 ft/s

Method 2

$$K = \frac{2Q}{(C_{*}r)(T_{*}+H-A)} = \frac{(2)(0.045)}{(39.5)(0.5)(135.1+125.1-10)} = 0.000018 \text{ ft/s}$$

Example 3: Zone 3 Giran: U.A.r.b.

Given: U, A, r, h_2, Q , and L are as given in example 1. D=100 ft, $h_1=82$ ft, and S=60 ft.

If the distance from the Bourdon gage to the bottom of the intake pipe is 97 ft, the total L is (9.7)(0.76) = 7.4 ft.

H = 82 + 57.8 - 7.4 = 132.4 ft

$$\frac{A}{r} = \frac{10}{0.5} = 20$$
 and $C_{*} = 39.5$ from figure 5.4

Method 1

$$K = \frac{Q}{(C_* + 4)rH} = \frac{0.045}{(39.5 + 4) (0.5) (132.4)} = 0.000016 \text{ ft/s}$$

Method 2

$$K = \frac{Q}{C_{i} r H} = \frac{0.045}{(39.5) (0.5) (132.4)} = 0.000017 \text{ ft/s}$$

REF: U.S. BUREAU OF RECLAMATION (1977) FIGURE 5.5 PRESSURE PERMEABILITY TESTS EXAMPLES OF METHODS 1 AND 2 (CONCLUDED)

JE JACOBS ENGINEERING GROUP INC.

AL PLIQUEDOUE OPERATIONS MANULAL	SECTION	16.1.5
ALBOQUERQUE OFERATIONS WANDAL	DATE	5/27/87
	EFFECTIVE	08/18/88
ADUTEED DIMP TESTING	SUPERSEDE	s 8.2.12 4/8/85
AQUITER FORF TESTING	PAGE 1	OF 3

1.0 PURPOSE

To define the procedures to be followed by TAC hydrologists in conducting pumping tests of aquifers for characterization of hydrogeologic units on the UMTRA Project.

2.0 SCOPE

The pumping test will be performed to estimate the hydrologic characteristics of the hydrogeologic interval. The equipment used in the test will depend on the rate of flow of fluid encountered and drill hole stability conditions. Because of the great variety of hydrogeologic conditions that are likely to be encountered during drilling and testing, several different test tool arrangements are planned in order to be able to perform the pumping test.

3.0 PROCEDURE

- 3.1 The test methods used will follow the guidelines for well testing in Technical Memorandum No. 1 (April 9, 1985) which is provided as Attachment #2. The data obtained during the test, regardless of method or equipment, shall include:
 - a) Static water level.
 - b) Water discharge rate.
 - c) Pressure versus time data for each gauge.
 - d) Water temperature and specific conductivity.
 - e) Depth of water intake.
 - f) Whether or not a flowmeter/trace ejector survey was performed.
- 3.2 An aquifer test data form (Attachment 1) shall be completed for each test in order to document the test results. This form shall be the lead page of the data records of the test. The complete, original test data records shall be sent to the Albuquerque Office, to the attention of the site hydrologist, after each test, and filed in the Project Document Control Center files. Additionally, a copy of these records shall be retained by the field representative in charge of the test.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.5

PAGE 2 OF 3

- 3.3 A diagram of the equipment actually used in each pumping test shall be prepared by the TAC site hydrologist or received from the subcontractor responsible for the testing. The diagram will show the tubing dimensions, the depth of water intake, and the location of gauges and packers. This diagram shall become part of the data records of each test.
- 3.4 All gauges and flowmeters used in the pumping test procedures shall be calibrated before arrival on site. Copies of the documentation of instrument calibration shall be obtained from the instrument or testing service company by the TAC site hydrologist. The calibration records shall consist of laboratory measurements and, if performed, any on-site zero adjustment and/or calibration.
- Water pumped from the well during the pumping test shall be 3.5 disposed of in such a manner so that no adverse impacts are created on surface water quality or on the quality of the drawdown data from observation wells.
- In cases where a weir or an orifice are used to measure flow 3.6 volumes or rates, these devices will be checked on-site, using a bucket of known volume and stopwatch, and their accuracy will be documented before testing proceeds.
- All equipment shall be installed, and it shall be demonstrated 3.7 to the technical representative that the system is in proper working order and performing to specifications at least 24 hours before the proposed start of the pumping test.
- For air-lift pumping tests, water samples will be collected by 3.8 wireline sampler which shall be put down the hole after the pumping has stopped. The actual depth of the interval which is producing the fluid and the depth at which the sample was collected shall be recorded.
- For pumping tests which employ mechanical pumps, samples will be 3.9 collected above ground level from the discharge line.
- 3.10 After collection, the water sample will be treated in accordance with Water Sampling/Preserving/Shipping and Testing Standard Operating Procedure 16.2.1, and will be analyzed by a geochemical laboratory in accordance with Standard Operating Procedure 16.2.3.

4.0 REFERENCES

Thackston, J., 1985, Technical Memorandum No. 1, "Bore Hole Hydraulic Testing for Aquifer Characterization."

ALBUQUERQUE OPERATIONS	SECTION 16.1.5
------------------------	----------------

THE MARKE

The Martin

Prepared	By: Jahn R. ashipuy
Reviewed	By: Task Managen
Reviewed	By Bull Assurance Manager
Approved	By: AProject Manager

Ward March March 19



JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.5 Tech Memo 1 ALBUQUERQUE OPERATIONS

TABLE OF CONTENTS

TECHNICAL MEMORANDUM NO. 1 BORE HOLE HYDRAULIC TESTING FOR AQUIFER CHARACTERIZATION

Section																									Page
1.0	Introduc	tion				4			*					÷,		4				i,			÷	x.	1
1.1 1.1.1 1.1.2	Slug Tes Purpose Design, of Slug	Dril	lin We	 g a	nd	Co	nst	tru	c.t	i oi	, n	•	•	•	•	•		•	•	•	•	•	* *	•	1
1.1.3 1.1.4 1.1.5	Equipmer Procedur Data Ana	nt . re . ilysi	s	 	•	•	• •			•		•		•		•		•	•	•	•	•	•		3 4 5
1.2 1.2.1 1.2.2 1.2.3 1.2.4 1.2.5 1.2.6	Aquifer Purpose Pump Wel Equipmer Observat Procedur Data Ana	Pump 1 De it ion res ilysi	ving sig Wel s	Te n . i S	st iti		• •																		6 6 7 8 9 10 12
List of R	eferences	5		• •					,	×	i.		÷	ŝ					•	a,				•	14.
Figure 1-	1 - Range	of	Hyd	rau	110	: C	ond	iuc	ti	vi	ty	٧ĉ	11	les						,		•		,	15
Appendix after Coop	1-1 - Exa per, Brec	ample jehoe	of ft,	An an	aly d F	vsi Pap	s (add	of	S1 10	ug s	Te (19	e s 1 96 7	t, 7)												16
Appendix Record an Rice, 197	1-2 - Exa d Analysi 6	ample is of	of S1	Pr ug	ess Tes	sur st,	e at	fra fte	ns r l	du Boi	cei uwe	r (er	Cha ar	art nd										•	22
Appendix	1-3 - Exa	ample	of	An	aly	si	s c	of	Aq	ui	fei	r 1	Tes	st	Da	ata	3								26
Attachmen	t 1 - Aqu	lifer	Te	st	Dat	ta	For	rm											,			i.			28

JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS

16.1.5

TECHNICAL MEMORANDUM NO. 1 BORE HOLE HYDRAULIC TESTING FOR AQUIFER CHARACTERIZATION

. 1.0 INTRODUCTION

This technical memorandum provides guidelines and procedures for determining the hydraulic parameters of aquifers using slug tests and pumping tests. The hydraulic conductivity (K) of an aquifer is a measure of the aquifer's ability to conduct water. The greater the value of hydraulic conductivity, the more readily the aquifer conducts water. Figure 1 illustrates. the typical range of hydraulic conductivity values associated with a variety of earth materials. The transmissivity (T) of an aquifer is defined as the ability of the aquifer to transmit water through its entire thickness and is equal to the hydraulic conductivity (K) times the saturated thickness of the aquifer (b); this can be stated as T = Kb, for cases in which K is constant with depth. Otherwise, transmissivity is equal to the integration of hydraulic conductivity over depth, which can be stated as T = K(Z)dz.

The specific yield of an aquifer is the ratio of the volume of water drained by gravity from a unit volume of aquifer material. The storage coefficient of an aquifer is the volume of water in storage released from a column of aquifer with unit cross section under a unit decline of head.

SLUG TEST 1.1

1.1.1 PURPOSE

The purpose of this section is to outline procedures for the conduction and evaluation of slug tests, which are a quick and inexpensive method of estimating the ydraulic conductivity or transmissivity of many aquifer systems.

A slug test will generally work in aquifers where the hydraulic conductivity is less than or equal to 10-2 cm/sec. In aquifers of greater hydraulic conductivity the water level may return to static level before a sufficient number of water level readings can be made. In aquifers of low hydraulic conductivities (10-4 to 10-7 cm/sec), slug tests are generally a reliable field method of determining hydraulic conductivity or transmissivity near the screened zone of the well.

The advantages of using slug tests to estimate hydraulic conductivities are:

o Estimates can be made in situ and errors incurred in laboratory testing of disturbed samples are avoided.

JE JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS

16.1.5

- o Tests can be performed quickly at relatively low cost, because a pumping well and observation wells are not required.
- o The hydraulic conductivity of small discrete portions of an aquifer can be made, such as sand layers in a clay.

There are disadvantages in using slug tests which must be evaluated before the test is conducted. These include:

- o Only the hydraulic conductivity of the area immediately surrounding the well is estimated. These results may not be representative of the average hydraulic conductivity of the area.
- o Often only a range of hydraulic conductivity values rather than an average value can be deduced from test results.
- o Certain assumptions are made in the analysis process. If the assumptions made are inappropriate for the geologic conditions at the site, the results may be erroneous.
- o The storage coefficient (S) usually cannot be determined.
- o Data sufficient for analysis may not be collected if the hydraulic conductivity is relatively high.

This technical memorandum explains procedures and equipment commonly used in conc. ing most slug tests, and a list of analysis techniques used under different geologic conditions and for different bore hole or well systems. Also included are example calculations of typical slug tests (Appendices 1-1 and 1-2).

1.1.2 DESIGN, DRILLING AND CONSTRUCTION OF SLUG TEST WELL

Several factors enter into the design of a well for the performance of slug testing. These include depth and diameter of the test well, length of screened interval and size of screen openings. Some of these factors can be chosen as drilling proceeds or be predetermined from previous borings. Well design is generally determined from an evaluation of drill cuttings, corings, and geophysical logging.

The depth and screened interval of the hole are determined by the location of the zone of interest. The diameter of the hole and casing to be inserted in the hole should be large enough in order that the volume of the slug does not overflow the top of casing. This may be a problem when the water level in the bore hole is relatively close to the ground surface. The screened interval should consist of screen with a percentage of open intervals large enough so as not to excessively restrict water flowing from the bore hole to the subsurface materials tested. Wells with screened intervals which require a gravel pack should be packed with material of a grain size distribution that would result in approximately an order of magnitude of higher hydraulic conductivity than the subsurface materials being tested.

IC	JACOBS ENGINEERING GROUP	INC.	SECTION Tech Memo 1
JE	ALBUQUERQUE OPERATIONS		PAGEOF28

A variety of drilling techniques can be used in completing the test well. These may include mud-rotary, air-rotary, coring and auger techniques. The type of drilling method used depends on the behavior of the material being drilled and the specific objectives of the project. Drilling without drilling fluids is preferred because this prevents further contamination of the subsurface. Mud-rotary techniques are sometimes required in order to lubricate the bit and hold up the bore hole walls when relatively thick, coarse, unconsolidated materials are encountered. Augering is normally performed without the use of drilling fluids and used in the drilling of shallow holes with relatively cohesive materials.

The construction of the test well is relatively simple and involves the assembly of the appropriate diameter casing (usually PVC because of lower cost) for insertion into the bore hole. A pre-slotted screened section is prepared and attached to the casing string, normally with a cap on the bottom (to prevent material from entering the bottom of the screen), and the whole assembly is lowered into the hole. If a gravel pack is necessary as a filter due to the aquifer containing very fine unconsolidated sand, then it should be placed opposite the screened interval using a tremie pipe. The annulus is then grouted from the top of the gravel pack up to the ground surface in order to allow hydraulic connection only between the test well and the zone to be tested. The hole should be flushed or bailed before grouting if drilling fluid remains in the hole. If the ground water in the hole is highly contaminated, special safety precautions should be taken to control the discharge water before test well development.

1.1.3 EQUIPMENT

Equipment necessary to conduct a slug test includes:

- o Well or bore hole
- o Water level measuring device
- o Known solid volume of particular shape (slug) that will easily fit into the well or bore hole or container to add or remove a known volume of water
- o Watch with second hand
- o Semi-log graph paper
- o Indelible pen or pencil and paper

A number of water level measuring devices can be used for measuring the depth to water in a well. An excellent device for measuring the depth to water during a slug test is the pressure transducer. An example strip chart data record is shown in Appendix 1-2. Accurate readings may also be obtained with electric water level indicators or weighted tapes with ploppers. Water level measurement by the wetted-tape method should be avoided as it may be difficult to obtain a sufficient number of readings for analysis of the test if the water level recovers in a short amount of time.



JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS

1.1.4 PROCEDURE

Before beginning the slug test, the following information should be recorded:

- o Site ID and well or bore hole location
- o Location and elevation of reference point from which water depth measurements are made
- o Elevation of ground water with respect to the reference point
- o Date and time of test
- o Well depth, screen length and depth, riser pipe radius, well screen radius, and radius of the gravel pack, plus the bore hole depth and radius
- o Aquifer or ground water zone being tested
- o Volume of water added or withdrawn or volume of solid cylinder (slug)
- o Type of measuring device used
- o Names of personnel conducting test

The following procedures will apply to most slug tests. The field procedures required for a particular slug test may be slightly different, however, and the procedures should be modified to that test as necessary.

- o Determine the static water level in the well by measuring the depth to water periodically for several minutes and taking the average of the readings.
- "Instantaneously" introduce or remove a known volume of water to 0 the well. Another method is to introduce a solid cylinder of known volume to displace and raise the water level, allow the water level to restabilize and remove the cylinder. It is important to remove or add the volumes as quickly as possible because the analysis assumes an "instantaneous" change in volume is created in the well.
- . With the moment of volume addition or removal assigned time zero, measure and record the depth to water and the time at each reading. Depths should be measured to the nearest 0.01 foot. The number of depth-time measurements necessary to complete the test are variable. It is critical to make as many measurements as possible in the early part of the test.

o Continue measuring and recording depth-time measurements until the water level returns to equilibrium conditions or a sufficient number of readings have been made to clearly show a trend on a semi-log plot of time versus depth.

The time required for a slug test to be completed is a function of the volume of the "slug", the hydraulic conductivity of the formation and the type of well completion. The slug volume should be large enough that a sufficient number of water level measurements can be made before the water level returns to equilibrium conditions. The length of the test may range from less than a minute to several hours.

If the well is to be used as a monitoring well, precautions should be taken that the well is not contaminated by introduced material. If water is added to the monitoring well, it should be from an uncontaminated source and transported in a clean container. Bailers or measuring devices should be cleaned prior to the test. If tests are performed on more than one monitoring well, care must be taken to avoid cross contamination of the wells.

1.1.5 DATA ANALYSIS

The analysis of slug test data is based on the modific: cion of well-known ground water flow equations (either the Theim equation, the Theis equation, or subsequent modification). Several autions have presented analytical solutions for the analysis of slug test data. Most solutions require a semi-logarithmic plot of the data collected: dimensionless head (logarithmic scale) or residual head (logarithmic scale) versus time (arithmetic scale).

Hvorslev (1951) was one of the first researchers to publish techniques of analysis of either constant or slug (falling head) tests in near-surface saturated soils. His analysis for slug tests involves a semi-logarithmic plot of the falling head (or water level) divided by the initial head against time. Basic algebraic equations are presented for different configurations of the soil relative to the test hole. In general, the permeability is proportional to a "shale factor" and inversely proportional to a "time lag". The "shape factor" is determined from the test well characteristics or dimensions. "Time lag" is determined from the semi-logarithmic plot.

Cooper et al (1967) and Papadopulos et al (1973) developed a set of type curves for analyzing slug test data, particularly for tests run in materials which are confined (under artesian pressure). The field data are plotted as dimensionless head (arithmetic scale) versus time (logarithmic scale) and matched to a set of type curves. The match point values are substituted into simple algebraic formulated to obtain a value for K (Appendix 1-1).

Bouwer and Rice (1976) and Bouwer (1978) developed a technique for analyzing slug test data collected from completely or partially penetrating wells in unconfined aquifers. Their analysis involves a plot of residual head (logarithmic scale) versus time (arithmetic



JE JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS

16.1.5

scale). A straight line is applied to the early-time data and used to calculate a value for K. An example of the use of this technique is presented in Appendix 1-2. A plot of the data and calculations are included.

Pressurized slug test methods have been developed for testing extremely low conductivity materials (10-8 cm/sec or lower) (Bredehoeft and Papadopulos, 1980). Basically, the pressurized slug technique is a modification of the conventional slug test discussed previously. The advantage of the pressurized slug technique is the reduction of time required to perform a test in tight formations. This method involves creating an instantaneous surcharge on drawdown in the test zone, then closing a valve to shut in the well. Based on the rate of decay of the pressure slug and the geometry of the test zone, the transmissivity, hydraulic conductivity, and storativity may be calculated.

AOUIFER PUMPING TEST 1.2

1.2.1 PURPOSE

The purpose of this section is to briefly describe the use of aquifer pumping tests to evaluate aquifer characteristics. Aquifer pumping tests, commonly referred to as pump tests, are used to determine hydraulic characteristics and properties of water-bearing zones. Pumping tests provide results that are often more representative of the aquifer's characteristics than do slug tests.

Aquifer characteristics which may be obtained from pumping tests include hydraulic conductivity (K), transmissivity (T), and specific yield (Sy) for unconfined aquifers and storage coefficient (S) for confined aquifers.

Equipment, personnel and time commitments needed to conduct pumping tests are greater than those required for slug tests. Briefly, a pumping test consists of pumping one well and recording the drawdown in the pumping well and in other nearby observation wells.

There are advantages and disadvantages of pumping tests. Some of these are listed below.

Advantages of pumping test:

- o A greater portion of the aquifer is tested and the results obtained may be more reliable and representative of the aquifer characteristics.
- o Coefficient of storage and specific yield values can be determined.

Disadvantages of pumping test:



16.1.5

- o In low permeability aquifers long-term pumping may be required to complete the test and obtain reliable aquifer characteristics.
- o Disposal of discharged water may require special provisions if the ground water is contaminated.
- Test may be very expensive when compared to slug tests.

There are several types of pumping tests. The most common type is the constant rate discharge test (Todd, 1980). Variable rate tests are also employed under some conditions. Although the analysis is more complicated, any sort of temporal variations in flow rate can be accounted for by assuming the law of superimpositions holds. Usually this is a good assumption. The most widely used variable rate tests are the step-drawdown test, the constant head test, and the air-lift pump test (Kruseman and DeRidder, 1976).

Another useful technique is injection testing. Injection tests, both constant rate and variable rate, are analytically identical to pumping tests except for considering the flow rate a surcharge on the aquifer rather than a withdrawal. The data quality is also similar. Although injection tests are commonly used in the petroleum industry (Earlougher, 1977), numerous applications exist in hazardous waste Obviously, water sampling for hydrochemical practice. characterization of the aguifer must be conducted prior to injection in application of this technique. The injection fluid should be free of suspended solids and should be of equal or higher quality than the ground water at the test site.

One major advantage of this type of test is that contaminated ground water is not removed from the formation and, thus, is not a disposal problem. A potential disadvantage of the injection test is that in certain cases the injection well would have to be able to withstand some induced hydraulic pressure. Cases where the injection rate is large enough to elevate the water level above the well casing would require a sealed well head configuration and proper well construction so that there would be no leakage of injected fluid on to the ground surface. These requirements may be relatively expensive and difficult to operate in the field without the necessary precautions. Once an injection system is working at a constant rate of flow, the data collected are analyzed in much the same way as those from an aquifer pumping test, except that the water level buildup is used in the analysis instead of drawdown.

1.2.2 PUMP WELL DESIGN

Design of the pump well is an important consideration in aquiier testing. In some cases an existing well will be used for pumping. When sufficient funds and conditions permit, a pumping well can be designed and constructed specifically for the test. The pumping well should be screened throughout the thickness of the aquifer to be tested with standard well screen. The well should be gravel packed,

JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.5 ALBUQUERQUE OPERATIONS PAGE 8 OF 28

16.1.5

if necessary, to minimize sand production in unconsolidated fine-grained aquifers. Standard well construction techniques are discussed in Johnson (1986). The well should be correctly sealed from overlying and underlying units that are not to be directly pumped, and so that leakage down the well annulus cannot occur which may interfere with the interpretation of the test. The completed pumping well should be developed by the appropriate methods to remove drilling fluid from the well, and to wash and grad the grain sizes of the gravel pack and surrounding aquifer materials. Proper development of the well may prevent an unexpected variation in the pumping rate during the constant discharge test, which could lead to inconsistent drawdown data from the pumping well.

1.2.3 EQUIPMENT

The amount of equipment necessary to perform a constant-rate pumping test is greater than that used in a slug test. This is part of the reason why the aquifer pumping test is considerably more expensive. Equipment needed for a pumping test includes:

- o Pumping well and pump. Pumps are commonly of the submersible or turbine type. The pumping well should be properly developed prior to testing.
- o One or more observation wells hydraulically connected to the pumped aquifer and completed to the proper specifications for the particular test.
- o An orifice, weir, flow meter, container or other type of water measuring device to accurately measure and monitor the discharge from the pumping well.
- o Sufficient pipe to transport the discharge from the pumping well away from the area to prevent infiltration in the area of influence of the pumped zone.
- o Gate valve on the discharge pipe to control the pumping rate.
- o Outlet near well head for water quality determination, and sampling.
- o Depth-to-water measuring devices for each observation well and the pumping well (these may include steel tapes, electric sounding probes, Stevens Recorders, or pressure transducers).
- o A thermometer and other water quality equipment necessary.
- o Watches capable of reading to the nearest second for all observers, or a stopwatch.
- o 3-cycle by 5-cycle log and 5-cycle semi-log graph paper.

- o Indelible pens or pencils, and forms for recording times and drawdown measurements at each well.
- o Appropriate references, and calculator for determinations in the field.
- o A barometer or recording barograph (for tests conducted in confined aquifers).

1.2.4 OBSERVATION WELL SITING

The location and number of observation wells depends on several factors:

- o Whether the aquifer tested is confined or unconfined
- o The thickness of the aquifer
- o The anistropy of the aquifer
- o Location of the screened interval of the pumping well relative to the total aquifer thickness
- o Location of aquifer boundaries, whether positive (lake or stream) or negative (impermeable boundaries)
- o Practical and economic considerations

Any number of observation wells may be considered. Often, four wells are desirable, three on a line passing through the center of the pumped well and one on a line normal to that line and also passing through the pumping well. A number of guidelines for location of observation wells are presented in the Ground Water Manual (prepared by the U.S. Department of Interior), and Kruseman and DeRidder (1976).

As a general rule for tests performed in both confined and unconfined aguifers, the observation wells should be screened or completed in a substantial portion of the aquifer thickness in approximately the median depth of the test zone. In some cases, special tests require that observation wells be selectively completed in several depth zones in order to accurately determine aquifer characteristics such as anisotropy and vertical hydraulic conductivity.

The location of observation wells, relative to the pumping well, generally depends on four aquifer conditions:

- o Most aquifers with fully penetrating pumping wells: observation wells should be located at a distance estimated by using the Theis formulation (Theis, 1935). Use of this formulation is described by Walton (1970). Assumed aquifer parameters are used to determine a location which will give the amount of drawdown required for proper analysis.
- o Thin aquifers with fully penetrating pumping wells: for confined aquifers the nearest observation well should be located at least 25

JE JACOBS ENGINEERING GROUP INC. SECTION __ IG. 1.5

16.1.5

feet from the pumping well. For unconfined aquifers, observation wells should generally be located 15 to 100 feet from the pumping well.

- o Thick, isotropic aquifers with a partially penetrating pumping well: observation wells should be located one and one-half to two times the aquifer thickness from the pumping well.
- o Thick, anisotropic aquifers with partially penetrating pumping well: observation wells should be located at a minimum distance from the pumping well equal to twice the thickness of the aquifer times the square root of the ratio of the horizontal to the vertical hydraulic conductivity.

1.2.5 PROCEDURES

The field procedures to conduct pumping tests are similar to field procedures to conduct slug tests. The basic procedure consists of monitoring the water level over time in the pumping well and each observation well as the pumping well is discharged at a constant rate.

The following data must be recorded accurately at the time the test is performed:

- o Well identification number or location of pumping well and each observation well
- o Location of each well
- o Location and elevation of reference point from which water depth measurements are made, and elevation of ground with respect to the reference point
- o Date and time of test
- o Well depth, pump depth, screen length, well radius and radius of gravel pack plus well screen for each well

All gauges, transducers, flow meters, etc. used in conducting pumping tests shall be calibrated before use at the site. Copies of the documentation of instrumentation calibration should be obtained by the TAC hydrogeologist and later filed with the test data records. The calibration records will consist of laboratory measurements and, if performed, any on-site zero adjustment and/or calibration. In cases where a weir or an orifice is used to measure flow rates, the device will be checked on site using a container of measured volume and a stopwatch. Accuracy of the meters must be verified before testing proceeds.

If funds are available, it is normally advised to monitor pre-test water levels at the test site for about one week prior to performance of the test. This can be accomplished by using a continuous-recording device such as a Stevens Recorder. This information allows the

determination of the barometric efficiency of the aquifer when barometric records are available. It also helps determine if the aquifer is experiencing an increase or decrease in head with time due to recharge or pumping in the nearby area, or diurnal effects of evaportranspiration. Changes in barometric pressure are recorded during the test (preferably with an on-site barograph) in order to correct water levels for any possible fluctuations which may occur due to changing atmospheric conditions. Pre-test water level trends are projected for the duration of the test. These trends and/or barometric changes are used to "correct" water levels during the test so that they are representative of the hydraulic response of the aquifer due to pumping of the test well.

During an aquifer test, water levels should be measured to give at least ten observations of drawdown within each log cycle of time. During the early part of the test, sufficient personnel should be available to use at least one person at each observation well and at the pumping well. After the first two hours, two people are usually sufficient to continue the test. It is not necessary that readings at the wells be taken simultaneously. It is very important that depth to water readings be measured accurately and readings recorded at the exact time measured. A typical aquifer pump test data form is shown as Attachment 1.

The duration of the test is determined by the needs of the project and the aquifer properties. One simple test for determining adequacy of data is when the log-time versus drawdown for the most distant observation well begins to plot as a straight line on the semi-log graph paper. There are several exceptions to this simple rule of thumb; therefore, it should be considered a minimum criteria. Different hydrogeologic conditions can produce straight line trends on long-time versus drawdown plots. In general, longer tests produce more definitive results. If budget permits, a duration of one to several days is desirable, followed by a similar period of monitoring the recovery of the water level. Unconfined aquifers and partially penetrating wells may have shorter test durations. A knowledge of the local hydrogeology, combined with a clear understanding of the overall objectives of the project, is necessary in interpreting just how long the test should be conducted. There is no need to continue the test if the water level becomes constant with time. This normally indicates that a hydrogeologic source has been intercepted and that additional useful information will not be collected by continuing the pumping portion of the test.

Measurement of recovering water levels after pumping is concluded is often done to verify results obtained from the pumping portion of the test. The recovering water levels in the pumping well and the observation wells are measured for a period immediately following cessation of pumping. Monitoring of the recovery period should, by rule of thumb, be at least one-half the length of the pumping portion of the aquifer test. This length of time usually allows the collection of useful data for additional analysis. JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.5 ALBUQUERQUE OPERATIONS PAGE 12 OF 28

16.1.5

1,2,6 DATA ANALYSIS

Numerous techniques of analysis have been developed by researchers to evaluate water level data collected from constant discharge aquifer pumping tests. Many of the analyses use the graphical "curve matching" technique which involves the matching of theoretical "type" curves to plots of log-drawdown versus log-time from the observation wells. Other analyses rely on other graphical techniques such as application of a straight line to plots of drawdown versus log-time.

Theis (1935) developed a theoretical formulation which analyzed the lowering of the potentiometric level in an aquifer due to the constant withdrawal of water from the aquifer. This classic formulation has been widely used to predict water level response in aquifers due to specified pumping stresses. Several authors have described the use of this formulation in estimating transmissivity and storativity from aquifer test data Lohman (1972), Walton (1970), Todd (1980), and Freeze and Cherry (1979). In brief, the method involves superimposing the Theis solution (or "type" curve) on a log-log plot of drawdown versus time data from a test well. The coordinate axes of the type curve must be kept parallel to the field data plot axes. Data from the "match point" is used to solve two algebraic equations which give the value of transmissivity (T) and storativity (S) (or storage coefficient). Hydraulic conductivity (K) may be computed by dividing T by the aquifer thickness (b). An example of the application of this technique is presented in Appendix 1-3.

The same aquifer parameters can also be determined from a semi-log plot of drawdown versus time for either the pumping well or observation wells. Cooper and Jacob (1946) showed that after a relatively short time after pumping begins an analysis of the data, plotted as described above, provided essentially the same T, S, and K values as the Theic technique.

Hydrogeologic judgment should be used in applying straight line (Cooper-Jacob method) fits to these plots because different hydrogeologic conditions can produce similar trends in the data plot. For example, bounded aquifers produce straight line trends that may yield apparent transmissivity values that are too low. Knowledge of the hydrogeology of the area and the type of aquifer conditions is essential for correct analysis of test data.

Since the development of the ineis equation, several other formulations have been published which attempt to describe the response of certain aquifers to a constant pumping stress. Hantush (1955, 1956) developed formulations and a corresponding set of type curves for an aquifer which is overlain and/or underlain by a confining bed which has significant storage and experiences vertical flow when the adjacent aquifer is pumped. The same curve matching procedure is used here as with the Theis type curve except that several type curves are available for matching to the data depending on the degree of leakage into the pumped aquifer. Hydraulic

characteristics of the confining unit can be calculated using the set of Hantush type curves.

Another commonly used set of "type" curves was developed by Boulton (1954, 1963) and is used when the aquifer tested is unconfined and may exhibit a phenomenon known as delayed yield. Delayed yield is a result of retardation of drainage due to such a force as capillary tension. Boulton's type curves include a set of early-time type curves and late-time type curves. The early-time data and late-time data are respectively matched to the early-time and late-time type curves to give values for transmissivity, storage coefficient, and specific yield. The particular curve used in the set of "type" curves yields additional information concerning how the unconfined aquifer responds to pumping. Neuman (1972, 1974, 1975) also developed a set of type curves for the response of unconfined aquifers to pumping but uted different assumptions concerning the physical processes in ellect. In practice, Neuman and Boulton curves have been shown to give similar results.

Other less common analytical techniques have been described in detail by the authors mentioned above. A publication by Kruseman and DeRidder (1976) is one of the most complete works on the analysis and evaluation of pumping test data. Methods described (with practical examples) include treatment of aquifers from confined to unconfined under both non-steady and steady-state pumping conditions. Methods of correcting drawdown data from tests in which partially penetrating wells exist are also discussed by Kruseman and DeRidder (1976). In addition, the analyses of data from aquifers which have a non-ideal shape or are bounded in some fashion by different vdrogeologic units are presented. Techniques for analysis of recovery data collected after the completion of pumping are presented. Analyses for various types of variable discharge rate tests, such as the Aron-Scott method, are also presented in Kruseman and DeRidder (1976).

This memo has briefly described equipment, procedures, and other factors which must be considered for constant-rate pumping tests. keferences which provide additional information are presented in the list of references. Ground Water and Wells (Johnson, 1986) is a field guide with many suggestions on how to perform the tests. Ground Water Hydrology (Todd, 1980) and other references, such as Kruseman and DeRidder (1976), provide analytical techniques and example problems of pumping tests conducted under different geologic conditions.

JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS

PAGE 14 OF 28

REFERENCES

Boulton, N. S., 1954, "The Drawdown of the Water-Table under Non-Steady Conditions Near a Pumped Well in an Unconfined Formation," Paper 5979 in Proceedings of the Institution of Civil Engineers, Vol. 3, p. 564.

Boulton, N. S., 1963, "Analysis of Data from Non-Equilibrium Pumping Tests Allowing for Delayed Yield from Storage," Paper 6693 in Proceedings of the Institution of Civil Engineers, Vol. 26, pp. 469-482.

Bouwer, H., 1978, Groundwater Hydrology, McGraw-Hill.

Bouwer, H., and Rice, R. C., 1976, "A Slug Test for Determining Hydraulic Conductivity of Unconfined Aquifers with Completely or Partially Penetrating Wells," Water Resources Research, Vol. 12, No. 3.

Bredehoeft, J. D., and Papadopulos, S. S., 1980, "A Method for Determining the Hydraulic Properties of Tight Formations," Water Resources Research, Vol. 16, No. 1, pp. 233-238.

Cooper, Jr., H. H., Bredehoeft, J. D., and Papadopulos, S. S., 1967, "Response of a Finite-Diameter Well to an Instantaneous Charge of Water, Water Resources Research, Vol. 13, No. 1.

Cooper, Jr., H. H., and Jacob, C. E., 1946, "A Generalized Graphical Method for Evaluating Formation Constants and Summarizing Well-Field History," American Geophysical Union Transactions, Vol. 27, No. 4, pp. 526-534.

Earlougher, R. C., 1977, Advances in Well Test Analysis, Society of Petroleum Engineers of AIME.

Ferris, J. G., Knowles, D. B., 1954, "The Slug Test for Estimating Transmissivility," U. S. Geological Survey Ground Water Note 26.

Freeze, R. Allen, and Cherry, John A., 1979, Groundwater, Prentice-Hall, Inc., Englewood Cliffs, New Jersey.

Hantush, M. S., 1956, "Analysis of Data from Pumping Tests in Leaky Aquifers," American Geophysical Union Transactions, Vol. 37, No. 6, pp. 702-714.

Hvorslev, 1951, "Time Lab and Soil Permeability in Ground-Water Observations," Bulletin No. 36, U.S. Army Corps of Engineers, p. 50.

Johnson Division, 1986, UOP, Inc., Ground Water and Wells, Johnson Division, UOP, Inc., St. Paul, Minnesota.

Kruseman, G. P., and DeRidder, N. A., 1976, "Analysis and Evaluation of Pumping Test Data," International Institute for Land Reclamation and Improvement, Wageningen, The Netherlands.

ŧ.

JE JACOBS ENGINEERING GROUP INC.

SECTION Tech Meno 1 PAGE 14A OF 28

Lohman, S. W., 1972, "Ground-Water Hydraulics," U.S. Geological Survey. Professional Paper 70%, p. 70.

Neuman, S. P., 1172, "Theory of Flow in Unconfined Aquifers Considering Delayed Response of the Water Table," Water Resources Research, Vol. 8, No. 4, p. 1031.

Neuman, S. P., 1974, "Effect of Partial Penetration on Flow in Unconfined Aquifers Considering Delayed Gravity Response," Water Resources Research, Vol. 10, No. 2, p. 303.

Neuman, S. P., 1975, "Analysis of Pumping Test Data from Anistropic Unconfined Aquifers Considering Delayed Gravity Response," Water Resources Research, Vol. 11, No. 2, p. 329.

Papadopulos, S. S., Bredehoeft, J. D., Cooper, Jr., H. H., 1973, "On the Analysis of Slug Test Data, " Water Resources Research, Vol. 9, No. 4.

Theis, C. V., 1935, "The Relation Between the Lowering of the Piezometric Surface and the Rate and Duration of Discharge of a Well Using Ground Water Storage," American Geophysical Union Transactions, Vol. 16, p. 519-524.

Todd, David K., 1980, Ground Water Hydrology, John Wiley & Sons, Second Edition.

U.S. Department of the Interior, 1981, Ground Water Manual, U.S. Government Printing Office, Washington, D.C.

Walton, William C., 1970, Groundwater Resource Evaluation, McGraw-Hill.



JE JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS

APPENDIX 1-1 EXAMPLE OF ANALYSIS OF SLUG TEST

JE JACOBS ENGINEERING GROUP INC. SECTION Tech Memo 1 ALBUQUERQUE OPERATIONS PAGE 17 OF 28

APPENDIX 1-1

EXAMPLE OF ANALYSIS OF SLUG TEST AFTER COOPER, BREDEHOEFT AND PAPADOPULOS (1967)

SLUG TEST MW-20

From Recorder Chart

Time (sec)	x 0.556	Head (ft)	H/Ho
30	2.7	1.50	0.94
60	2.5	1.39	0.87
120	2.2	1.22	0.76
180	1.9	1.06	0.66
330	1.4	0.78	0.49
480	1.0	0.56	0.35
630	0.7	0.39	0.24

From 2-Cycle Semi-Log Plot

H_ = 1.60 ft

From 4-Cycle Semi-Log Plot at Match Point ($\propto = 10^{-5}$ Curve) Tt = 1.23 H/H₀ = 0.66 t = 180 sec $T = \frac{1.23(0.125)^2}{180} = 1.07 \times 10^{-4} \text{ ft}^2/\text{sec}$ $K = \frac{T}{4.00}(30.48) = 8.14 \times 10^{-4} \text{ cm/sec}$ (using screen length)

	 (205)	3NI1					
						Page 19	0
7							2.0
							FO X
C-L XIONJO							×
ing in the second secon							
					en portantes en constantes en		0'1
	02 MM		Grand A.E.	615			Ш

		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
		001

- - -






JE JACOBS ENGINEERING GROUP INC. SECTION _______ 16.1.5 ALBUQUERQUE OPERATIONS PAGE ______ OF _____

APPENDIX 1-2 EXAMPLE OF PRESSURE TRANSDUCER CHART AND ANALYSIS OF SLUG TEST





JEG-AL-CA-21(8/87)

JE JACOBS ENGINEERING GROUP INC. SECTION 16.1.5 ALBUQUERQUE OPERATIONS PAGE 25 OF 28

APPENDIX 1-2 (continued)

EXAMPLE OF ANALYSIS OF SLUG TEST DATA ANALYSIS FOR AN UNCONFINED ADUIFER After Bouwer and Rice (1976)

4) Evaluate the Hydraulic Conductivity, K

$$K = \frac{r_c^2 \ln\left(\frac{r_e}{r_w}\right)}{2 Lt} \ln \frac{y_o}{y_t}$$

=
$$3.4 \times 10^{-6}$$
 ft/sec = 1.0×10^{-4} cm/sec = 2.2 gpd/ft²

5) Evaluate the Transmissivity, T

T = K x D = 61 gpd/ft



210

1

JE JACOBS ENGINEERING GROUP INC. SECTION _______ 16.1.5 ALBUQUERQUE OPERATIONS PAGE ______ 26___OF _____28

APPENDIX 1-3

EXAMPLE OF ANALYSIS OF AQUIFER TEST DATA



APPENDIX 1-3 : EXAMPLE OF ANALYSIS OF AQUIFER TEST DATA

Section 16.1.5 Tech Memo 1 Page: 28 of 28 Attachment 1, Page 1 of 2

	JE -	EVENERALE OPERATION	u u		Pene Lat
	Lastrandia at a	AQUIFER	TEST		Page 1 01
SITE ID: START TEST: DATE TIME DATUM ELEV. (FT. MSL): STATIC WATER LEVEL (FT. MSL): INITIAL (AT 10) FINAL			LOCATION ID: END TEST: DATE TIME AVG. PUMPING RATE (GAL/MIN): MEASUREMENT METHOD: WELL DEPTH (FT.): CASING DIAMETER (IN.):		
NATER LEVEL MEASUREMENT METHOD: DISTANCE FROM PUMPED WELL, r (FT.): DATA ADJUSTED FOR REGIONAL WATER LEVEL CHANGES? FIELD REP(S): COMMENTS:					
ELAPSED TIME (MIN.)	DEPTH TO WATER * (FT.)	PUMPING RATE (GAL/MIN)	E	LAPSED TIME (MIN.)	DEPTH TO WATER* (FT.)
				99 42 40 49 49 49 49 49 49 49 49 49 49 49 49 49	
	second second and and provide the second second second			and the state of a state of the	internet internet of Commission Ind." (And
				nanta e dominio de de la como	

Section 16.1.5 fech Memo 1 Page: 28a of 28 Attachment 1, Page 2 of 2

		AQUIFER TEST		Page of_
	PUMP TEST		RECOVERY TEST	
ELAPSED TIME (MIN.)	DEPTH TO WATER * (FT.)	PUMPING RATE (GAL/MIN)	ELAPSED TIME (MIX.)	DEPTH TO WATER * (FT.)
Annual Schemen (Construction)		And the second sec	A REAL PROPERTY OF A READ REAL PROPERTY OF A REAL P	
Statements and an entropy of	and which a set of a state of the	different aller men kann men Appellin direk	MALE ADDRESS OF A DESCRIPTION OF	· · · · · · · · · · · · · · · · · · ·
And and a second second second second	A SUCCESSION OF A DESCRIPTION OF A DESCRIPANTA DESCRIPTION OF A DESCRIPTION OF A DESCRIPTION OF A DESCRIPTI	A designed of the second of the second distances		sero. Warney considerations
the statement of the second second		An and the or the research of the second second second	Construction in a solution interest	and a result of the set of a set of the set
Bertran ber i berta ber bertan ber	An order of the state of the st	and the state of a second se	BRANCE AND A DESCRIPTION OF A DESCRIPTIO	And the second second second second second second
	NUMBER OF STREET, STRE		And and the second second second	and a second sec
	and the second s	and a second	Building and the second s	ADDIVED BY COMPANY ADDIVIDUAL
			And Address of Concession, Name of Concession, Name	and the other designs of the second
An example of the owner designst	and the second second second second second		and the second s	
	and the same service state and the same state			
	Party or interest of the second se			
States and an other states and the states of				
-	Mandred Andread Mandred Stream of Stream Orac 2000 (100-1000) and a	Manual and a state of the second s		
-	Resident and Manual Street in a subdification	and the second		And all the other states in the second
Production of Province Advances in	errors and end of a new constructions.			distance and the second states of the second states of the
-	and a train sine and a second processing and a second second	And the American Conception and the American A	And and a second s	generation and the second second second
	and successive series as a consideration states	and a second second second second second second second	and the second second second second	And interest of the local division of the second se
	And the second difference of the second differ	Annual date and have been a substantial to be	And and the second second second	and the second second second second
-			And an Address of the Address	All reports for the second shift in a second second second
Management interest of the American			the subscript on the balance of the	Annual to be reading of the second second second second second
	All for the pairs of the same	Minerio	And the second second second second second	and control of the boot second the Proposition
-		and the second se		and the second sec
	March & March & Annual Statements of Annual Statements and the	Address of the local day we have the second of	And in case of the owner of the owner of the	Water and the first state of the state of the state of the
		Party and a second seco	AL MERICAN CONTRACTOR	and the second s
		Manufacture description of	And and a second diversity of the	and the second se
-	And inclusion, which are not included in the state of	and the second design of the s	and a real first out of the second se	
-	Party and the state of the other state of the state of th	Manual and works of the owner of the data in the owner of the		
	Married	Real Providence in the second second		
	Ministerio Contra anticipate a destante contra en	All the second se	Benefit and a second second second	
-		and we descend a second se		And the According to A the Art Transportation
			and of a subscription of the subscription of t	
-	Antonio anteres anteres. Anteres	Birth Contract Contract Contract	And the second data and the second data and the	
-	The second s	Anter service to an entering the service designed as	Man Statistic Advances View with Spinster	
-	-	Bertretensteinen einen einen einen einen		
-	an and an an and an and a state of the second data was		and the second s	-
-	A first strategies are reliant, some a strategies data per a state of the		COMPANY AND ADDRESS OF TAXABLE PARTY.	-
			Address of the American State	
Management in a section of			the spectrum is a second second second	
And the distribution of the second second		Received and a contract of the second s	-	
		Name of Address of Street and Str		And and the second second second
-		Barren a characteristic de la construcción de la co	Approximation and a subscription of	-
		general mapping and an and state from		
-		And Designation and the second state		
-	Research - Transmission - Constraints			
-	No house or the relative second second second	And and an operation of the second		
	-	And a second	And in case of the local division of the loc	

JEG-AL-ENG-5 (6/87)

* BELOW DATUM



JACOBS ENGINEERING GROUP INC.

AL PLICIE ODERATIONS MANUAL	SECTION 16.1.6 DATE 9-12-90	
ALBOQUERQUE OPERATIONS MANUAL		
SOIL-WATER SAMPLER INSTALLATION	EFFECTIVE 09/25/90	
	SUPERSEDES 06/22/89	
AND SAMPLE COLLECTION (SUCTION LYSIMETERS)	PAGE 1 OF 11	

1.0 PURPOSE

To provide a procedure for the installation and documentation of pressurevacuum soil water samplers and for the collection of water samples.

2.0 SCOPE

This procedure section will outline the steps for the installation of a pressure-vacuum soil water sampler. Also, procedures for conducting field measurements, filtration, preservation, and shipment of samples are presented.

10

- 3.0 EQUIPMENT AND SUPPLIES
 - 3.1 Porous ceramic cup
 - 3.2 Body tube
 - 3.3 Nail or similar object
 - 3.4 Neoprene plug
 - 3.5 Clamp ring
 - 3.6 Pinch clamps
 - 3.7 Polyethylene tubing: 1/4" diameter
 - 3.8 Neoprene tubing: 2/16" i.d. x 1/8" wail
 - 3.9 Distilled water
 - 3.10 Vacuum grease
 - 3.11 Tamping rod: PVC, 1" i.d. x 5' (several 5-foot sections, threaded) or a tremie pipe
 - 3.12 PVC pipe: for casing purposes, 6" diameter, should be Schedule 80 type
 - 3.13 PVC pipe cap: for above
 - 3.14 Hollow stem auger (HSA): with nominal 6.25" O.D., and 3.25 I.D. drill auger

JACOBS ENGINEERING GROUP INC. SECTION _ 16.1.6 ALBUQUERQUE OPERATIONS

PAGE_____ OF___11

- 3.15 Air rotary drill
- 3.16 Split spoon: 3" diameter (o.d.)
- 3.17 Vacuum-pressure pump, hand pump
- 3.18 Sample bottles
- 3.19 pH paper
- 3.20 Filter apparatus
- 3.21 Filters, 0.45 u membrane
- 3.22 Syringe filter with filters, disposable
- 3.23 Kim-wipes or equivalent
- 3.24 Rubber gloves, disposable
- 3.25 pH meter and electrode; with temperature compensator and with slope adjustment
- Alkalinity test kit 3.26
- 3.27 Thermometer
- 3.28 Silica flour: minus 200-mesh silica flour, dry. See Figure 2.
- 3.29 Silica sand, quartz: ~50-mesh, clean, well-graded
- 3.30 Bentonite, powdered
- 3.31 Native soil: from sampler site location (if site is uncontaminated)
- Wooden dowels: 3/16" diameter (o.d) x 2" long 3.32
- 3.33 Posts: wood (4" x 4" x ~6'), steel (~4" diameter x ~6'), or steel tee-bar (~6'). See 16.1.1 (3.7.7).
- 3.34 (Optional) cement for posts
- 3.35 Barbed wire
- Flagged material 3.36

4.0 CHEMICALS AND REAGENTS

NOTE: Whenever required by the procedure, JEG purchases and uses prepared, calibrated, and certified solutions from chemical supply firms. The following are listed in the event certified solutions are not readily on



JACOBS ENGINEERING GROUP INC. SECTION __16.1.6 ALBUQUERQUE OPERATIONS

PAGE ____ OF ____

hand or are used whenever procedure parameters do not require certified solutions.

4.1 Nitric Acid: HNO_x, ~70%, concentrated 16 M (~16 N)

5.0 PROCEDURE: GENERAL COMMENTS

The number of required soil-water samplers (samplers) shall be determined by the Site Hydrologist (SH) and approved by the Task Manager of the Hydrological Services and Technical Data Review Committee (TDRC). The SH shall assign the area and depth, up to a maximum of 50 feet, where the samplers are to be installed. Samplers should not be placed within five (5) feet of a tensiometer. No recommended method for determining station locations can be given that is applicable to all conditions. Each specific site requires preliminary investigation, interpretation of early results, adjustments in frequency, and location of tensiometer stations to meet the unique requirements of each site. The number of installations to be made would depend upon cost, the variability encountered, and the desired precision. Standard sampling analysis which incorporates these variables may be used to establish the sampling frequency (see References 14.1 and 14.2). Fencing around the site(s) may be necessary due to the potential for damage from intrusion by people, wildlife, livestock, or mechanical equipment. The surface area above the samplers should not be covered in a manner that would interfere with the normal percolation of soil moisture down to the depth of the sampler. The following guidelines are provided for drilling holes, installing the suction water samplers, collection of samples, and shipment of samples to the analytical laboratory.

6.0 PROCEDURE: Assembly of Soil Water Sampler

- 6.1 Before drilling, immerse the porous ceramic cup of the sampler in distilled water for approximately 1 hour or until the tip is saturated. To ensure that the tip is saturated, immerse in a bucket of distilled water, and put suction on the sampler. Proper operation of the sampler can also be observed at this time. During drilling, leave the sampler immersed in a bucket of distilled water to ensure that the tip does not dry out. Prior to installation, disassemble the sampler and remove the distilled water that has entered the sampler body. The sampler should be free of all distilled water and foreign objects prior to installation.
- 6.2 Once the hole has been drilled, determine the length of access tubes (i.e., 1/4" i.d. polyethylene tubing). Two lengths/colors are required. See Figure 1.
- 6.3 Push the discharge tube through the neoprene plug so that the end of the tubing reaches to within 1/8" of the bottom of the porous ceramic cup.

ALBUQUERQUE OPERATIONS PAGE _4_ OF _11

- 6.4 The pressure vacuum tube should protrude approximately 1" through the neoprene plug.
- 6.5 Leave a sufficient length of tubing sticking above the neoprene plug so that the access tubes are about 6" above ground level after installation of the soil water sampler. Cut the discharge tube approximately 6" longer than the pressure-vacuum tube to identify the different tubes, or use different colored tubing for discharge and vacuum lines.
- 6.6 Slip 6" of neoprene tubing (2/16" i.d. x 1/8" wall) over the ends of each access tube.
- 6.7 After installing the access tubes, coat the sides of the neoprene plug with vacuum grease, and insert the plug into the body tube.
- 6.8 Make sure the plug is held in place with the clamp ring by tightening the clamp with a nail or similar object. Tighten the ring until it meets the body tube. See Figure 1.

7.0 PROCEDURE: Installation of Soil Water Sampler

Depending on subsurface conditions, drilling shall be performed with either a Hollow Stem Auger (HSA) or an air rotary drill. An alternate drilling method can be used that does not involve the use of drilling mud or that does not excessively disturb soils. Holes shall be drilled using a fourinch minimum diameter bit. All drilling shall be accomplished without the addition of drilling mud or water. If the site is uncontaminated, native soils extracted from the drill hole should be removed, saved, and used as backfill.

7.1 Hollow Stem Auger (HSA) Drilling

Perform drilling with a HSA and as provided by a standard soilsampling drill rig. HSA should be used in unconsolidated granular soils where there is a potential for borehole-wall collapse and cave-in.

- 7.1.1 Terminate the augering about one foot above the desired depth of instrument installation. Do no withdriw auger (HSA) string at this time. Stockpile the soil removed from the borehole during drilling according to depth (i.e., soil from the 5-10' interval should be kept separate from the 10-15' interval) and returned to the borehole at the same depth from which it was removed. Place all stockpiled soil in plastic sheets or tarps to avoid surface contamination of the soil cuttings.
- 7.1.2 Use a split-spoon (nominal 3" o.d.) to knock the plug out of the bottom of the auger and to collect the final foot of

ALBUQUERQUE OPERATIONS GROUP INC. SECTION 16.1.6

material. NOTE: if site is uncontaminated, save soil for 7.3.5 and use for backfill. See TR.

- 7.1.3 After completing the borehole, drop approximately one to two pounds of dry silica flour down such a manner that most of the silica hits the bottom of the hole with considerable force. Wet the tip of the saturated ceramic cup with distilled water and rub tip with dry silica flour until a silica cake has formed on the tip.
- 7.1.4 With the silica flour patted onto the ceramic tip, lower the sampler through the hollow stem (while the auger is in the ground) to the bottom of the borehole so that it rests directly on the native soil. Push lightly on the top of the sampler with the tamping rod to ensure good contact between the ceramic cup, silica flour, and native soil.
- 7.1.5 Backfill the borehole to the top of the sampler (about 2.5') with the stockpiled native soil retrieved from the appropriate depth. Tamp soil in place using the tamping rod.
- 7.1.6 Pull a five (5) foot section of auger from the nole using a sand line and chain. Use the tamping rod to hold the instrument in place while pulling out the auger.
- 7.1.7 Once the five (5) foot section of auger has been removed. backfill the borehole with stockpiled material from the appropriate depth (i.e., from the bottom of the borehole to the face of the auger bit). Perform backfilling in 6-inch lifts, as a maximum, followed by compaction with a tamping rod. After sufficient soil has been added to the borehole so as to bring the level of the soil up to the face of the auger bit, remove an additional five (5) foot section of auger and backfill as stated previously. Continue backfilling and removing auger until a depth equal to 1/3 of the distance from the surface to the top of the sampler is reached. For example: given a 20-foot borehole with a 2-foot sampler, backfill to 12 feet from the surface. NOTE: Use the tamping rod to keep the instrument seated as the HSA auger string is retracted by sand line and chain
- 7.1.8 Install a one (1) foot bentonite seal using powered bentonite. Hydrate the bentonite by pouring in approximately one (1) gallon of distilled water. Wait approximately 30 minutes for bentonite to fully hydrate.
- 7.1.9 Once the bentonite has been given sufficient time to hydrate, continue pulling five (5) foot sections of auger and backfilling the borehole to a depth of 2 feet below land surface.

PAGE 6 OF

7.3 Final Installation

Once the soil water sampler has been assembled, the sampler is insta. Id into the borehole. The primary concern during installation is that the porous ceramic cup be in tight and intimate contact with the soil so that soil moisture can readily move from the soil to the soil water sampler.

- 7.3.1 JEG personnel shall check the soil water sampler for proper operation prior to final installation.
- 7.3.2 Replace any instrument which fails to operate properly in the same borehole or in another borehole drilled adjacent to the first.
- 7.3.3 Refer to operating instructions (Attachment 1) for the Model 1920 pressure-vacuum soil-water sampler for additional methods of installing the pressure-vacuum sampler in the event the method outlined in the SOP is not satisfactory.
- 7.3.4 As soon as practical after installation, the FTR shall use the pressure-vacuum pump to purge the sampler to obtain representative soil fluids. Continue purging until the slurry water is removed and the soil moisture can be sampled. If the SOP is used, purging will not be necessary because water was not added to the installation process.
- 7.3.5 Where the lysimeters are to be placed in a traffic area, sink a PVC protective casing in the bentonite-soil plug, thus surrounding the soil-water vacuum tubes. See 3.12.
- 7.3.6 Extend the casing down about two (2) feet into the plug and allow the casing to protrude above land surface about one (1) foot. Cover the top of the pipe with a slip-on PCV cap. Mark the lycimeter ID on the protective casing with an indelible ink pen.
- 7.3.7 Protect the soil water sampler from possible damage by vehicular traffic or grazing livestock whenever appropriate. Install 3-4 wood, steel, or tee-bar driven steel posts at a 4-foot radius around each PCV casing. Flagging should be attached to a post to visually mark the location. If livestock present a threat to the sampler, three strands of barbed wire should be affixed to the posts. NOTE: See Section 16.1.1 entitled "Monitor Well Installation" (Step 3.7.7) for details on installation of posts.

ALBUQUERQUE OPERATIONS PAGE_7_OF_11

8.0 PROCEDURE: Collection of Soil-Water Samples

A vacuum applied within the soil water sampler causes moisture to move from the soil through the porous framic cup into the sample bottles. Because the rate at which a samply collected is a function of the unsaturated hydraulic conductivity r is soil and the amount of vacuum which is created, the time required to collect a sample may vary. Generally, a vacuum between 50 to 80 centibars is sufficient to collect a sample in a few hours. However, under conditions of low conductivity or low soil moisture content, several days or weeks may be required to collect a sample. Because it is usually not possible to collect enough sample for a complete analysis, the order in which the constituents are analyzed must be prioritized. Table I lists the bottle ID and minimum volume required by the lab for analysis in order or priority. Constituents in each bottle are listed from high to low priority. The SH may establish a different set of analysis(es) priorities if concurrence is obtained from the Task Manager of Hydrological Services.

Bottled ID	Minimum Volume Requirement	Analytes
М	100 ml	Ba, Mo, As, Cd, Cr, Hg, Pb, Sb, Se, Al, Fe, Mn, Ag. Co, Cu, Ni, Sn, V, Zn, U, Be, Tl, Na, K, Mg, Ca, Sr
Ra-226 G α β Ra-228 N1 CN A	1000 ml 200 ml 1000 ml 100 ml 50 ml 100 ml	Ra-226 G α β Ra-228 NO ₃ , NO ₂ +NO ₃ , NH ₄ CN F, C1, SO ₄ , PO ₄ , SiO ₂ , B, TDS, Br
S Th ²³⁰ TOC	50 m ⁷ 500 m1 200 m1	S ² Th ²³⁰ TOC

Table I Listed in Decreasing Priority

8.1 To collect a sample,* close the pinch clamp on the discharge access tube and connect the vacuum port of the pressure-vacuum hand pump to

PAGE_8_0F_11

the pressure-vacuum access tube. Stroke the pump until a vacuum of approximately 65 centibars is created within the sampler. Check the readout on the pump gauge.

- 8.1.1 *Samples collected from, on, or near tailings piles may be hazardous. Wear disposal surgical-type gloves when collecting such samples. Care should also be taken to avoid splashing the sample liquid in the eyes or on the clothes.
- 8.2 Securely close the pinch clamp on the pressure-vacuum access tube to seal the sampler under vacuum. Remove the hand pump, if required for other uses. Allow the sampler to set for a period of time under vacuum (see Figure 5B).
- 8.3 To recover a soil-water sample, attach the pressure-vacuum access tube to the pressure port on the pump. Place the discharge access tube into a small, clean collection bottle and open both pinch clamps. Apply a few strokes on the hand pump to develop enough pressure within the sampler to force the collected water out of the sampler and into the collection bottle (see Figure 6).
- Collect subsequent samples by again creating a vacuum within the 8.4 sampler and repeating steps 8.1 through 8.3. Be sure to collect each sampler in a separate and clean container. NOTE: Always leave a vacuum on the soil water sampler when leaving a site.

9.0 PROCEDURE: Sample Filtration

Before performing field measurements, filter the samples through a 0.45 micron filter. If more than 500 ml of sample is available, use a large filtering system; otherwise, use a syringe/filter system. Filter samples collected from tailings with a syringe/filter. Follow the Standard Operating Procedures for water sampling documented in Section 16.1.1. Procedures specific to collecting soil-water samples are found below.

- 9.1 Large filtering system
 - Disassemble the filter apparatus and discard the old filter. Use distilled water to thoroughly rinse all 9.1.1 surfaces which come into contact with the sample.
 - 9.1.2 With clean hands, install a new filter. Handle filter only by its edges. Reassemble the apparatus. NOTE: If possible, use a pair of tweezers. Allow no dirt or dust to blow onto the cleaned apparatus or filter.
 - 9.1.3 Before filtering any samples, rinse filter and filter apparatus with a few ml of sample water. Filter and discard water.

JACOBS ENGINEERING GROUP INC. SECTION __ 16.1.6 ALBUQUERQUE OPERATIONS

PAGE 9 OF 11

- Fill the sample bottles. Allow no dirt or dust to blow 9.1.4 into the bottles or onto the bottle caps.
- 9.2 Syringe/filter system
 - 9.2.1 Insert sy: noe tip into sample solution and fill the syringe with sample by withdrawing the plunger.
 - Connect a new syringe filter to the tip of the syringe and 9.2.2 force sample through the filter into a clean sample bottle by depressing syringe plunger. NOTE: Sampler should practice with a syringe, filter system, and distilled water before filtering a sample.
 - Repeat above steps for each sample until entire sample has 9.2.3 been filtered.
 - 9.2.4 Discard used syringe and filter.
 - 9.2.5 Repeat above steeps with a new syringe', filter system, and sample bottle for each sample until all of the samples have been fi'tered.
- 10.0 PROCEDURE: Field Management of Samples for Temperature, pH, Specific Conductivity, Alkalinity, and Total Acidity

Record all field measurements and cements on Attachment 2 entitled, "soil Water Sampler Data Record." Complete all lines on the field form. If some steps were not taken or were not applicable, so indicate in the appropriate space. Complete the forms with nonwater soluble ink (felt-tip type pens or pencils are not acceptable). If any procedure was not performed as prescribed, state the reason on the field form. Once samples have been collected, perform measurements for temperature, pH, and alkalinity with procedures documented in Section 16.1.10. However, since some sample loss will occur during testing, perform measurements only if a 500-ml or more sample is collected. Also, omit alkalinity tests unless the sample pH is greater than 4.5. As always, exercise care to protect the samples from cross-contamination.

11.0 PROCEDURE: Sample Preservation

- 11.1 If there is less than 250 ml of sample, check sample pH, and record pH on Attachment 2. NOTE: If soil-water sample pH is less than 2.0 without being acidified, do not add any acid. Record this on Attachment 2.
 - 11 1 1 If the sample pH is 5 or more, acidify the sample with nitric acid (HNO3 in sufficient volume (-0.15 ml) to lower the sample pH to 2.0. Check sample pH. NOTE: Unless directed otherwise by TR/FTR or SH, use HNOz.

à

JACOBS ENGINEERING GROUP INC.

SECTION _______ 16.1.6

- 11.1.2 If micropipets are not available, add HNO₃ drop by drop. Mix sample solution well before checking sample pH.
- 11.1.3 Record amount of HNO₃ added to the sample on Attachment 2.
- 11.2 If there is more than 250 ml of sample, put the next 150 ml into a bottle unacidified. Record on Attachment 2.

12.0 PROCEDURE: Shipment of Collected Sample

- 12.1 Or ce HNO3 or other appropriate acid has been added, seal the sample bottles with electrical tape.
- 12.2 Compare information on sample containers with Attachment 2 to make sure that identification (e.g., sampling date, sample number, site number and location, date, analysis(es), etc.) are correct, sample containers are properly labeled, and that Attachment 2 has been filled with required field data and notes.
- 12.3 Separate samples into sample lots that consist of at least nine (9) samples, and one (1) known solution.
 - 12.3.1 Include any samples classified as "blank" or "spike" samples.
 - 12.3.2 Fill out chain-of-custody and other required forms with information from Attachment 2 or other field notes.
- 12.4 Place sample(s), chain-of-custody forms, sample identification forms, etc., into an ice chest containing "blue ice" or its equivalent and seal with appropriate tape.
- 12.5 Affix laboratory address and JEG return address. Ship samples by fastest mode with appropriate carrier.
- 12.6 Notify laboratory of time/date of shipment of sample(s), carrier, and expected time/date of shipment arrival at the laboratory.
- 12.7 Perform close out procedures.
 - 12.7.1 Cleanup field equipment and instruments.
 - 12.7.2 Replace expended time.
 - 12.7.3 kepair and report broken equipment. Replace equipment if necessary.
- 12.8 Submit completed and xeroxed copies of Attachment 2 and other field forms to the Site Manager, TR/FTR, and Data Manager.

JACOBS ENGINEERING GROUP INC. SECTION __________ ALBUQUERQUE OPERATIONS

PAGE_____OF___11

12.8.1 Inform SH by memorandum of completion of soil-water sampler installation and/or sample collection with shipment of sample(s) to the laboratory.

13.0 ATTACHMENIS

- 13.1 Figure 1 - Description of soil-water samplers
- 13.2 Figure 2A-C - View of installation of soil-water sampler
- Figure 3A-B View of installed soil-water sampler 13.3
- 13.4 Figure 4 - Assembly of soil-water sampler
- Figure 5A, B Procedure for sample collection 13.5
- 13.6 Figure 6 - Procedure for sample removal
- 13.7 Attachment 1 Operating instructions for the Model 1920 pressurevacuum soil-water sampler
- 13.8 Attachment 2 Soil-Water Sampler Data Record

14.0 REFERENCES

- Allmaras, R.R, 1955, in C. A. Black, ed., "Methods of Soil Analysis," Agronomy Monograph No. 9, Part 1, American Society of 14.1 Agronomy, Madison, Wisconsin, pp. 24-42.
- Krumbein, W.C. and F.A. Graybill, 1965. An Introduction to 14.2 Statistical Models in Geology, McGraw-Hill, New York, New York, pp. 171-181.
- 14.3 Soil Moisture Equipment Corp., P.O. 3002J, Santa Barbara, CA 93105.

15.0 PROCEDURE REVIEW AND APPROVAL

Johe. Prepared By: Reviewed By: thu h Task Manager ander 9/24/90 Reviewed By: Quelity Assurance Manager Maen cours Approved By: Project Manager





.

FIGURE 2A



4



VIEW OF INSTALLATION OF SOIL WATER SAMPLER







Attachment 1

OPERATING INSTRUCTIONS for the

Model 1920

PRESSURE-VACUUM SOIL WATER SAMPLER



SITE LOCATION

The Model 1920 Pressure-Vacuum Soil Water Sampler can be installed in any location. at any depth up to a maximum of 50 ft. The access tubes from the sampler may be terminated directly above the sampler Installation or, if conditions require, the access tubes may be laid in a trench and carried to a remote location, terminated above the soll surface. The access tubes should be located so that they will be free from damage by mechanical equipment and by livestock. The surface area directly above the sampler should not be covered in any manner that would interfere with the normal percolation of soll moisture down to the depth of the sampler. Once the depth and location for the pressure-vacuum soil water sampler has been decided, the length of the access tubes should be determined so that they can be cut and mounted into the soil water sampler.

INSTALLATION OF ACCESS TUBES IN THE SAMPLER

(Fig.1) The access tubes are normally made from 1/4" diameter copper or polyethylene tubing. When installing the tubes, one tube should be pushed through the neoprene plug so that the end of the tubing reaches almost down to the bottom of the porous ceramic cup. This "discharge" access tube should be marked at the other end in some fashion as to identify it. The other "pressure-vacuum" access tube should be inserted into the neoprene plug so that it extends through the plug perhaps one inch.

(Fig.2) After the tubes are installed, tighten the clamp ring with a nail or similar object inserted through the holes provided in the clamp ring. Tighten only until it meets the body tube. See note on page 5 If potting is desired.





BACK FILL

SLURRY

(FIG.3)

CORING THE HOLE

in rock-free uniform soils at shallow depths, use a 2" screw or bucket auger for coring the hole.

If the soil is rocky, a 4" auger should be used. The soil should then be sifted through a 1/4" mesh screen to free it of pebbles and rocks. This will provide a reasonably uniform backfill soil for filling in around the soil water sampler. The Model 230 Series Soil Augers can be used for this purpose.

INSTALLATION OF PRESSURE - VACUUM SOIL WATER SAMPLER USING A SOIL SLURRY

(Fig.1) After the hole has been cored, mix a substantial quantity of soll from the bottom of the hole with water to make a slurry which has a consistency of cement mortar. This slurry is then poured down to the bottom of the cored hole to insure a good soll contact with the porcus ceramic cup.

(Fig.2) Immediately after the slurry has been poured, push the soil water sampler down into the hole so that the porcus ceramic cup is completely embedded in the soil slurry.

(Fig.3) Backfill the remaining area around the soll water sampler, tamping lightly with a small diameter rod. Backfill the remainder of the hole, tamping soll firmly, to prevent surface water from running down the cored hole. Packfill hole with native soll free of pebbles and rocks.





ADDITIONAL METHODS OF INSTALLING THE PRESSURE + VACUUM SAMPLER

(Fig.1) Core hole to desired depth, insert soil water sampler and backfill the hole with native soil, tamping continuously to insure good soil contact with the porous ceramic cup and complete sealing of the cored hole.

(Fig.2) Core hole to desired depth, pour in a small quantity of crushed 200 mesh pure silica-sand of almost taicum powder consistency (commercially available under trade names of Super-Sil and Silica Flour). Insert soil water sampler and pour another layer of the 200 mesh silicasund at least six inches deep around cup of the soil water sampler. Backfill the hole with soil free of pebbles and rocks, tamping continuously with a long metal rod to insure against surface water channeling down between the soil and the body tube of the sampler.

(Fig.3) Core hole to desired depth, pour in a small quantity of wet bentonite clay. This will isolate the sampler from the soil below. Pour in a small quantity of 200 mesh silicasand and insert soll water sampler. Pour another layer of 200 mesh silica-sand at least six inches deep around the cup of the soll water sampler. Backfill with native soll to a level just above the soli water sampler and again add a small quantity of bentonite as a plug, to further isolate the soll water sampler and guard against possible channeling of water down the hole. Backfill the remainder of the hole slowly, tamping continuously with a long metal rod. Again backfill should be of native soil free of pebbles and rocks.

There are other methods of installing the soll water samplers that may be used, largely dictated by the type of sollyou are concerned with and the tools available. The primary concern in any method of installation is that the porous ceramic cup of the sampler be in tight, intimate contact with the soll, so that soll moisture can move readily from the pores of the soll through the pores in the ceramic cup and into the soll water sampler.



COLLECTING SOIL WATER SAMPLE

After the pressure-vacuum soil water sampler has been installed in the field, the accessory items are added as shown on page 6.

(Fig.1) To collect a sample, the pinch clamp on the discharge access tube is closed and the vacuum port of the pressure-vacuum hand pump is connected to the pressurevacuum access tube. The pump is then stroked until a vacuum of perhaps 60 centibars (18" of mercury) is created within the sampler, as read out on the gauge connected to the pump.

(Fig.2) The pinch clamp on the pressurevacuum access tube is then closed securely to seal the sampler under vacuum. The hand pump can then be removed for other uses. The sampler is allowed to set for a period of time under vacuum.

The vacuum within the sampler causes the moisture to move from the soil, through the porous ceramic cup, and into the sampler. The rate at which the soil solution will collect within the sampler depends on the capillary conductivity of the soil, the soil suction value within the soil (as measured with tensiometers), and the amount of vacuum that has been created within the sampler. In moist soils of good conductivity, at field capacity (10 to 30 centibars of soil suction as read on a tensiometer) substantial soil water samples can be collected within a few hours. Under more difficult conditions it may require several days to collect an adequate sample.

In general, vacuums of 50 to 85 centibers (15" to 25" of mercury) are normally applied to the soil water sampler. In very sandy solls it has been noted, however, that very high vacuums applied to the soil water sampler seem to result in slower rate of collection of the sample than lower applied vacuums. It is our feeling that in these coarse, sandy soils, the high vacuums within the sampler may deplete the moisture in the Immediate vicinity of the porous ceramic cup and hence reduce the capillary conductivity, which creates a barrier to the flow of moisture to the cup under these circumstances. In loams and gravelly clay loams, users have reported collection of 300 to 500 ml of solution over a period of a day with applied vacuum of 15" of mercury (50 centibars) when soils are at field capacity. On waste water disposal sites, some users have obtained up to 1500 m of sample within 24 hours after cessation of irrigation with 1" to 2" of waste water on sandy or clay loam soil.



(Fig.3) To recover a soil water sample, attach the pressure-vacuum access tube to the pressure port on the pump. Place the discharge access tube in a small collection bottle and open both pinch clamps. Apply a few strokes on the hand pump to develop enough pressure within the sampler to force the collected water out of the sampler and into the collection bottle.

Subsequent samples are collected by again creating a vacuum within the sampler and following the steps as outlined above.

MAINTENANCE

There are no maintenance requirements for the pressure-vacuum soil water samplers, other than protecting the access tubes from damage. Tube ends should be covered or plugged to prevent debris from entering the tubes and later contaminating the sample.

Freezing conditions will not damage the samplers. The samplers are normally left permanently in place throughout the year.

INSTRUCTIONS FOR POTTING

With our current design, we do not feel that potting of the neoprene plug is necessary to maintain a permanent pressure-vacuum seal; however, if you desire the design has been arranged so that these elements can be potted with an epoxy resin.

- (1) INSTALL ACCESS TUBES
- (2) INSTALL FIRST CLAMP WASHER
- (3) INSTALL NEOPRENE PLUG AND ADD EPOXY TO THE OUTSIDE DIAMETER
- (4) INSTALL SECOND CLAMP WASHER
- (5) SCREW CLAMP RING IN PLACE, AND POUR EPOXY INTO CAVITY TO COMPLETELY SEAL THE NEOPRENE PLUG AND ACCESS TUBES



-6-

ATTACHMENT 1 CONTINUED

Soilmoisture Equipment Corp.

About Our Soil Water Samplers

Our Soil Water Samplers, which are sometimes referred to as "suction lysimeters" or "lysimeters" are now in general use, and many thousands of these are currently installed in this country and other countries throughout the world. The soil water samplers had their origin back in 1961 when we cooperated with Dr. George H. Wagner at the University of Missouri to fabricate a porous ceramic cup for this purpose. The outgrowth of this work was the first commercial soil water sampler, which was assigned our Cat. No. 1900. Since that time, these samplers have been generally accepted as an ideal tool for securing soil water samples in situ for a great variety of soil moisture monitoring work.

The most extensive initial use of these samplers was made by Pennsylvania State University, largely under the direction of Dr. L. T. Kardos and others, on the Pennsylvania Waste Water Project. Modifications of the original Cat. No. 1900 Soil Water Sampler to produce a pressure-vacuum type unit was done by Richard R. Parizek and Burke E. Lane at the Pennsylvania State University, and reported on in the Journal of Hydrology. Since that time, we have made available commercially our Model 1920 Pressure-Vacuum Soil Water Sampler. Some of our soil water samplers have been in continuous use now for a period of ten years and are still yielding satisfactory soil moisture samples. The samplers find application not only in research work but for pollution control purposes, such as monitoring the moisture under sanitary landfills and under areas irrigated with wastewater. Various models of our Soil Water Samplers are available for installation at depths ranging from the surface to 300 ft.

Our Model 1900 Soil Water Sampler is designed for installation near the surface, at depths ranging from 6" to 6'. This type of sampler projects several inches above the soil surface after installation and is serviced at the exposed end. The Model 1900 Soil Water Sampler, along with the Model 1920 Pressure-Vacuum Soil Water Sampler, is pictured and described on the backpage of our six-page general brochure entitled, "A Full Range of Soilmoisture Extractors". The enclosed operating instructions for this unit will give you details of installation and operation and also spell out the accessory items required. All items are carried on our current price list.

Our Model 1920 Pressure-Vacuum Soil Water Sampler is designed to be completely buried in the soil. The overall length of this unit is approximately 24" and is serviced by two access tubes. This is the type of sampler frequently used in monitoring for polution, since the access tubes can be located at a distance from the sampler and in an area out of the way of surface traffic.



These samplers are normally installed at a minimum depth of 3 ft. and can be used at depths of up to 50 ft. It is possible to use these units at somewhat greater depths than 50 ft., perhaps as high as 70 ft.; however, at greater depths, a substantial amount of the collected sample is forced back into the soil, through the porous ceramic cup, by the high air pressure necessary to force the collected sample up to the surface. If you are concerned with extracting solutions at great depths, our Model 1940 Hi/Pressure-Vacuum Soil Water Sampler should be used.

The enclosed instruction information on the Model 1920 Pressure-Vacuum Soil Water Sampler will give you details of installation and operation. Page 6 of these instructions shows a cross section of the 1920 Pressure-Vacuum Soil Water Sampler installed in the field and indicates the specific accessory items required for its operation. All items are carried on our current price list.

If you are involved with sampling moisture at depths beyond the range of our Model 1920, we have available our Model 1940 Hi/Pressure-Vacuum Soil Water Sampler that can be used up to depths of 300 ft. The enclosed engineering schematic shows the operating features of this type of sampler. Basically, the unit incorporates two check valves which permit the flow of water from the soil into an upper chamber when vacuum is applied. When pressure is subsequently applied to remove the collected sample, the pressure is prevented from reaching the porous ceramic cup chamber by one of the check valves. With this type of arrangement, high pressures can be exerted to remove the collected sample from great depths without damage to the porous ceramic cup. Other than the features mentioned above, the operation of the Hi/Pressure-Vacuum Soil Water Sampler is similar to the Model 1920 Pressure-Vacuum Soil Water Sampler. Prices are carried on the current price list.

Soilmoisture Equipment Corp. P.O. Box 30025, Santa Barbara, Calif. 93105 U.S.A.





OPERATING INSTRUCTIONS for the

Model 1900

SOIL WATER SAMPLER



SITE LOCATION

The Modei 1900 Soil Water Sampler can be installed in any location. The sampler can be installed in well drained soil or in areas where the water table is above the sampling depth. The surface area directly above the sampler should not be covered in any manner that would interfere with the normal percolation of soil moisture down to the depth of the sampler.

The samplers are normally installed vertically in the soil. However, they can be installed at an angle if this is necessary to reach some otherwise inaccessible point.

The samplers are available in various stock lengths for installation at depths up to 6 ft. Extra length samplers can be provided on special order, if this is necessary. However, for depths greater than 6 ft. It is normally less expensive to use the Model 1920 Pressure-Vacuum Soil Water Sampler.

The Model 1900 Soil Water Sampler has been designed so that the body tube of the sampler projects 2" above the soil surface when the sampler is installed to the proper depth, as shown in the figure to the left.

-1-






CORING THE HOLE

In rock-free uniform soils at shallow depths, use a 2" screw or bucket auger for coring the hole.

If the soll is rocky, a 4" auger should be used. The soll should then be sifted through a 1/4" mc screen to free it of pebbles and rocks. This will provide a reasonably uniform backfill soll for filling in around the soll water sampler. The Model 230 Series Soll Augers can be used for this purpose.

INSTALLATION OF SOIL WATER SAMPLER USING A SOIL SLURRY

(Fig. 1) After the hole has been cored, mix a substantial quantity of soil from the bottom of the hole with water to make a slurry which has a consistency of cement mortar. This slurry is then poured down to the bottom of the corad hole to insure a good soil contact with the porous ceramic cup.

(Fig. 2) Immediately after the slurry has been poured, push the soil water sampler down into the hole so that the porcus ceramic cup is completely embedded in the soil slurry.

(Fig. 3) Backfill the remaining area around the soil water sampler, tamping soil firmly, to prevent surface water from running down the cored hole. Backfill hole with rative soil free of pebbles and rocks.



ADDITIONAL METHODS OF INSTALLING THE SOIL WATER SAMPLER

(Fig.1) Core hole to desired depth, insert soll water sampler and backfill the hole with native soll, tamping continuously to insure good soll contact with the porcus ceramic cup and complete sealing of the cored hole.

(Fig.2) Core hole to desired depth, pour in a small quantity of crushed 200 mesh pure silica-sand of almost talcum powder consistency (commercially available under trade names of Super-Sil and Silica Flour). Insert soil water sampler and pour another layer of the 200 mesh silicasand at least six inches deep around cup of the soil water sampler. Backfill the hole with soil free of pebbles and rocks, tamping continuously with a long metal rod to insure against surface water channeling down between the soil and the body tube of the sampler.

(Fig. 3) Core hole to desired depth, pour in a small quantity of wet benton te clay. This will isolate the sampler from the soll below. Pour in a small quantity of 200 mesh silica-sand and insert soll water sampler. Pour another layer of 200 mesh silica-sand at least six inches deep around the cup of the soll water sampler. Again, add a small quantity of bentonite as a plug to further isolate the ceramic cup and guard against possible channeling of water down the hole. Backfill the remainder of the hole with native soll free of pebbles and rocks, tamping continuously with a long metal rod.

There are other methods of installing the soil water sampler that may be used, largely dictated by the type of soil you are concerned with and the tools available. The primary concern in any method of installation is that the porous cup of the sampler be in tight, intimate contact with the soil, so that soil moisture can move readily from the pores of the soil through the pores in the ceramic cup and into the soil water sampler.



COLLECTING SOIL WATER SAMPLE

After the soil water sampler has been installed in the field, the accessory items as shown on page 6 are used for collecting soil water sample. (Fig.1) To collect a sample, the pinch clamp on the stopper assembly is opened. The serrated tube fitting on the end of the vacuum dial gauge adapter is then inserted into the neoprene tube of the stopper assembly. The vacuum hand pump is then stroked until a vacuum of perhaps 60 centibars (18" of mercury) is created within the sampler, as read out on the vacuum dial gauge.

(Fig.2) The pinch clamp is then closed securely to seal the sampler under vacuum. The hand pump can then be removed for other uses. The sampler is allowed to set for a period of time under vacuum.

The vacuum within the sampler causes the moisture to move from the soil, through the porous ceramic cup, and into the sampler. The rate at which the soil solution will collect within the sampler depends on the capillary conductivity of the soll, the soll suction value within the soil (as measured with tensiometers), and the amount of vacuum that has been created within the sampler. In moist soils of good conductivity, at field capacity (10 to 30 centibars of soil suction as read on a tensiometer), substantial soll water samples can be collected within a few hours. Under more difficult conditions it may require several days to collect an adequate sample.

In general, vacuums of 50 to 85 centibars (15" to 25" of mercury) are normally applied to the soll water sampler. In very sandy soils it has been noted, however, that very high vacuums applied to the soll water sampler seem to result in slower rate of collection of the sample than lower applied vacuums. It is our feeling that in these coarse, sandy solls, the high vacuum within the sampler may deplete the moisture In the immediate vicinity of the porous ceramic cup and hence reduce the capillary conductivity, which creates a barrier to the flow of moisture to the cup under these circumstances. In loams and gravelly clay loams, users have reported collection of 300 to 500 ml of solution over a period of a day with applied vacuum of 15" of mercury (50 centibars) when soils are at field capacity. On waste water disposal sites,



some users have obtained up to 1500 ml of sample within 24 hours after cessation of irrigation with 1" to 2" waste water on sandy or clay loam soil.

(Fig.3) To remove the soll water sample from the sampler, a simple assembly is usually made up consisting of a small diameter (3/32" O.D. or less) plastic tube, a two-hole rubber stopper, a flask or bottle, as shown.

The pinch clamp on the sampler is opened and the small diameter plastic tube is inserted into the end of the neoprene tube on the stopper assembly and pushed down until it reaches the bottom of the sampler.

(Fig.4) The vacuum hand pump is then connected to the other hole in the stopper. Stroking the hand pump creates a vacuum within the bottle or flask which in turn sucks the sample up from the sampler and into the collection bottle or firsk.

If it is more convenient, the stopper assembly can be removed from the sampler so that the collected cample can be removed with a pipette couther cons. However, repeated removal and repide ment of the rubber stopper assembly can disturb the seal between the soil and the body tube of the sampler, particularly on shallow units.

Subsequent samples are collected by again creating a vacuum within the sampler and following the steps as outlined above.

MAINTENANCE

There are no maintenance requirements for the soil water samplers other than protecting the exposed end of the body tube and the stopper assembly from physical damage. The end of the neoprene tube on the stopper assembly should be covered or plugged to prevent debris from entering the tube and later contaminating the sample.

Freezing conditions will not damage the samplers. The samplers are normally left permanently in place throughout the year.



-6-



JE JACOBS ENGINEERING GROUP INC.

SOIL WATER SAMPLING RECORD

FINAL FIELD VALUES:
pH (S.U.): Ec (mhos/om) TEMP, (°C): ALKANUNITY (mgl/ CaCO ₃): LOCATION DESCRIPTION

	CONTAINER SIZE	NONACIDIFIED	ACIDIFIED	VOL ACD (mi)
NUMBER OF	ONE-LITER	Britisher & Lawrence and Britisher and	NOT THE OWNER OF THE OWNER OF THE OWNER OF	
CONTAINERS	500 ml			
COLLECTED :	250 ml	-		
SPECIFY OTHERS :				
				-

RELD REPRESENTATIVE .

ATTACHMENT 2

.

COMMENTS :

Q ****

JEG-AL-ENG-NB (8/87)

AND AND AND AND



.

-

6

-

ROCKT HOUNTAIN FOREST KANGE EXPERIMENT STATION 2205 COLUNLIA SE ALBUQUERQUE, NEW MEXICO 87106 APPENDIX I EMERGENCY EVACUATION ROUTE



X - Fire Pull Box XX - Fire Extinguisher

EXHIBIT I

1 3

JE JACOBS ENGINEERING GROUP INC.

AL PLIQUEDOUE OPERATIONS MANUAL	SECTION 16.1.7
ALBOQUERQUE OFERATIONS MANOAL	DATE 06/09/89
	EFFECTIVE 12/04/89
INSTALLATION/SERVICING OF TENSIOMETERS AND	SUPERSEDES
ALASUREMENT OF SULL BALLA FULLING	PAGE 1 OF 10

1.0 PURPOSE

To provide a procedure for the installation and servicing of tensiometers so that field programs for the collection of tensiometer/pressure potential data are properly conducted and documented.

2.0 SCOPE

This procedure is applicable to the FAC and its subcontractors who have dutics and responsibilities for the collection of all tensiometer/ pressure potential data from tailings sites (both onsite and offsile) and at alternate disposal sites.

3.0 EQUIPMENT & SUPPLIES

- 3.1 Tensiometer
- 3.2 Tensiometer Pressure Gauge
- 3.3 Bourdon Tube Gauge
- 3.4 Distilled water
- 3.5 Tamping Rod: PVC, 1" i.d. x 5' (several 5' sections, threaded) OR a Tremie Pipe.
- 3.5 PVC Pipe: for casing purposes. ~6" diameter, should be Schedule 80 type.
- 3.7 PVC Pipe cap: for 3.6 above.
- 3.8 Hollow Stem Auger Drill Rig or
- 3.9 Air Rotary Drill Rig or
- 3.10 Head auger
- 3.11 Split Spoon: 2" diameter (o.d.)
- 3.12 Vacuum-Pressure Pump, Hand pump
- 3.13 Kim-wipes or equivalent
- 3.14 Bentonite, raw or granular.

- 3.16 Bentonite/soil seal: a 1:1 mixture of bentonite, clean native soil, and distilled water as specified by the SH.
- 3.17 Native Soil: from Sample Site Location only if site is uncontaminated.
- 3.18 Wooden dowels: 3/16" diameter (o.d.) x 2" long
- 3.19 Posts: wood (4" x 4" x \sim 5'), steel (\sim 4" diameter x \sim 5'), or Tee-bar steel driven. 3-4 posts per site. See Section 16.1.1 (3.7.7).
- 3.20 (Optional). Cement for posts.
- 3.21 Barbed wire, 100 ft. roll.
- 3.22 Flagging Material

4.0 PROCEDURE: Pre-field Preparation

The following steps shall be performed no less than ten (10) days before leaving for the field.

- 4.1 The Site Hydrogeologist (SH) shall determine the number of required tensiometers and locations and shall be approved by the Task Manager of Hydrological Services and the Technical Data Review Committee. NOTE: The tensiometer is an instrument which provides a direct measurement of the negative pressure potential or suction in soil. Tensiometers are used in field experiments where it is necessary to measure the gradient in pressure potential in order to determine the direction and rate of water movement in a soil profile.
- 4.2 The SH or the Field Technical Representative (FTR) shall check all equipment to ensure that a) there is a sufficient amount of equipment, b) they are in good condition, and c) they are working properly.
- 4.3 The FTR shall check the supply of distilled water to ensure that there is a sufficient quantity for the proposed field activities or that distilled water can be obtained near the site location before the tensiometers are to be installed.
- 4.4 For transportation to a site, the SH or FTR shall properly pack the tensiometers so that the dial gauges and ceramic tips are not damaged. Individual tensiometer pieces should be separately wrapped in paper and must be packaged so that they are secure inside the shipping boxes. Use special care so that ceramic tips do not touch grease or similar material that could clog the ceramic tip pores. Label all boxes with proper shipping and return addresses.

4.5 The FTR shall use Attachment 12.1 entitled "Tensiometer Equipment Checklist" to record the checkout and readiness of the equipment.

5.0 PROCEDURE: Principles for Field Placement of a Tensiometer

- 5.1 Location of the tensiometer stations and the number of tensiometers installed depends on the variability of the soil at the site, the nature of the topography, depth to the water table, and the depth at which water flow is to be studied. Locate stations in areas which represent the site's soil/water characteristics. The SH shall select tensiometer depths and locations (stations) based on data such as borenole geophysical data, core samples, and surface geophysical data.
- 5.2 Stations should be located in areas of either maximum past or potential contaminant sources. Each station must have tensiometers at multiple depths. Increased density of tensiometer installation will give more information about water movement within the soil profile. In most cases, install the tensiometers at a minimum of three depths above the capillary fringe region of the water table at increments of two feet or less. Monitoring should be intensified on those sites which exhibit a high potential for contributing contamination to the ground-water system.
- 5.3 Each specific site requires preliminary investigation, interpretation of early results, and may require adjustments in sampling frequency and number of tensiometer stations to ensure adequate characterization. The number of installations to be made would depend upon cost, the variability of the soil types and geochemical environments encountered, and the desired precision. Statistical analyses which incorporate these variables may be used as guidelines in establishing the sampling frequency. See References 13.1 and 13.2.
- 5.4 Protection of the tensiometers is essential. The FTR may need to construct a fenced area around each station to limit damage from intrusion by people or animals. See 3.7.7 of AOM, Section 16.1.1 (3.7.7) entitled "Well Installation".

6.0 PROCEDURE: Tensiometer Assembly and Preparation

6.1 Tensiometer Assembly

6.1.1 defore assembling the tensiometer, make sure all parts are clean and have not been damaged in shipping. Do not use any damaged equipment. NOTE: See Figure 1 for component parts of a tensiometer. JACOBS ENGINEERING GROUP INC. SECTION 16.1.7 ALBUQUERQUE OPERATIONS PAGE 4 OF 10

- 0.1.2 Screw the vacuum dial gauge into the threaded port located in the side of the body tube until the backup washer on the stem body touches the body tube. NOTE: Do not overtighten the dial gauge in the body tube as the "O" ring on the stem makes the vacuum seal.
- 6.1.3 Screw the ceramic tip into the body tube to the desired length as specified by the FTR or SH.
- 6.1.4 Before putting the tensiometer to use, vent the dial gauge by momentarily removing the vent screw in the center of the dial gauge cover. NOTE: This is done to adjust the dial gauge to local conditions because the gauges are usually set at sea-level atmospheric conditions.
- 6.2 Tensiometer Preparation
 - 5.2.1 Fill the tensioneter with distilled water and place the tensioneter body tube in an upright and vertical position until fluid completely saturates the ceramic tip and drips from the end of the tip for about five minutes.
 - 5.2.2 Once the tip is saturated, fill the tube completely with distilled water and pull a vacuum within the top of the tensiometer using the vacuum hand pump. See Figure 2. Alternatively, if the hand pump is not available, air can be adequately removed from the tensiometer by pushing the reservoir button down repeatedly, after the tensiometer and reservoir have been filled. See Figure 3.
 - 5.2.3 Screw the fill reservoir cap onto the top of the tube body.
 - 5.2.4 Peel the neoprene reservoir cover back and fill the reservoir ~3/4 full with distilled water. Replace the neoprene cover back over the fill reservoir cap and push the reservoir button down several times to remove air from the reservoir pump cylinder and any accumulated air in the tensiometer. See Figure 3.
 - 6.2.5 The vacuum/pumping operation should be repeated four or five times until no further air bubbles from the stem of the vacuum gauge.
 - 6.2.6 Because commercial gauge type tensiometers generally are not corrected for the length of the tensiometer, adjust the dial on the vacuum gauge to zero when the tip of the tensiometer is immersed in water. See Figure 4.
 - 5.2.7 If the tensiometer is not to be installed immediately, store the tensiometer with the ceramic tip immersed in distilled water or covered with a plastic bag to prevent evaporation.

JEG-AL-QA-21 (8/87)

ALBUQUERQUE OPERATIONS

PAGE_5_0F_10

7.0 PROCEDURE: Drill Tensiometer Site

Drilling can be accomplished using either a hollow-stem auger (HSA), air-rotary equipment or hand auger, depending on subsurface conditions. Drill holes with a two (2) inch minimum diameter bit. Perform all drilling without the addition of drilling mud or water.

HSA drilling as provided by a standard soil-sampling drill rig should be used in unconsolidated, granular soils where there is a potential for borehole-wall collapse and cave-in. Air rotary drilling should only be used for well-consolidated borehole soils, clays, or tight silts where the borehole wall can be expected to maintain its integrity. A hand auger may be used when drill=site accessibility is a problem for shallow (<25') installations.

- 7.1 Hollow-stem auger drilling
 - 7.1.1 Terminate the augering approximately one foot above the desired depth at which the tensiometer is to be installed. Leave the auger string in the ground. NOTE: Stockpile the soil cuttings on a clean plastic sheet in the order in which they were removed from the borehole and save the cuttings for 7.1.4.
 - 7.1.2 Use a split spoon to knock the plug out of the bottom of the auger and to collect the final foot of material. See 3.10.
 - 7.1.3 Install the tensiometer by lowering the instrument through the hollow stem auger until the tensiometer makes contact with the bottom of the borehole. Push down lightly on the tensiometer to insure that good contact is made between the ceramic tip and native soil.
 - 7.1.4 Remove one five-foot joint of auger using a sand line and cnain. Add ~6 inches of native soil to the borehole. NOTE: Use the stockpiled cuttings of 7.1.1 and add the cuttings to the borehole at the same depth from which they were removed so as to simulate as near as possible the original downhole conditions.
 - 7.1.5 Compact the backfill around the tensiometer using a 1-inch tamping rod. See 3.5. NOTE: Initial backfill lifts should be limited to 6-inches or less to insure that sufficient compaction of the soil around the instrument occurs.
 - 7.1.5 After ~4 feet of backfill has been added and compacted, place a one (1) foot thick bentonite seal in the borehole so that no vertical migration of water is possible. See 3.18 and 3.19. NOTE: Introduce the bentonite seal into the borehole by either pouring it down the borehole or by use of a tremie pipe.

ALBUQUERQUE OPERATIONS

JACOBS ENGINEERING GROUP INC. SECTION_ 16.1.7 PAGE 6 OF 10

- 7.1.7 Once the bentonite has been placed, pour ~5 gallons of distilled water down the borehole in order to hydrate the bentonite.
- 7.1.8 wait at least 30 minutes between the addition of the water and the addition of the next lift of native soil so that the bentonite has sufficient time to hydrate.
- 7.1.9 After at least 30 minutes has elapsed, remove another five (5) foot section of auger. Add bentonite/native soil mixture to the borehole and compact the bentonite/soil mixture with a one-inch tamping rod. See 3.15 and 3.5.
- 7.1.10 Continue backfilling and tamping until the borehole is filled to ground level.
- 7.2 Air rotary drilling
 - 7.2.1 Terminate the drilling at the desired depth of tensiometer installation. NOTE: Stockpile the soil cuttings on a clean plastic sheet in the order in which they were removed from the borehole and save the cuttings for 7.2.4.
 - 7.2.2 withdraw and remove all drill pipe and bit from the borehole.
 - 7.2.3 Install the tensiometer by lowering the instrument to the bottom of the borehole. Push down lightly on the tensiometer to insure that good contact is made between the ceramic tip and native soil.
 - 7.2.4 Add ~o inches of native soil to the borehole. NOTE: Use the stockpiled cuttings of 7.2.1 and add the cuttings to the borehole at the same depth from which they were removed so as to simulate as near as possible the original downhole conditions.
 - 7.2.5 Compact the backfill around the tensiometer using a 1-inch tamping rod. NOTE: Initial backfill lifts should be limited to ö-inches or less to insure that sufficient compaction of the soil around the instrument occurs.
 - 7.2.0 After ~4 feet of backfill has been added and compacted, place a one (1) foot thick bentonite seal in the borehole so that no vertical migration of water is possible. NOTE: Introduce the bentonite seal into the borehole by either pouring it down the borehole or by use of a tremie pipe.
 - 7.2.7 Once the bentonite has been placed, pour ~5 gallons of distilled water down the borehole in order to hydrate the bentonite.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.7 ALBUQUERQUE OPERATIONS

- FAGE 7 OF 10
- 7.2.8 Wait at least 30 minutes between the addition of the water and the addition of the next lift of native soil so that the bentonite has sufficient time to hydrate.
- 7.2.9 Add bentonite/native soil mixture to the borehole and compact the bentonite/soil mixture with a one-inch tamping rod. See 3.15 and 3.5.
- 7.2.10 Continue backfilling and tamping until the borehole is filled to ground level.
- 7.3 Hand Auger Drilling
 - 7.3.1 Terminate the drilling at the desired depth of tensiometer installation. NOTE: Stockpile the soil cuttings on a clean plastic sheet in the order in which they were removed from the borehole and save the cuttings for 7.2.4.
 - 7.3.2 Withdraw and remove the hand auger from the borehole.
 - 7.3.3 Install the tensiometer by lowering the instrument to the bottom of the borehole. Push down lightly on the tensiometer to insure that good contact is made between the ceramic tip and native soil.
 - 7.3.4 Add ~6 inches of native soil to the borehole. NOTE: Use the stockpiled cuttings of 7.2.1 and add the cuttings to the borehole at the same depth from which they were removed so as to simulate as near as possible the original downhole conditions.
 - 7.3.5 Compact the backfill around the tensiometer using a 1-inch tamping rod. NOTE: Initial backfill lifts should be limited to 6-inches or less to insure that sufficient compaction of the soil around the instrument occurs.
 - 7.3.6 After~4 feet of backfill has been added and compacted, place a one (1) foot thick bentonite seal in the borehole so that no vertical migration of water is possible. NOTE: Introduce the bentonite seal into the borehole by either pouring it down the borehole or by use of a tremie pipe.
 - 7.3.7 Once the bentonite has been placed, pour ~5 gallons of distilled water down the borehole in order to hydrate the bentonite.
 - 7.3.8 Wait at least 30 minutes between the addition of the water and the addition of the next lift of native soil so that the bentonite has sufficient time to hydrate.
 - 7.3.9 Add bentonite/native soil mixture to the borehole and compact the bentonite/soil mixture with a one-inch tamping rod. See 3.15 and 3.5.



ALBUQUERQUE OPERATIONS

7.3.10 Continue backfilling and tamping until the borehole is filled to ground level.

8.0 PROCEDURE: Tensiometer logiallation

- 8.1 TAC personnel shall check the tensiometer for proper operation prior to final installation.
- 3.2 Replace any tensiometer which fails to operate properly in the same borehole, or in another borehole drilled adjacent to the first.
- 8.3 Sink a PVC pipe as protective casing into the bentonite/soil plug, thus surrounding the tensiometer top and gauge.
- 8.4 Extend the casing down approximately two feet into the plug and allow the casing to protrude above the land surface about one (1) foot. Cover the top of the casing with a slip-on PVC cap.
- 8.5 Protect the tensiometer from possible damage by vehicular traffic or grazing livestock whenever appropriate. Place 3-4 wood, steel, or tee-bar driven steel posts in a four-foot radius around each protective casing to protect the casing. Set each post two feet below ground surface and extend at least four feet above the ground surface. Attach flagging to a post to mark the location. If livestock present a threat to the tensiometer, affix three strands of barbed wire to the posts. See Albuquerque Operations Manual, Section 16.1.1, paragraph 3.7.7 for details on the installation of the posts.

9.0 PROCEDURE: Gauge Calibration

- 9.1 The tensiometer section pressures are measured using a Bourdon tube gauge. These gauges are calibrated by the manufacturer. Check, recalibrate, or replace the gauges under the following circumstances:
 - 9.1.1 If it is known that the tensiometer has been subjected to sub-freezing conditions.
 - 9.1.2 If there is evidence of physical damage to the gauge.
- 9.2 Check the zero adjustment of the gauges after they are taken out of the soil and before the onset of freezing conditions.
 - 9.2.1 Place the ceramic tip in distilled water and read gauge value. The gauge should read zero because it was set at zero at the time of installation.

JEG-AL-QA-21 (8/87)

ALBUQUERQUE OPERATIONS

9.3 If freezing conditions are expected to occur, keep the water level in the tubes at least five feet below the surface. NOTE: Do not refill and calibrate until actual measurements are to be made. The gauge should be protected by insulation or placed in the project storage facility for the winter to protect from freezing.

SECTION 16.1.7

10.0 PROCEDURE: Field Measurements

- 10.1 Most tensiometers have a dial type manometer to indicate pressure potential. Their scale is given in centibars, from 0 to -100. Therefore, -50 centibars is equal to -0.5 bar which in turn is equivalent to a pressure potential of -38.0 cm water.
- 10.2 Read gauge and record tensioneter pressure potentials on the field form entitled "Tensiometer Sampling Record". See Attachment 12.2.

11.0 PROCEDURE: Completion of Duties

- 11.1 Clean field equipment.
- 11.2 Replace and report any broken or damaged items.
- 11.3 Make copies of field forms and give the original data to document control. Transmit copies of the field forms to the SH.

12.0 AFTACHMENTS

- 12.1 Tensiometer checklist
- 12.2 Tensiometer Sampling Record
- 12.3 Figure 1: Component Parts of a Tensiometer
- 12.4 Figure 2 and 3: Removing Air from Tensiometer
- 12.5 Figure 4: Adjusting the Vacuum Gauge Dial

13.0 REFERENCES

- 13.1 Allmaras, R. R., 1965. Bias, in C. A. Black, ed., "Methods of Soil Analysis", Agronomy Monograph No. 9, Part 1, Madison, Wisconsin, American Society of Agronomy, pp. 24-42.
- 13.2 Krumbein, W. C., and F. A. Graybill, 1965. An Introduction to Statistical Models in Geology, McGraw-Hill, New York, New York, pp. 171-181.

JE ALBUQUERQUE OPERATIONS JACOBS ENGINEERING GROUP INC. SECTION ______ 16.1.7

	NMIN
Prepared	By: htplowe
	(AC Jeonnical Representative
Reviewed	By: Mater
	7 Task Manager
Reviewed	By: Ach KBeugles
	Quarity Assurance Manager
Approved	By: Mant Ais
	Project Manager

JE JACOBS ENGINEERING GROUP INC.

<u>Tensiometer checklist</u> :	Location identification [ate: Name:
Tensiometer tubes:	
Length: 6" 12" 18" 24" 30" 36" 42" 48" 60"	Quantity:
Other:	
Vacuum dial gauges:	
Quantity:	Condition:
Ceramic tips:	
Quantity:	Condition:
Reservoir fill caps:	
Quantity:	Condition:
Installation equipment:	4" soil auger: insertion tool: 1/4" soil screen: vacuum hand pump: other:

Comments:

đ,

Attachment 12.1 - Tensiometer equipment checklist

	and the state of the		tenne levente rece posts or entre sector and a sector sector and
	JE	JACOBS ENGINEERING GROUP INC.	
	TENS	IOMETER SAMPLING RECOR	RD
	SITE ID		
	LOCATION ID	anna an an an ann an an an an an an ann an a	
	SAMPLE ID	TENSIOMETER DEPTH (INCHES)	PRESSURE POTENTIAL (CENTIBARS)
۱			
2.			and an effect of the state of the
3.	-	Notice and the second	
4.			Inside the second s
5.	-		
6.			
7.			
8.	-		
9.		-	
10.			

COMMENTS:

FIELD REP. (S):

ATTACHMENT 12.2



CONPONENT PARTS OF A TENSIOMETER



FIGURE 2 FILLING TENSIONMETER TUBE BY INDUCING VACUUM





FIGURE 3 REMOVING AIR FROM TENSIOMETER



	JE JACOBS EN	GINEERING GROUI Atrona	P INC.
I	RANSMITTAL	OF DOCI	JMENTS
ATE SENT : 11/29/89	PAGE OF	<u> </u>	CONTROL NUMBER
O: F TITES		FROM :	Quality Assurance Department
YPE AND TITLE OF MATER	RIAL SENT :		
lbuquerque Operations Man oil-Water Sampler Install 6/22/89	ual, Section 16. ation and Sample	1.6 Collection	
Neproce ord version with	new version		
	UEDDIONO .		
DISPOSITION OF PREVIOUS Discard	VERSIONS :		
DISPOSITION OF PREVIOUS Discard <u>ACKNO</u>	WLEDGMENT	OF RECEIPT	OF MATERIALS
DISPOSITION OF PREVIOUS Discard <u>ACKNO</u> ACKNOWLEDGE RECEIPT	WLEDGMENT OF THE MATERI	OF RECEIPT	OF MATERIALS
DISPOSITION OF PREVIOUS Discard <u>ACKNO</u> ACKNOWLEDGE RECEIPT <i>Had</i>	WLEDGMENT OF THE MATERI Column	OF RECEIPT	OF MATERIALS OVE.
DISPOSITION OF PREVIOUS Discard <u>ACKNO</u> ACKNOWLEDGE RECEIPT <u>Ha</u> J SIGNATE	WLEDGMENT OF THE MATERI MALLA URE	OF RECEIPT	OF MATERIALS OVE. 12/01/89 DATE

ż,

JE JACOBS ENGINEERING GROUP INC.

ALBUQUEBQUE OPERATIONS MANUAL	SECTION	16.1.8
ALDOGOLINGOL OF LITATIONS WANDAL	DATE	9/12/88
	EFFECTIVE	10/27/88
BATCH AND COLUMN LEACH TESTING	SUPERSEDE	S
	PAGE 1	OF 5

1.0 PURPOSE

To provide standard instructions for personnel conducting batch and column leach tests on uranium mill tailings, soils, sediments, or combinations thereof.

2.0 SCOPE

This procedure establishes guidance for setting up and conducting batch or column leach experiments. In addition, criteria are set forth that address record keeping, instrument calibration, and the various data necessary to allow reconstruction of the experiments at a later date by people who were not associated with the original work.

3.0 PROCEDURES

Collect soil, sediment, or tailings samples using clean utensils and sample containers. The methods for cleaning will be dependent upon the components of interest (e.g., samples collected for organic analyses should not come in contact with plastic, etc.). Logs cf test pits and core holes shall be prepared as per AOM, Section 14.1.4.

- 3.1 Moisture content determination
 - a. A 100g homogenized aliquot of each sample will be weighed, oven dried at 100° and reweighed per ASTM Method 2216 (ASTM Annual Book of Standards, 1984, Volume 04.08, Soil and Rock).
 - b. Record the results in the laboratory log book.
- 3.2 Batch leach experiments

Batch leach experiments will be according to the following procedure (modified from ASTM Standard Test Method 4319-83):

- 3.2.1 Apparatus:
 - a. Pyrex glassware or FEP TFE (teflon) beakers, centrifuge tubes, flasks, etc.; cleaned by washing with a laboratory detergent, rinsed a minimum of three times in deionized water (ASTM Class II or better) and air dried.

b. Centrifuge (min. cap 1400g) or filtering apparatus.

PAGE 2 OF 5

- JACOBS ENGINEERING GROUP INC. SECTION 16.1.8
 - c. Laboratory shaker or rotator.
 - d. Analytical balance.
 - e. Laboratory log book.
 - 3.2.2 Solutions
 - a. Site specific waters, synthetic groundwater or lixiviant solutions can be used as a leach medium.
 - Prepare synthetic groundwater or lixiviant solutions b., using reagent grade chemicals.
 - c. Record masses of chemicals and volumes of liquid used in making the solutions in the laboratory log bock.
 - 3.2.3 Procedure:
 - a. Based upon the results of the moisture content determination, weigh out the equivalent of 75g + 5 g (dry weight) of sample and record the weight (nearest 0.01 g).
 - b. Introduce about 300 mL of solution into the sample container. The actual volume (mL) of lixiviant should be four times the weight of the solid sample in grams.
 - Shake the contents of the vessel to ensure contact C . between the liquid and solid phases. Place the vessel on a laboratory shaker/rotator and gently agitate for a minimum of two hours per day for a minimum of three days.
 - d. At the end of the contact period, centrifuoe (minimum 1400g) or filter (maximum 0.45 mm pore size) the lixiviant solid sample mixture. One hundred mL of the solution will be acidified to pH 2.0 with HN03 and placed in refrigerated (4°) storage. The remainder of the solution will be refrigerated (4°C) without treatment and will have the following parameters measured upon it within one hour of collection: pH, Eh (Redox Potential), and electrical conductivity.
 - 3.3 Column leach experiments
 - 3.3.1 Apparatus:
 - a. 4" diameter by 12" long polycarbonate cylinders with either rubber stoppers or machined fittings to seal

JE JACOBS ENGINEERING GROUP INC. SECTION

SECTION 16.1.8 PAGE 3 OF 5

each end. The lower end seal will be breached by an exit tube with a stopcock or other means to control flow. The interior end of this exit tube will be protected from soil sample intrusion by an inert filter. The cylinders and all associated apparatus will be washed using a laboratory detergent and will be rinsed a minimum of three times with deionized water (ASTM Class II or better) and air dried.

3.3.2 Solution

- a. Defonized water (ASTM Class II or better)
- b. Synthetic or natural groundwaters
- c. Lixiviant solutions

3.3.3 Procedures

- a. The solid sample mixture will be packed into columns to a depth of 10 inches. The upper the inches of the column will serve as a fluid reservoir. The tailings should be packed into the column in ten lifts. The density should produce a transmissivity of 10⁻³ to 10⁻⁵ cm sec⁻¹. In the event that lixiviants are to be added in their dry form, measured amount should be either mixed in with the solids or added as a layer to the top of the solids column.
- b. A deionized water, lixiviant, or selected specific water reservoir will be connected to the top of the columns and the solids will be allowed to come to saturation. At saturation, the flow rate will be controlled at one pore volume per day by adjusting the exit flow. The head should be adjusted to ensure a minimum flow rate of one pore volume per five days.
- c. The experiment will be concluded at the end of 20 days (or a specified number of pore volumes). Samples will be collected for analysis after elution of the following pore volumes: 2, 5, 10, and immediately before termination of the experiment.
- d. Sample aliquots will be collected directly from the effluent stream. A 100 mL aliquot will be collected, acidified to pH 2.0 with HNO3 and refrigerated (4°C). This will be followed by the collection of a 200 mL aliquot which will be refrigerated until collection for analysis. The pH, Eh, and electrical conductivity will be measured on this second sample aliquot.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.8 ALBUQUERQUE OPERATIONS

PAGE 4 OF 5

4.0 RECORDS

In order to document the activities associated with each testing a. laboratory notebook will be maintained. This notebook will contain the following information:

- Α. Instrumentation calibration
 - 1. pH meter listed by number or description
 - a. Date and time of calibration
 - b. Standard used as set point
 - c. Readings for other known pHs
 - d. Slope factor
 - 2. Redox potential (Eh) meter listed by number or description a. Date and time of calibration
 - b. Standard used as set point
 - c. Readings for other known Ehs
 - Conductivity meter listed by number or description a. Date and time of calibration b. Conductivities of standards
- Solid sample moisture content Β.
 - 1. Date and time of determination
 - 2. Sample number and description
 - 3. Mass of sample before drying
 - 4. Drying time and temperature
 - 5. Mass of dry sample
 - 6. Calculated percent moisture
- Effluent characteristics С.
 - 1. Date and time of sample collection
 - 2. Sample number and description
 - 3. Physical description clear brown, cloudy yellow, etc.
 - 4. Chemical parameters measured at time of collection . dependent upon type of experiment.
 - a. pH
 - b. Eh
 - c. Conductivity
 - d. Excess lixiviant
 - e. Alkalinity
 - f. Any other parameters that may be specific to the type of experiment

5.0 REFERENCES

ASTM, 1984, Method D2216, Laboratory Determination of Water (Moisture) Content of Soil, Rock, and Soil Aggregate Mixtures; Vol. 04.08, Soil and Rock, Building Stones; ASTM, 1916 Race St., Philadelphia, PA.



JACOBS ENGINEERING GROUP INC.

SECTIC : 16.1.8 PAGE 5 OF 5

ASTM, 1984, Method 4319, Distribution Ratios by the Short-Term Batch Method, Vol. 04.08, Soil and Rock, Building Stones; ASTM, 1916 Race St., Philadelphia, PA.

	- 111 -
Prepared	By: Athonic
Reviewed	By: F.T. France
Reviewed	By: But Assurance, Manager
Approved	By: Anthe Broject Martager

JE JACOBS ENGINEERING GROUP INC.

AL RUOLEROUE OPERATIONS MANUAL	SECTION	16.1.10
ALBOQUERQUE OFERATIONS WANDAL	DATE	8/29/90
FIELD MEASUREMENT OF WATER SAMPLES FOR TEMPERATURE, CONDUCTIVITY, pH, ALKALINITY,	EFFECTIVE	9/10/90
	SUPERSEDE	S
AND TOTAL ACID	PAGE 1	OF 16

1.0 PURPOSE

To provide procedures by which field determinations for temperature, pH, conductivity, alkalinity, and total acidity will be performed and documented.

2.0 SCOPE

These procedures are applicable to members of the TAC and its contractors, subcontractors with responsibilities and duties for the field measurements of collected water samples prior to shipment to the analytical laboratory.

3.0 EQUIPMENT

- 3.1 Pumps
- 3.2 Hoses/Tubes
- 3.3 pH Meter & Electrode with "slope" adjustment feature and with temperature compensator.
- 3.4 Conductivity Meter & Electrode with thermometer.
- 3.5 Thermometer
- 3.6 Magnetic Stirrers and Magnetic stirring bars (Available at Laboratory)
- 3.7 Titrator
- 3.8 Filter Apparatus and Filters
- 3.9 Erlenmeyer Flask: 125 mL (Available at Laboratory)
- 3.10 Volumetric Flasks, Class A: 100, 250, 500, and 1000 mL (Available at Laboratory) All but 100 mL are required only if preparing solutions listed in 4.0.
- 3.11 Graduated Cylinder: 50, 100, 250, and 500 mL (Available at Laboratory) All but 100 mL are required only if preparing solutions listed in 4.0.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.10 ALBUQUERQUE OPERATIONS

PAGE ____ OF ____16

- Beakers: 150, 250, and 400 mL (Available at Laboratory) Required 3.12 only if preparing solutions listed in 4.0.
- Sample Bottles: 250, 500, 1000 mL (or U.S. equivalents) 3.13
- 3.14 Cooler w/"blue ice" or equivalent
- 3.15 Parafilm or its equivalent

4.0 REAGENTS & SUPPLIES

NOTE: Whenever required by the procedure, JEG purchases and uses prepared, calibrated and certified solutions from chemical supply firms. The following are listed in the event certified solutions are not readily available or are used whenever procedure parameters do not require use of certified solutions. In the event that solutions require preparation, they will be prepared at the laboratory before going into the field. Items 4.12 through 4.21 are required only as needed.

- Potassium Chloride: KCl. Reagent Grade. Purchased from chemical 4.1 supply firms.
 - 4.1.1 0.2 M Potassium Chloride: 0.2 M KCl Standard Solution (equivalent to 24,820 umhos/cm and 14,910 mg KCl/L). Weigh out 14.91 grams of KCl crystals into a 1000 mL volumetric flask containing 750 mL of distilled water. Mix solution to dissolve KCl salt. Add distilled water to 1000 mL mark, stopper flask, and mix solution well. Store solution in a polyethylene or glass bottle labelled with "24,820 umhos/cm: 14,910 mg KC1/L", "0.2 M KC1: 14.91 g KC1/L distilled water", date, and preparer's initials. Make all dilutions from this bottle.

Concentra	Conductance o	f KCl Solutions @ 2	'5° C.*
	tion	Conductance (umho/cm)
M 0.0 0.0001 0.0005 0.001 0.005 0.01 0.02 0.05 0.1 0.2 0.5	<u>PPM</u> 7.5 37.3 74.6 373 746 1,491 3,728 7,455 14,910 37,276 74 551	Equivalent 149.85 149.43 147.81 146.95 143.55 41.27 138.34 133.37 128.96 124.08 117.27 111.87	Specific 14.94 73.90 147.0 717.8 1,413 2,767 6,668 12,900 24,820 58,640 111.900

*From Reference 12.1. See 12.0.

ALBUQUERQUE OPERATIONS GROUP INC. SECTION

4.2 KCI-A Solution: Pipet 10 mL of the 0.2 M KCI Standard Solution into a 200 mL volumetric flask containing 150 mL of distilled water and mix solution well. Add distilled water to 200 mL mark of flask, stopper flask, and mix solution well. Label flask as "KCI-A", "1,413 umhos/cm", "10 mL of KC3 Standard Solution/ 200 mL of distilled water", date, and preparer's initials.

16.1 10

PAGE _____ OF ____ 16

- 4.3 KCI-B Solution: Pipet 10 mL of the KCI Standard Solution into a 100 mL volumetric flask containing 70 mL of distilled water and mix solution well. Add distilled water to 100 mL mark, stopper flask, and mix solution well. Label flask as "KCI-B", 2,767 umhos/cm", 10 mL of KCI Standard Solution/100 mL of distilled water", date, and preparer's initials.
- 4.4 KC1-C Solution: Pipet 25 mL of the KC1 Standard Solution into a 100 mL volumetric flask containing 70 mL of distilled water and mix solution well. Add distilled water to 100 mL mark, stopper flask, and mix solution well. Label flask as "KC1-C", "6,668 umbos/cm", "25 mL of KC1 Standard Solution/100 mL of distilled water", date, and preparer's initials.
- 4.5 KC1-D Solution: Pipet 50 mL of the KC1 Standard Solution into a 100 mL volumetric flask containing 40 mL of distilled water and mix solution well. Add distilled water to 100 mL mark, stopper flask, and mix solution well. Label flask as "KC1-D", "7,446 unhos/cm", "50 mL of KC1 Standard Solution/100 mL of distilled water", date, and preparer's initials.
- 4.6 Hydrochloric Acid: HCl. 36.5-38%, 12 M (~12 N); Reagent Grade. Purchased from chemical supply firms.
- 4.7 Nitric Acid: HNO₃. 70%, 16 M (~16 N); Reagent Grade. Purchased from chemical supply firms.
- 4.8 Sulfuric Acid: H₂SO₄. 98%, 18 M (~36 N); Reagent Grade. Purchased from chemical supply firms.
- 4.9 Sodium Hydroxide: NaOH. Reagent Grade. In pellet form. Purchased from chemical supply firms.
- 4.10 -1 N NaOH Solution: Add 40 g of reagent grade Sodium Hydroxide pellets to a 1000 mL volumetric flask containing 900 mL of distilled water or its equivalent and swirl solution to dissolve the NaOH pellets. Cool solution in an ice water bath. Add distilled water to the 1000 mL mark, stopper flask, and mix solution well. Store solution in a polyethylene bottle labeled with "-1 N NaOH", "40 g NaOH/1000 mL of distilled water', date, and preparer's initials. Prepare fresh solution every 30 days and add new date.

4.11 1.6 <u>N</u> NaOH. HACH cartridge. Purchased from chemical supply firms.
4.12 0.16 <u>N</u> NaOH. HACH cartridge. Purchased from chemical supply firms.

JACOBS ENGINEERING GROUP INC. SECTION _________

- 4.3 pH 1 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased from chemical supply firm. Supplied in tablet or powder form. NOTE: Prepare fresh before use as instructed. Discard solution after use.
- 4.14 pH 4 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.15 pH 5 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.16 pH 7 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.17 pH 10 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.18 pH 12.45 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased from chemical supply firm. Supplied in powder form. NOTE: Prepare fresh before use as instructed. Discard solution after use. Due to the instability of this high pH buffer solution, prepare this buffer solution fresh as nerded and do not store.
- 4.19 Distilled Water or Deionized Water
- 4.20 Masking Tape, Nylon Tape, Electricians Tape, and Magic Marker.
- 4.21 Plastic Sheet and Aluminum Foil.
- 4.22 Lint-free Tissue(s).
- 4.23 Data Collection Sheets: Water Quality Sampling Record Furms, Chain-of-Custody Forms, and Sample Collection Procedures.

5.0 PROCEDURE: Field Measurements of a Collected Water Sample

5.1 Whenever practical, final field measurements are to be taken after at least three (3) borehole volumes have been pumped from the well and as close to the time of sampling as is possible.



JACOBS ENGINEERING GROUP INC.

SECTION 16.1.10 PAGE 5 OF 16

- 5.2 Field measurements to determine temperature, conductivity, and pH are performed on a water sample as it is being collected and are to be recorded on Attachment 16.1.10 2.
- 5.3 Alkalinity is determined on an aliquot of the raw water sample and prior to shipment of the sample to the laboratory. Titration volumes at a specified pH are observed and recorded on Attachment 16.1.10.2, page 3 of 4. Known alkalinity solution data are observed and recorded on Attachment 16.1.10.3.
- 5.4 Total Acid determinations are performed on selected groundwater or lysimple: (pore water) samples. This data is recorded in the "COMMENTS" portion of Attachment 16.1.10.2.
- 6.0 PROCEDURE: Temps ature Measurement

8100

A conductivity a cer and electrode with a thermometer built into the er and are size agly recommended for this procedure.

- Use a brush to scrub the thermometer. Rinse thermonology well with distilled water and blot dry with lint-free tissue(s).
 - 6.1.1 NOTE: <u>Always</u> blot dry thermometer and electrice with issue. Water droplets remaining on these protes may billie or otherwise affect measurement data. We bing proce with tissue may result in warming the thermometer or produce an electrostatic charge on the electrode and so directly alter meter readings.
 - 0.1.2 If Conductivity Meter has a thermometer built into the electrode, use instructions of 6.1.
- 6.2 A combination conductivity-temperature probe is used in the field. This thermometer must be checked against a lab grade mercury thermometer at least once a month. At the same time, the conductivity meter is also tested for accuracy, using the following PTTERE 4.
 - 6.2.1 For approximately 30 mL of lab prepared KCL solution of 4.3 above into a clear beaker. Place the clean temper ture- conductivity probe into the solution. Place the beaker in an ice bath, and using a lab grade mercury thermometer, chill the solution to as near 0° C as possible. Soit should be placed on the ice to reach this low temperature.
 - 6.2.2 When the KCL solution has reached a maximum of 3° C, read and record the toperature from the lab thermometer and the meter. Also read and record the conductance.

ALBUQUERQUE OPERATIONS

• Section 16.1.10 PAGE 6 OF 16

NOTE: If meter has an automatic temperature compensator (ATC), turn off ATC or turn dial to "0." Do not use ATC for this test.

6.2.3 Remove the beaker from the ice bath to allow it to warm. Continue reading and recording temperature pairs as well as conductance at least every 2° C up to 25° C. Remove thermometer from the beaker, clean with distilled water, pat dry with tissue(s) and store.

6.2.4 Remove temperature-conductivity puble from beaker, rinse with distilled water, and blot dry with tissues. Pour approximately 20 mL of the other three KCl standards (at room temperature) into four separate clean beakers. Measure and record the conductance of each solution, being sure to rinse and dry the probe between measurements.

- 6.2.5 Remove temperature-conductivity probe from the final beaker, rinse with distilled water, dry with tissue(s) and store.
- 6.2.6 Using the data from 6.2.2, 6.2.3, and 6.2.4, plot lab thermometer temperature versus meter temperature on arithmetic graph paper. Plot meter temperature versus meter conductivity and meter conductivity versus known conductivity on arithmetic graph paper. Submit the data and graphs to the water sampling manager.
- 6.2.7 The meter is now ready for measuring temperature and conductance in the field.
- 6.3 Insert the dry and clean thermometer into the flow-through cell. Allow the temperature to stabilize. Record the temperature measurements periodically on Attachment 16.1.10.2 throughout the time of pumping.
 - 6.3.1 Place the flow-through cell close to the wellhead. Be sure to keep the cell and discharge hose/tube out of direct sunlight.
 - 6.3.2 Use the temperature data to adjust the pH temperature compensator.
- 6.4 When the sample bottle is full, shut off the pump, and carefully withdraw the thermometer from the flow-through cell. NOTE: If water sample looks "dirty" and contains residue that will not be removed

with distilled water, clean thermometer probe as stated in 6.1.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.10 LEUQUEZQUE OPERATIONS

PAGE __ 7 __ OF __ 16

- 6.5 Rinse thermometer probe with distilled water and blot dry with lint-free tissue(s). The probe is now ready for another water sampling activity or storage.
 - 6.5.1 If the probe is to be stored, place it in its carrying case. Label with "clean", date, and initials.
- 6.6 Clean equipment as described in 8.8 through 8.10 of Section 16.2.1.
- 6.7 If Conductivity, pH, Alkalinity, or Total Acidity determination is required, proceed with 7.0, 8.0, 9.0, or 10.0.
- 6.8 To prepare sample for filtration (if required), preservation (if required), and shipment of samples to the analytical laboratory, proceed with 12.0 of Section 16.2.1.

7.0 PROCEDURE: Conductivity Measurement

A Conductivity Meter and electrode with a thermometer built into the electrode is strongly recommended for this procedure. Because conductivity measurement instruments and instrument operation instructions vary with each manufacturer, the analyst is advised to use the specific instructions for instrument operations that are provided with the conductivity meter to be used by the analyst.

If not already performed, test conductivity meter as described in Section 6.2.

- 7.1 Remove electrode and cable from carrying case, rinse well with distilled water, and blot dry with tissue.
 - 7.1.1 Always blot dry electrode with tissue. Water droplet(s) remaining on the electrode may dilute or otherwise affect conductivity measurement(s). Rubbing electrodes with tissue may also produce electrostatic charges on the electrode and thus directly alter meter readings.
- insert clean and dry electrode/cable into the flow-through cell 7.2 before water sampling is to begin. Read sample conductivity and record value(s) periodically throughout the sampling period and especially just after sampling is completed.
- 7.3 Withdraw electrode and cable, rince well with distilled water, and blot dry with tissue paper. Electrode/cable now ready for another sampling activity or storage.
 - 7.3.1 label with "clean", date, and initials if electrode is to be stored.
JACOBS ENGINEERING GROUP INC. SECTION 16.1.10 ALEUQUERQUE OPERATIONS PAGE 8 OF 16

- 7.4 If pH, Alkalinity or Total Acidity determination is required, proceed with 8.0, 9.0, or 10.0. If pH, Alkalinity or Total Acidity determination <u>is not</u> required, proceed with 12.0 of Section 16.2.1.
 - 7.4.1 To prepare water sample(s) for filtration (if required), preservation (if required), and shipment to the analytical laboratory, proceed with 12.0 of Section 16 2.1.
- 7.5 Check Attachment 16.1.10.2 to make sure that ril of the required sampling data has been recorded.
- 7.6 Proceed with 12.9 of Section 16.2.1 to complete required duties.

8.0 PROCEDURE: pH Measurement

A pH meter and electrode with a temperature compensator and a slope adjustment feature are strongly recommended for this procedure. Because pH measurement instruments and instrument operation instructions vary with each manufacturer, the analyst is advised to use the specific instructions for instrument operations that are provided with the pH meter to be used by the analyst. The meter and electrode can be readily adjusted to the temperature of the sample solution and can be calibrated with pH buffer solutions so that the analyst may easily and quickly work with samples having a pH range of 1-4, 4-7, 7-10, or >10.

- 8.1 Rinse the pH electrode with distilled water and blot dry with lint-free tissue(s).
 - 8.1.1 <u>Always</u> blot dry pH electrode with tissue. Water droplet(s) remaining on the electrode may dilute or otherwise affect measurement data. Rubbing electrodes with tissue may result in producing an electrostatic charge on the electrode and so directly alter the meter readings.
 - 8.1.2 When not using pH meter/electrode, <u>always</u> place pH electrode in a container of pH 4 buffer solution and seal with parafilm or its equivalent.
- 8.2 Prior to performing pH measurements in the field or in the lab, at least once per quarter, test the pH meter and Electrode: Determine Temperature and pH. The description below is for a meter with an Automatic Temperature Compensator (ATC). For meters not equipped with an ATC, the procedure is the same.
 - 8.2.1 Prepare a pH 5.0 buffer solution as described on the pH tablet bottle and label as "pH 5.0 Buffer Solution", add date, and write initials on bottle.
 - 8.2.2 Pour 30 mL of pH 5.0 buffer solution into each of three (3) beakers labelled as "A-pH 5.0" (beaker A), "B-pH 5.0" (beaker B), and "C-pH 5.0" (beaker C).

JACOBS ENGINEERING GROUP INC. SECTION 16.1.10 ALBUQUERQUE OPERATIONS

PAGE 9 OF 16

- 8.2.3 Place beaker A in an ice water bath, beaker B on the lab bench, and beaker C in a hot water bath. Allow beakers 10-15 minutes to acclimate to their media. Swirl beakers to mix solution.
- 8.2.4 With automatic temperature compensator (ATC) off or dialed to "O", place a clean and dry pH electrode and a clean and dry lab thermometer into beaker A. Read pH to the nearest 0.01 pH units and temperature to the nearest 0.10 C. Record pH and temperature on Attachment 16.1.10.1.
- 8.2.5 Withdraw electrode and thermometer, rinse each with distilled water, and dry with tissue.
- 8.2.6 Repeat 3.2.4 and 8.2.5 with beaker B.
- 8.2.7 Repeat 8.2.4 and 8.2.5 with beaker C. Store clean electrode as stated in 8.1.2. Store clean thermometer as stated in 6.5.
- 8.2.8 Plot pH and temperature data on semi-log graph paper (pH versus temperature). Attach this data to Attachment 16.1.10.1.
- 8.2.9 Unless foregoing data indicates that pH meter/electrode are not operating properly, pH meter/electrode is ready for sample collection trip in the field or laboratory work.
- 8.3 Calibrate the pH meter and electrode with standardized pH buffer solutions. NOTE: See 4.13 to 4.21.
 - 8.3.1 If pH meter has a slope adjustment feature, calibrate pH meter and electrode with buffer solutions which bracket the sample pH.
 - A. Set ATC on pH meter to the temperature of the water sample. Use temperature data from 6.3. Alternatively, if the pH meter does not have an ATC, chill the pH buffer solution(s) to equal the water sample temperature.
 - B. To calibrate pH meter/electrode for pH 4-7 range, insert clean and dry pH electrode into a labelled beaker containing a fresh solution of pH 7 buffer. Measure pH. Adjust pH reading to 7.0. Withdraw electrode, rinse with distilled water, and blot dry with tissue. Insert clean and dry pH electrode into a labelled beaker containing a fresh solution of pH 4. Measure pH. Adjust the slope to give a pH reading of 4. Withdraw electrode, rinse with distilled water, and blot dry with tissue.
 - To calibrate pH meter/electrode for pH 1-4 range, use 0 pH 1 and pH 4 buffer solutions. Use pH 4 buffer

PAGE 10 OF 16

solution instead of pH 7 buffer solution and pH 1 instead of pH 4 as stated in B. above.

- To calibrate pH meter/electrode for pH 7-10 range, 0 use pH 7 and pH 10 buffer solutions. Use pH 7 buffer solution and pH 10 buffer solution instead of pH 4 as stated in B. above.
- o To calibrate pH meter/electrode for pH 10-12.45 range, use pH 10 and pH 12.45 buffer solutions. Use pH 10 instead of pH 7 buffer solution and pH 12.45 buffer solution instead of pH 4 as stated in B. above.
- C. Insert clean and dry pH electrode into a labelled beaker containing a fresh solution of one of the buffers which bracket the sample. Measure pH. Withdraw electrode, rinse with distilled water, and blot dry with tissue. Repeat with the other buffer which brackets the sample.
 - o If pH is within ±0.02 units, depending on certification of pH buffer solution, proceed with 8.4 and with pH measurements of water samples.
 - o If pH is not within ±0.02 units, depending on certification of pH buffer solution, repeat A, B, and C until pH meter/electrode gives repetitive readings. Repeat at least twice. If pH meter or electrode does not work satisfactorily, see Troubleshooting Instructions for pH meter and electrode.
- 8.3.2 If the pH meter does not have a slope adjustment feature, set ATC on pH meter to the temperature of the water sample. Use temperature data from 6.3. Alternatively, if the pH meter does not have an ATC, chill the pH buffer solution(s) to be used for the calibration or calibration checks to equal the water sample temperature.
 - A. Insert a clean and dry pH meter into a pH 7 buffer solution. If necessary, adjust pH meter to 7 or pH reading on pH 7 buffer bottle. Record pH meter reading. NOTE: Record all pH readings on Attachment 16.1.10.2.
 - B. Withdraw probe, rinse probe well with distilled water, and blot probe dry with tissue paper.
 - C. Repeat A and B.
 - D. Insert clean and dry pH probe into the sample solution, obtain pH reading, and record pH reading.

JACOBS ENGINEERING GROUP INC. SECTION 16.1.10 ALEUQUERQUE OPERATIONS PAGE 11 OF 16

- E. Repeat B.
- F. Insert clean and dry pH probe into a pH buffer solution that best approximates the pH of the sample solution. Obtain and record pH reading.
- G. Repeat B.
- H. Repeat A and B. NOTE: Analyst should determine when and how often 1) pH meter should be recalibrated with pH 7 buffer solution, and 2) pH buffer solution that best approximates the sample pH should be used.
- Analyst should continue with sample(s) until all of pH data has been obtained and recorded. Use A through H.
- 8.4 Insert clean and dry pH electrode into the flow-through cell soon after well pumping has begun. This will eliminate or reduce pH meter drift.
- 8.5 If the pH meter has an ATC, adjust the temperature setting on the pH meter to the temperature of the sample solution. NOTE: Use temperature measurement data from 6.3.
 - 8.5.1 Alternatively, if the pH meter has no ATC, reduce the temperature of the pH buffer solutions when calibrating the pH meter.
- 8.6 Read pH and record measurement within five (5) minutes or after drift has ceased, whichever is sooner. Rinse and blot dry electrode. If possible, perform pH measurement at least three (3) times during the purging of the well.
- 8.7 Periodically during purging, check accuracy of the pH meter with buffer solution(s) that bracket the sample solution and adjust pH meter as needed according to 8.3.1 or 8.3.2.
- 8.8 When purging is completed, withdraw the electrode from the flowthrough cell, rinse the electrode with distilled water, and blot dry with tissue(s). Measure the final pH of each of the two (2) bracketing buffers, being sure to clean the electrode between use of the buffer solutions.
 - 8.8.1 If the meter is accurate within 0.10 units of each buffer, record the results on Attachment 16.1.10.2. Proceed to 8.9.
 - 8.8.2 If the meter is inaccurate (greater than 0.10 units variance with either buffer) recalibrate the meter. Continue purging and recording pH of the sample for at least four (4) to eight (8) liters of sample. Return to 8.7 and repeat until consistent pH reading within 0.10 units is achieved.



JACOBS ENGINEERING GROUP INC. SECTION 16,1.10 ALBUQUERQUE OPERATIONS

- When a sufficient volume of sample has been collected in the sample 8.9 bottle, remove hoses and pumps from the well hole.
- Clean pH electrode: rinse with distilled water and blot dry with 8.10 tissue. pH electrode now ready for another sampling activity or To store electrode, see 8.1. Label with "clean", date, storage. and initials.
- Clean equipment as described in 8.8 through 8.10 of Section 16.2.1. 8.11
- Check Attachment 16.1.10.2 to make sure that all of the sampling 8.12 data has been recorded on the form.
- 8.13. If Alkalinity or Total Acidity determination is required, proceed with 9.0 or 10.0 respectively.
- 8.14 To prepare sample for filtration (if required), preservation (if required), and shipment to the analytical laboratory, proceed with 12.0 of section 16.2.1.

9.0 PROCEDURE: Alkalinity Measurement

Alkalinity determination (total carbonate) is performed on an aliquot of the raw, unfiltered, and unpreserved water sample by titration of the sample with an acid and measured by a pH meter.

- Fill in all sample information on Attachment 16.1.10.2, page 3 of 4. 9.1 NOTE: Record all procedure information on Attachment 16.1.10.2, page 3 of 4.
- Wash the Erlenmeyer flask(s) and volumetric flask(s), to be used for 9.2 this determination, with soap and water. Rinse the flasks THREE (3) times with 20 mL of distilled water. Discard rinses. Set flasks upside down to drain. After 10 minutes, remove remaining water droplets by shaking flasks vigorously.
- 9.3 connect clean pH electrode to the pH meter and rinse electrode well with distilled water and blot dry with tissue.
- Calibrate the pH meter/electrode as described in 8.3 and store 9.4 electrode as described in 8.1.
- Without too much splashing or other vigorous agitation of the water 9.5 sample, carefully pour the water sample into a labelled 100 mL volumetric flask. Add water sample to the 100 mL mark. NOTE: Pour sample down the side of the flask.
- With as little agitation as possible, carefully pour the water 9.6 sample from the flask into a labelled 125 mL Erlenmeyer flask. If available, add a magnetic stirring bar, turn magnetic stirrer on,



JACOBS ENGINEERING GROUP INC.

SECTION 16.1.10 PAGE 13 OF 16

and stir sample slowly. Record sample aliquot volume on Attachment 16.1.10.2, page 3 of 4.

- 9.7 Insert the pH electrode into the sample and suspend the electrode in the water sample. Adjust the ATC on the pH meter to the temperature of the water sample. Determine sample pH and record value.
- 9.8 Place a 1.6 <u>N</u> H₂SO, cartridge in the titrator. Record the titrant lot number of the cartridge (on Attachment 16.1.10.2). Eject a few drops of titrant from the titrator tip, wipe tip with a clean tissue, and reset the counter to "0000".
- 9.9 While gently stirring the sample, titrate the sample with acid by turning the titrator delivery knob. Obtain pairs of pH and alkalinity readings at each pH value listed on Attachment 16.1.10.2, page 3 of 4. NOTE: The alkalinity value at the inflection point between pH 5 and pH 4 will be determined by the Site Hydrologist. The volume in mL equals digital titrator reading divided by 800.
 - 9.9.1 Alkalinity titration endpoints depend upon the total alkalinity concentration. The following endpoints will be used:

Alkalinity	End Point
0-30 mg/L	pH 5.1
31-150 mg/L	pH 4.8
150 mg/L	pH 4.5

- 9.9.2 When the titration is complete, check pH meter calibration (See 8.3). If the meter is accurate within 0.2 units of each buffer value, record the results on Attachment 16.1.10.3. If the meter is inaccurate (greater than 0.2 units of each buffer value), recalibrate the meter and repeat the alkalinity test.
- 9.10 Repeat this procedure with a second Tech Rep performing the titration on a fresh aliquot of water sample.
- 9.11 Calculate the relative error (R.E.) as:

R.E. (%) = $\frac{(X_1 - X_2) \times 100\%}{X_s}$

Where X_1 is the first alkalinity measurement at pH endpoint, X_2 is the second alkalinity measurement at pH endpoint, and X_s is the lesser of X_1 or X_2 .

9.12 If the percentage error is greater than ten percent (>10%), repeat the foregoing procedure until two (2) consecutive measurements are within 10% relative error.

JACOBS ENGINEERING GROUP INC. SECTION PAGE 14

SECTION 16.1.10 PAGE 14 OF 16

- 9.13 Perform alkalinity determinations on three (3) known solution(s) submitted to JEG by a quality control laboratory and record the pH and alkalinity values on Attachment 16.1.10.3. When performing this procedure, the standard solution identification, site identification, alkalinity test kit identification, pH meter identification, date, and Tech Rep name must be filled out on Attachment 16.1.10.3. Send this attachment to the Quality Control Laboratory
 - 9.13.1 Send a copy of Attachment 16.1.10.3 to the Data Manager who will record this data on a form similar to Attachment 16.1.10.3 provided by a quality control laboratory.
 - 9.13.2 Field alkalinity values and known solution values must agree within 10% of each other or the field test will have to be repeated until the values are accurate.
- 9.14 To prepare the sample(s) for filtration (if required), preservation (if required), and shipment, proceed with 12.0 of Section 16.2.1.
- 9.15 If Total Acidity is required, proceed with 10.0.
- 9.16 To complete duties, proceed with 12.9 of Section 16.2.1.

10.0 PROCEDURE: Total Acidity

This procedure is performed on selected groundwater or lysimeter (pore water) samples to determine the total acidity of the solutions and to determine the acid-neutralizing potential of carbonate materials within silty soils or sediments.

Digital Titration with HACH Cartridge

- 10.1 Record site identification, location identification, date, and all titration data under "Comments" on the Soil Water Sampler Data Record, under "Notes" of Attachment 16.1.10.2, or on a separate data sheet.
- 10.? Calibrate the pH meter/electrode as described in 8.3 and store pH electrode as described in 8.1.
- 10.3 Rinse a 100 mL graduated cylinder and a 125 mL Erlenmeyer flask with distilled water. Place cylinder and flask upside down to allow excess water to drain. Remove residual water droplets by vigorously shaking the cylindar and flask. NOTE: If a 100 mL cylinder is unavailable, use a 100 mL volumetric flask.
- 10.4 Carefully pour a 100 mL aliquot of the unfiltered water sample into the graduated cylinder and then carefully pour the sample into the 125 mL titration flask. NOTE: Pour sample down the side of the cylinder and flask.

JACOBS ENGINEERIN'S GROUP INC. SECTION 16.1.10

10.4.11f necessary, use a sample aliquot of less than 100 mL to perform a second titration. NOTE: Two (2) sample aliquots of 50 mL each with titrations on each aliquot are preferred to one (1) sample aliquot of 100 mL and one (1) titration.

- 10.5 Place clean pH electrode into the water sample, add a magnetic stirring bar, place sample on a magnetic stirrer, and <u>gently</u> stir sample. Record the initial sample pH. Withdraw pH electrode, rinse electrode with distilled water and blot dry with tissue.
- 10.6 Place a 1.6 N NaOH cartridge in the titrator. Record the HACH titrant lot number of the NaOH cartridge. Eject a few drops of titrant from the titrator tip, wipe tip with clean tissue, and reset the counter to "0000".
- 10.7 While gently stirring the sample, turn titrator delivery knob and start titration. Record sample pH and corresponding volume of NaOH titrant after every 1-2 turns of the titrator delivery knob (volume in mL = digital titrator reading divided by 800). If pH endpoint of 8.3 is obtained after less than eight (8) turns, then re-do the titration.
 - 10.7.1 Do not use color indicator tablets to determine the pH endpoint.
 - 10./.2 Use a 0.16 <u>N</u> NaOH cartridge to re-do the titration. Record use of 0.16 <u>N</u> NaOH on the data sheet Form 10.1.
- 10.8 Rinse all glassware with distilled water. Set glassware upside down to allow excess water to drain.

10.9 Calculate the total acidity as follows:

Acidity, u equiv/liter =
$$\frac{V_t \times N_t \times 10}{V_a}$$

Where $V_* = Volume$, in mL, of NaOH titrant used to obtain pH 8.3, N, = Normality of NaOH used to titrate the sample, and $V_* = Volume$, in mL, of sample aliquot (should be 50 or 100 mL, unless lysimeter produces very little water). NOTE: Be sure to specify and record which NaOH and Volume of NaOH are used for the titration.

- 10.10 Repeat the titration procedure with a second Tech Rep performing the titration unless there is insufficient sample volume produced by the well or lysimeter.
- 10.11 Calculate the relative error in percent (R.E., %) as previously described in 9.11.
- 10.12 If the relative error is greater than 10%, repeat the procedure a third time, unless there is insufficient sample volume available.

JACOBS ENGINEERING GROUP INC. SECTION _______ 6.1.10 ALBUQUERQUE OPERATIONS

PAGE 16 OF 16

- 10.13 Check Attachment 16.1.10.2 to see that all of the required data has been recorded on the forms.
- See 12.9 of Section 16.2.1 to complete required 10.14 duties.

11.0 ATTACHMENT(S): Supporting Document(s)

- 16.1.10.1 Equipment Checklist
- 16.1.10.2 Water Quality Sampling Record Form

16.1.10.2 Alkalinity Titration Form (page 3 of 4)

16.1.10.3 Standard Alkalinity Test

12.0 REFERENCES

- 12.1 APHA/AWWA/WPCF, 1985. <u>Standard Methods for the</u> <u>Examination of Water and Wastewater</u>, 16th Edition. American Public Health Association, Washington, D.C. 20005.
- 12.2 Korte, N. and P. Kearl, 1984. "Proceaures for the Collection and Preservation of Ground Water and Surface Water Samples and for the Installation of Monitoring Wells," U.S. Department of Energy, Grand Junction, Colorado.

13.0

PROCEDURE REVIEW AND APPROVAL

Prepared	By:	- Jale Fut 8-30-90	
		TAC Representative Date	
Reviewed	By:	Task Manager Date	
Reviewed	By:	Dub & Bench 9-7-90	
Approved	Byr	Aunand - Asin	
	Prs	Mect Manager/Deputy Project ManagerDate	

JE ACOBS ENGINEERING GROUP INC.

AL BLIOLIEBOLIE OPERATIONS MANUAL	SECTION 16.1.10	
ALDOGOLINGOL OF LITATIONS WANDAL	DATE 8-29-90	
FIELD MEASUREMENT OF WATER SAMPLES FOR	EFFECTIVE 09/10/90	
TEMPERATURE, CONDUCTIVITY, pH, ALKALINITY, AND TOTAL ACID	SUPERSEDES	
	PAGE 1 OF 17	

1.0 <u>PURPOSE</u>

To provide procedures by which field determinations for temperature, pH, conductivity, alkalinity, and total acidity will be performed and documented.

2.0 SCOPE

These procedures are applicable to members of the TAC and its contractors, subcontractors with responsibilities and duties for the field measurements of collected water samples pricr to shipment to the analytical laboratory.

3.0 EQUIPMENT

3.1 Pumps

- 3.2 Hoses/Tubes
- 3.3 pH Meter & Electrode with "slope" adjustment feature and with temperature compensator.
- 3.4 Conductivity Meter & Electrode with thermometer.
- 3.5 Thermometer
- 3.6 Magnetic Stirrers and Magnetic stirring bars (Available at Laboratory)
- 3.7 Titrator
- 3.8 Filter Apparatus and Filters
- 3.9 Erlenmeyer Flask: 125 mL (Available at Laboratory)
- 3.10 Volumetric Flasks, Class A: 100, 250, 500, and 1000 mL (Available at Laboratory) All but 100 mL are required only if preparing solutions listed in 4.0.
- 3.11 Graduated Cylinder: 50, 100, 250, and 500 mL (Available at Laboratory) All but 100 mL are required only if preparing solutions listed in 4.0.

JE JACOBS ENGINEERING GROUP INC. SECTION _16.1.10 ALBUQUERQUE OPERATIONS

PAGE ____ OF ____

3.12 Beakers: 150, 250, and 400 mL (Available at Laboratory) Required only if preparing solutions listed in 4.0.

- 3.13 Sample Bottles: 250, 500, 1000 mL (or U.S. equivalents)
- 3.14 Cooler w/"blue ice" or equivalent
- 3.15 Parafilm or its equivalent

4.0 REAGENTS & SUPPLIES

NOTE: Whenever required by the procedure, JEG purchases and uses prepared, calibrated and certified solutions from chemical supply firms. The following are listed in the event certified solutions are not readily available or are used whenever procedure parameters do not require use of certified solutions. In the event that solutions require preparation, they will be prepared at the laboratory before going into the field. Items 4.12 through 4.21 are required only as needed.

- 4.1 Potassium Chloride: KCl. Reagent Grade. Purchased from chemical supply firms.
 - 4.1.1 0.2 M Potassium Chloride: 0.2 M KCl Standard Solution (equivalent to 24,820 umhos/cm and 14,910 mg KCl/L). Weigh out 14.91 grams of KCl crystals into a 1000 mL volumetric flask containing 750 mL of distilled water. Mix solution to dissolve KCl Add distilled water to 1000 mL mark, salt. stopper flask, and mix solution well. Store solution in a polyethylene or glass bottle labelled with "24,820 umhos/cm: 14,910 mg KCl/L", "0.2 M KCl: 14.91 g KCl/L distilled water", date, and preparer's initials. Make all dilutions from this bottle.

	Conductance of	KCl Solutions	@ 25° C.*
Conc	centration	Conductan	ce (umho/cm)
M	PPM	Equivalent	Specific
0.0		149.85	
0.0001	7.5	149.43	14.94
0.0005	37.3	147.81	73.90
0.001	74.6	146.95	147.0
0.005	373	143.55	717.8
0.01	746	141.27	1,413
0.02	1,491	138.34	2,767
U.05	3,728	133.37	6,668
0.1	7,455	128.96	12,900
0.2	14,910	124.08	24,820
0.5	37,276	117.27	58,640
1.0	74,551	111.87	111,900

*From Reference 12.1. See 12.0.

JEG-AL-QA-21(8/87)

JACOBS ENGINEERING GROUP INC. SECTION 16.1.10 ALBUQUERQUE OPERATIONS

PAGE_3___OF__17_

- 4.2 KC1-A Solution: Pipet 10 mL of the 0.2 M KC1 Standard Solution into a 200 mL volumetric flask containing 150 mL of distilled water and mix solution well. Add distilled water to 200 mL mark of flask, stopper flask, and mix solution well. Label flask as "KCl-A", "1,413 umhos/cm", "10 mL of KCl Standard Solution/ 200 mL of distilled water", date, and preparer's initials.
- 4.3 KC1-B Solution: Pipet 10 mL of the KC1 Standard Solution into a 100 mL volumetric flask containing 70 mL of distilled water and mix solution well. Add distilled water to 100 mL mark, stopper flask, and mix solution well. Label flask as "KC1-B", 2,767 umhos/cm", 10 mL of KCl Standard Solution/100 mL of distilled water", date, and preparer's initials.
- 4.4 KCl-C Solution: Pipet 25 mL of the KCl Standard Solution into a 100 mL volumetric flask containing 70 mL of distilled water and mix solution well. Add distilled water to 100 mL mark, stopper flask, and mix solution well. Label flask as "KCl-C", "6,668 umhos/cm", "25 mL of KCl Standard Solution/100 mL of distilled water", date, and preparer's initials.
- KC1-D Solution: Pipet 50 mL of the KCl Standard Solution 4.5 into a 100 mL volumetric flask containing 40 mL of distilled water and mix solution well. Add distilled water to 100 mL mark, stopper flask, and mix solution well. Label flask as "KCl-D", "7,446 umhos/cm", "50 mL of KCl Standard Solution/100 mL of distilled water", date, and preparer's initials.
- 4.6 Hydrochloric Acid: HCl. 36.5-38%, 12 M (12 N); Reagent Grade. Purchased from chemical supply firms.
- 4.7 Nitric Acid: HNO. 70%, 16 M (16 N); Reagent Grade. Purchased from chemical supply firms.
- 4.8 Sulfuric Acid: H,SO,. 98%, 18 M (36 N); Reagent Grade. Purchased from chemical supply firms.
- 4.9 Sodium Hydroxide: NaOH. Reagent Grade. In pellet form. Purchased from chemical supply "irms.
- 4.10 1 N NaOH Solution: Add 40 g of reagent grade Sodium Hydroxide pellets to a 1000 mL volumetric flask containing 900 mL of distilled water or its equivalent and swirl solution to dissolve the NaOH pellets. Cool solution in an ice water bath. Add distilled water to the 1000 mL mark, stopper flask, and mix solution well. Store solution in a polyethylene bottle labeled with " 1 N NaOH", "40 g NaOH/1000 mL of distilled water", date, and



SECTION _______ OF _____ OF _____ 17

preparer's initials. Prepare fresh solution every 30 days and add new date.

- 4.11 1.6 N NaOH. HACH cartridge. Purchased from chemical supply firms.
- 4.12 0.16 <u>N</u> NaOH. HACH cartridge. Purchased from chemical supply firms.
- 4.13 pH 1 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased from chemical supply firm. Supplied in tablet or powder form. NOTE: Prepare fresh before use as instructed. Discard solution after use.
- 4.14 pH 4 Buffer Solution: Certified as ± 0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.15 pH 5 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.16 pH 7 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.17 pH 10 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased in solution form from chemical supply firm. NOTE: Discard any solution whose expiration date has elapsed.
- 4.18 pH 12.45 Buffer Solution: Certified as ±0.02 pH units and with expiration date. Purchased from chemical supply firm. Supplied in powder form. NOTE: Prepare fresh before use as instructed. Discard solution after use. Due to the instability of this high pH buffer solution, prepare this buffer solution fresh ac needed and do not store.
- 4.19 Distilled Water or Deionized Water
- 4.20 Masking Tape, Nylon Tape, Electricians Tape, and Magic Marker.

1.1

- 4.21 Plastic Sheet and Aluminum Foil.
- 4.22 Lint-free Tissue(s).

See.